5.09 Estuarine and Coastal Sediments – Coupled Biogeochemical Cycling

DJ Burdige, Old Dominion University, Norfolk, VA, USA

© 2011 Elsevier Inc. All rights reserved.

5.09.1	Introduction	279
5.09.2	Introduction to Biogeochemical Processes in Sediments	280
5.09.2.1	Organic Matter Remineralization Processes	280
5.09.2.1.1	The anaerobic food chain for organic matter remineralization	281
5.09.2.2	Chemolithotrophic Reactions	281
5.09.2.2.1	Aerobic processes	281
5.09.2.2.2	Anaerobic processes	282
5.09.2.3	Linkages between Chemolithotrophic and Organic Matter Remineralization Processes	283
5.09.2.4	Seasonality and Non-Steady-State Processes in Estuarine and Coastal Sediments	284
5.09.3	Linkages of Biogeochemical Processes and Transport Processes and the Occurrence	
	of Mixed Redox Conditions in Sediments	285
5.09.4	The Classification of Estuarine and Coastal Sediments	285
5.09.4.1	Steadily Accumulating Sediments	287
5.09.4.2	Sediments Subject to Extensive Physical Reworking	289
5.09.4.3	High Permeability, Sandy Sediments	289
5.09.4.4	Bypass Zone Sediments	291
5.09.4.5	Macrophyte-Dominated Sediments	291
5.09.4.6	Intertidal Mudflats and Sandflats	292
5.09.5	Sediment Oxygen Consumption	293
5.09.6	Organotrophic Denitrification, Anammox and Anoxic Nitrification	293
5.09.6.1	Anammox	295
5.09.6.2	Anoxic Nitrification	295
5.09.6.3	Nitrogen Loss from Sediments: Closing Thoughts	296
5.09.7	Iron and Manganese Reduction	296
5.09.8	Sulfate Reduction and Sulfide Mineral Formation	300
5.09.8.1	Pyrite Formation in Sediments	301
5.09.8.2	Pyrite Burial and Sulfur Burial Efficiency	302
5.09.9	Methanogenesis, Methane Oxidation, and Methane Fluxes to the Atmosphere	303
5.09.10	Trace Metal Cycling	304
5.09.11	Concluding Remarks	307
References		308

Abstract

A variety of macroscopic physical and biological forcing functions act on estuarine and coastal sediments across a range of time and space scales, and can have a profound effect on the processes associated with the remineralization of sedimentary organic matter. These interactions exert a strong influence on the cycling of biogeochemically reactive elements (C, N, S, P and trace metals such as Mn and Fe), and result in linkages between and within these cycles that are far more complex than once envisioned. Biogeochemical processes in estuarine and coastal sediments also have implications on benthic-pelagic coupling in estuaries and the coastal ocean, and on some global oceanic cycles and biogeochemical budgets.

5.09.1 Introduction

Estuarine and coastal marine sediments are impacted by a wide range of macroscopic physical and biological forcings that occur over a broad range of time and space scales. These include, for example, bioturbation and bioirrigation by benthic macrofauna, pore-water advective exchange driven by nearseabed pressure gradients, oxygen input through macrophyte roots, and large-scale physical mixing (reworking) of the sediments by tides, surface currents, or storms. The interplay between these forcings and biological (largely microbial) processes directly or indirectly associated with the remineralization of sedimentary organic matter has a significant impact on the biogeochemical cycling of elements such as carbon, nitrogen, phosphorus, sulfur, and many trace metals, including manganese and iron. Furthermore, because estuarine and coastal systems are relatively shallow, biogeochemical processes that occur in the underlying sediments tend to have a greater overall impact on the dynamics of estuaries and the coastal ocean as a whole.

With this in mind, the goal of this chapter is to present an integrated view of estuarine and coastal sediment biogeochemistry, focusing on these considerations. First, the chapter briefly describes the key biogeochemical processes that occur in these sediments, using the biogeochemical zonation model (Claypool and Kaplan, 1974; Froelich et al., 1979) as the framework for the discussion. Next, it discusses how an understanding of the physical and biological forcings outlined above can be used to characterize estuarine and coastal sediments based on their functional diagenetic characteristics (e.g., Aller, 2004). The remaining sections then link these two discussions together to provide a unified discussion of the coupled biogeochemical processes that occur in estuarine and coastal sediments.

5.09.2 Introduction to Biogeochemical Processes in Sediments

This discussion largely focuses on bacterial metabolism, in part, because there is a much greater diversity of metabolic processes among prokaryotes than there is among eukaryotes. (As used here, the term 'bacteria' is synonymous with 'prokaryotes'. Except where specifically noted, it therefore includes organisms in the group *Archaea*, cyanobacteria, and true bacteria or eubacteria, i.e., organisms in the group *Bacteria*.) Most bacterial processes that impact sediment biogeochemistry result from dissimilatory metabolism (energy-acquiring processes), although assimilatory processes (i.e., production of new bacterial biomass) may be important in some aspects of sediment organic carbon remineralization and preservation (Burdige, 2006).

Bacterial biomass dominates the total benthic biomass in many, but not all, marine sediments (Aller, 1998; Alongi, 1998; Fenchel et al., 1998). Furthermore, because of their relatively large surface-to-volume ratio as compared to higher, eukaryotic organisms, bacterial metabolic processes (on a per organism volume basis) can have a much greater impact on overall sediment metabolism, even if they represent only a few percent of the total benthic biomass (Nealson, 1997). Additional aspects of this problem are discussed in Section 5.09.5

All of these considerations are most pronounced in anoxic (oxygen-deficient) sediments, since only a few protozoans and fungi are capable of anaerobic metabolism. Thus, in most estuarine and coastal sediments, which are anoxic within a centimeter or two (or less) below the sediment–water interface, bacterial metabolism dominates organic matter remineralization and, therefore, drives most of the biogeochemical cycles that will be discussed here.

5.09.2.1 Organic Matter Remineralization Processes

The occurrence of specific organic matter remineralization (respiratory) processes in any natural system appears to be controlled by the free energy yield per mole of organic carbon oxidized by a specific individual electron acceptor (Claypool and Kaplan, 1974; Froelich et al., 1979). Based on this bio-geochemical zonation model, the oxidant that provides the greatest amount of free energy will be utilized first, and when this oxidant is depleted then the next most efficient oxidant will be used (Table 1). In principle, this will continue until either all available oxidants are consumed or all oxidizable (metabolizable) organic carbon is utilized (e.g., Aller and Mackin, 1984).

On much of the Earth's surface and in the oceans, the amount of oxygen is sufficient relative to the amount of reduced (oxidizable) organic carbon, and aerobic respiration predominates. (Note, however, that this is not necessarily the case in all estuaries and coastal waters (see Chapter 5.05 for details).) However, in many estuarine and coastal settings, the flux of reactive organic carbon to the sediments is sufficiently high relative to O₂ transport into the sediments, and anaerobic remineralization occurs. Anaerobic remineralization processes are often referred to as suboxic remineralization (organotrophic denitrification, Mn and Fe reduction) and anoxic remineralization (sulfate reduction and methanogenesis), although strictly speaking suboxic remineralization occurs under anoxic conditions (e.g., in sediments that are devoid of oxygen; see Canfield and Thamdrup, 2009 for an alternate view of this classification scheme for remineralization processes). If we define denitrification in a general sense as the reduction of combined nitrogen (nitrate or nitrite) to gaseous end-products (predominantly N₂), then organotrophic denitrification is a process that couples this reduction to the oxidation of organic matter (e.g., Devol, 2008). Recent studies have also shown that denitrification can be coupled with a number of other oxidative processes (Hulth et al., 2005), hence the need for a more complete description of denitrifying processes.

Under most, but not necessarily all, circumstances, anoxic sediments are defined as those in which sulfate reduction and methanogenesis occur. The biogeochemical zonation illustrated in Table 1 can also be looked from the standpoint of the redox or half cell potential (E_H) of the reduction of each of these terminal electron acceptors. This then leads to their being a gradient from positive E_H values (oxidizing conditions) in

Process	Chemical reaction	∆G° (kJ mol C ^{-1)b}
Aerobic respiration	$C_6H_{12}O_6 + 6O_2 \rightarrow 6CO_2 + 6H_2O_2$	-471
Organotropic denitrification	$5C_6H_{12}O_6 + 24NO_3^- \rightarrow 12N_2 + 24HCO_3^- + 6CO_2 + 18H_2O_3^-$	-444
Organotrophic manganese reduction	$C_6H_{12}O_6 + 18CO_2 + 6H_2O + 12\delta - MnO_2 \rightarrow 12Mn^{2+} + 24HCO_3^{-1}$	-397
Organotrophic iron reduction	$C_6H_{12}O_6 + 42CO_2 + 24Fe(OH)_3 \rightarrow 24Fe^{2+} + 48HCO_3 + 18H_2O_3$	-131
Sulfate reduction	$2C_{6}H_{12}O_{6} + 6SO_{4}^{2-} \rightarrow 6H_{2}S + 12HCO_{3}^{-}$	-76
Methanogenesis	$2C_6H_{12}O_6 \rightarrow 6CH_4 + 6CO_2$	-49

Table 1	Standard state free energy	changes for	the maior or	danic matter r	remineralization	processes
	0	011001000101		900000000000000000000000000000000000000	01111101011201011	0.0000000

^a Data for C₆H₁₂O₆, δ-MnO₂, and Fe(OH)₃ are for glucose, fine-grained vernadite, and amorphous ferric oxide, respectively. Values of standard free energies of formation are from Stumm and Morgan (1996).

^b To convert the values listed here to kJ mol⁻¹ of glucose, multiply by 6.

oxic regions where aerobic respiration occurs, to negative $E_{\rm H}$ values (reducing conditions) in anoxic sediments where sulfate reduction (or methanogenesis) occurs.

Many pore-water profiles, particularly in deep-sea (pelagic and hemi-pelagic) sediments, appear, to a first approximation, to be consistent with this biogeochemical zonation model (Froelich et al., 1979; Burdige, 1993). However in estuarine sediments, particularly those that are bioturbated and/or bioirrigated (Section 5.09.4.1), the distribution of biogeochemical processes in the sediments is more complex than that predicted by this relatively simple conceptual model. These systems will be referred to here as mixed redox sediments (Aller, 1994a; Burdige, 2001), although in the literature they are sometimes discussed in terms of the exposure of the sediments to redox oscillations (see Section 5.09.3 for details).

It is also important to recognize that the biogeochemical zonation model is not necessarily meant to provide specific information about the controls on organic carbon remineralization and preservation in sediments (see Postma and Jakobsen, 1996), with the possible exception of the role that oxygen may play in controlling organic carbon remineralization in some sediments (Emerson et al., 1985; Calvert and Pedersen, 1992; Canfield, 1994; Burdige, 2007). Rather, the model provides information on the occurrence of different terminal electron acceptor (or respiratory) processes that comprise sediment organic carbon remineralization. Such information is of some interest since the occurrence of these specific reductive processes leads both directly and indirectly to a whole host of other processes that play important roles in the overall biogeochemistry of marine sediments. Therefore in spite of its limitations, the biogeochemical zonation model has proved to be quite useful for understanding many aspects of marine sediment biogeochemistry.

5.09.2.1.1 The anaerobic food chain for organic matter remineralization

The anaerobic decomposition of organic matter in sediments occurs in a stepwise manner, involving a consortium of several different functional groups of bacteria (Laanbroek and Veldkamp, 1982; Capone and Kiene, 1988; Fenchel et al., 1998). During this sequence the end product of one process becomes the reactant for the next process, such that in each reaction in the sequence the responsible organism acquires only a portion of the total energy available from the substrate being utilized, passing on the remaining energy in the end products it produces. Because these microbial processes are tightly coupled, the net reaction (i.e., the sum of these processes) can be expressed by the equations in Table 1, despite the fact that the terminal respiration process (e.g., sulfate reduction) is only responsible for the last portion of the overall decomposition sequence (see Burdige (2006) for additional discussions of this anaerobic food chain).

The first process here involves the breakdown (by solubilization, hydrolysis, or oxidative cleavage) of particulate organic matter, followed by processes such as fermentation that produce and consume dissolved organic matter (DOM) compounds of increasingly smaller molecular weights. This eventually leads to the production of a limited number of LMW-DOM compounds, primarily short chain fatty acids such as acetate and formate. These compounds, along with H₂ (also a fermentation end product), are then consumed by the terminal respiratory organisms in the sediments. Competition for these key substrates also provides the kinetic mechanism by which the thermodynamic basis of the biogeochemical zonation model expresses the observed zonation of processes. For example, significant amounts of methane production occur only in the absence of sulfate, and based on lab and field studies, this occurs because sulfate reducers outcompete methanogens for acetate and H₂, key substrates that are used by both groups of organisms (Crill and Martens, 1986; Fauque, 1995; Winfrey and Zeikus, 1977).

This anaerobic food chain model and competition for key substrates also appear to explain the zonation of iron reduction, sulfate reduction, and methanogenesis in freshwater sediments (Lovley and Phillips, 1987; Lovley, 1991), and may describe, in general, the zonation of all suboxic and anoxic processes in estuarine and marine sediments (see discussions in Thamdrup, 2000). However, this latter observation is not well tested.

5.09.2.2 Chemolithotrophic Reactions

In addition to organic matter respiration (i.e., redox reactions involving reduced organic carbon substrates), redox reactions involving only inorganic substances also occur in sediments. Many of these reactions are microbially mediated, by organisms generally referred to as chemoautotrophs or chemolithotrophs. For the most part, these organisms are autotrophs, and use the chemical energy obtained from these reactions to fix carbon and produce new biomass. However, some of these organisms can also assimilate fixed (reduced) carbon compounds from the environment for biomass production.

5.09.2.2.1 Aerobic processes

During anaerobic respiration, a number of reduced end products are produced whose oxidation by O_2 is thermodynamically favored (i.e., the reaction has a negative ΔG°). These compounds include ammonium, methane, reduced iron, and manganese, and several reduced forms of sulfur (e.g., sulfide and elemental sulfur).

Aerobic methane oxidation (methanotrophy) is an important process in many sediments (see Hanson and Hanson, 1996 for a review), and the balance between methane oxidation (both aerobic and anaerobic; see Section 5.09.9) and methanogenesis plays a key role in regulating the methane flux to the atmosphere from aquatic environments (Reeburgh, 2007). Ammonium oxidation, or nitrification, is an extremely common process in nature, both in sediments and in the water column (Ward, 2008). It occurs in two steps, with ammonium first being oxidized to nitrite by one group of organisms, and nitrite to nitrate by another group. The ammonium used in nitrification generally comes from particulate organic nitrogen that is regenerated during aerobic respiration. However in many types of sediments, ammonium produced at depth in the sediments by anoxic remineralization can then be nitrified in the oxic portion of the sediments (see Burdige, 2006 for details). While much of this nitrate may be subsequently consumed by organotrophic denitrification in the sediments, recent work on the anammox reaction (see reaction 1 in Table 2) suggests that the specific mechanism(s) of ammonium loss from sediments needs further examination

Pro	cess	Chemical reaction	⊿G (pH =7)	⊿G (pH= 8)
Rea	ctions coupled to NO_3^- reduction (units: kJ mol nitrate ⁻¹))		
1.	Anammox	$NH_4^+ + NO_2^- \rightarrow N_2 + 2H_2O$	-302	-302
	1a. Anammox ^b	$5/3NH_4^+ + NO_3^- \rightarrow 4/3N_2 + 3H_2O + 2/3H^+$	-421	-425
2.	Denitrification coupled to sulfide oxidation	$NO_3^- + 5/8FeS + H^+ \rightarrow 1/2N_2 + 5/8SO_4^{2-} + 5/8Fe^{2+} + 1/2H_2O_4^{2-}$	-419	-413
3.	Denitrification coupled to Fe ²⁺ oxidation	$NO_3^- + 5Fe^{2+} + 12H_2O \rightarrow 5Fe(OH)_3 + 1/2N_2 + 4 H^+$	-319	-370
4.	Denitrification coupled to Mn ²⁺ oxidation	$5/2Mn^{2+} + NO_3^- + 2H_2O \rightarrow 5/2MnO_2 + 1/2N_2 + 4H^+$	-49	-71
Rea	ctions coupled to manganese reduction (units: kJ mol M	nO2 ⁻¹)		
5.	Anoxic nitrification	$4MnO_2 + NH_4^+ + 6H^+ \rightarrow 4Mn^{2+} + NO_3^- + 5H_2O_3^+$	-44	-35
6.	Mn reduction coupled to oxidative N ₂ production	$3/2MnO_2 + NH_4^+ + 2H^+ \rightarrow 3/2Mn^{2+} + 1/2N_2 + 3H_2O_2$	-150	-141
7.	Mn reduction coupled to complete sulfide oxidation	$2H^+ + MnO_2 + 1/4FeS \rightarrow Mn^{2+} + 1/4SO_4^{2-} + 1/4Fe^{2+} + H_2O_4^{2-}$	-148	-137
8.	Mn reduction coupled to partial sulfide oxidation	$3H^+ + MnO_2 + HS^- \rightarrow Mn^{2+} + S^0 + 2 H_2O$	-136	-119
9.	Mn reduction coupled to iron oxidation	$MnO_2 + 2Fe^{2+} + 4H_2O \rightarrow Mn^{2+} + 2Fe(OH)_3 + 2H^+$	-42	-127

Table 2 Some possible anaerobic (non-02 utilizing) oxidation reactions that may occur in marine sediments^a

^{*a*} The following conditions were used to calculate ΔG values: T = 25 °C; $[NO_3^-] = 5 \mu M$; $[NO_2^-] = 5 \mu M$; $[NH_4^+] = 25 \mu M$; $pN_2 = 0.791$ atm.; $[Mn^{2+}] = 100 \mu M$; $[Fe^{2+}] = 5 \mu M$; $[SO_4^{2-}] = 24 m M$; $[H_2S] = 25 \mu M$; $[HS^-] = 5 \mu M$. Activity coefficients were assumed to be 1. Values for the standard free energies of formation (ΔG°) were taken from Stumm and Morgan (1996).

^b Although anammox organisms appear to use nitrite (rather than nitrate) as the electron acceptor, the source of this nitrite in natural systems is generally the reduction of nitrate (see Section 5.09.6.1 for details). Hence, the biogeochemistry of this process is often thought of in terms N₂ production via nitrate reduction coupled to ammonium oxidation.Modified from Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton; and Hulth, S., Aller, R.C., Gilbert, F., 1999. Coupled anoxic nitrification/manganese reduction in marine sediments. Geochimica Acta 63, 49–66.

(see Section 5.09.6.1). The bacterial oxidation of reduced sulfur species involves several oxidation reactions that eventually produce sulfate:

$$\underset{\text{sulfide})}{\text{HS}^-} \xrightarrow{} \underset{(\text{elemental sulfur})}{\text{S}^\circ} \xrightarrow{} \underset{(\text{thiosulfate})}{\text{S}_2\text{O}_3^{2-}} \xrightarrow{} \underset{(\text{sulfite})}{\text{S}_2^{2-}} \xrightarrow{} \underset{(\text{sulfate})}{\text{S}_2^{2-}} \prod_{\text{sulfate}} [1]$$

This complete oxidative process may actually involve several groups of organisms, each mediating only a portion of the overall sequence, which then results in the transient production of partially oxidized intermediates such as elemental sulfur or thiosulfate. Many of these organisms (e.g., *Beggiatoa, Thioplaca,* and *Thiothrix*) often form mats on the sediment–water interface, allowing them to use bottom water O_2 to oxidize sulfide that diffuses upward toward the sediment surface (see Levin et al., 2009 for a recent review). These organisms are also capable of using nitrate in place of O_2 as an electron acceptor (see Section 5.09.2.2 and Schulz and Jørgensen, 2001). The cycling of many of the intermediates in eqn [1] is an important part of sulfur dynamics in some environments (Canfield et al., 2005). However in many cases, the net oxidation of sulfide to sulfate is of most geochemical significance.

The oxidation of reduced iron and manganese (Fe²⁺ and Mn^{2+}) can also be biologically mediated, although at pH values typical of most marine sediments, abiotic (chemical) iron oxidation is extremely rapid (Stumm and Morgan, 1996) and likely outcompetes biological oxidation (Nealson, 1997). In contrast, Mn^{2+} is much more kinetically stable in the presence of oxygen at natural pH values, and its oxidation by O₂ is often times biologically catalyzed. The majority of the manganese oxidation occurring in sediments appears to be biologically catalyzed (Thamdrup et al., 1994b; Tebo et al., 1997), although unlike the other chemolithotrophic processes discussed in this section, evidence that manganese-oxidizing bacteria gain energy for growth at the expense of this oxidation reaction remains equivocal (Tebo et al., 1997).

5.09.2.2.2 Anaerobic processes

Field observations also suggest that a wide range of other inorganic oxidation reactions may occur in marine sediments using nitrate (or nitrite) and manganese oxides, rather than O_2 , as the oxidant (Aller and Rude, 1988; Bender et al., 1989; Aller, 1994b; Luther et al., 1997; Aller et al., 1998; Hulth et al., 1999; Anschutz et al., 2000; Thamdrup, 2000; Schippers and Jørgensen, 2001; Mortimer et al., 2002; Thamdrup and Dalsgaard, 2002; Trimmer et al., 2003; Mortimer et al., 2004; Engström et al., 2005; Bartlett et al., 2008). Under natural conditions, these reactions have negative ΔG values (see Table 2), and therefore chemolithotrophic bacteria should be able to couple these reactions to carbon fixation. These reactions generally occur in the absence of oxygen (i.e., suboxic or anoxic conditions), or under what is referred to as mixed redox conditions (see Sections 5.09.2.3 and 5.09.3 for details).

While many of the geochemical studies cited above are indeed consistent with these anaerobic oxidation reactions being biologically mediated, certain of the reactions can occur abiotically. For example, manganese reduction coupled to iron oxidation (reaction 9 in **Table 2**) is important in other biogeochemical processes in sediments (Burdige, 2006), although its kinetics are sufficiently rapid and the process likely occurs abiotically (Myers and Nealson, 1988a; Burdige et al., 1992). At the same time, for some reactions listed in **Table 2** evidence for their occurrence is equivocal; for example, recent studies in the suboxic waters of the Black Sea were unable to demonstrate the occurrence of denitrification coupled to Mn²⁺ oxidation (reaction 4 in **Table 2**; Schippers et al., 2005; Clement et al., 2009).

A final point to consider is that in many cases the organisms responsible for these anaerobic oxidation reactions have not yet been isolated and/or studied in pure culture. Exceptions to this do exist, that is, for denitrification coupled to Fe^{2+} oxidation (Straub et al., 1996) or to sulfide oxidation (Schulz and Jørgensen, 2001). However, such observations should not be

taken as evidence for lack of microbial involvement in any of these reactions, since the overwhelming majority of bacterial species thought to exist in nature have yet to be isolated in pure culture and grown in the lab. The same observation is also true of microbial phyla that have been identified in natural samples based solely on genetic sequences (16S rNA) isolated from these samples (Schloss and Handelsman, 2004).

5.09.2.3 Linkages between Chemolithotrophic and Organic Matter Remineralization Processes

Many of the chemolithotrophic reactions described above involve the oxidation of reduced end products of anoxic and suboxic respiration. In many sediments, this leads to the internal redox cycling of these elements, such that aerobic chemolithotropic or abiotic oxidative processes are coupled with suboxic, or anoxic, organic matter remineralization (Figure 1(a)). These processes can then be thought of as electron shuttles that bring O2 oxidation equivalents into deeper anoxic (O2-devoid) portions of the sediments (Jørgensen, 1982a; Burdige and Nealson, 1986; Aller, 1994b). Thus, in a net sense O₂ appears to be the oxidant of organic matter in the sediments, despite the fact that the actual remineralization occurs by anoxic, or suboxic, metabolism (also see related discussions in Aller, 1994a and Thamdrup and Canfield, 2000). While this coupling can occur through pore-water diffusive processes, more often it is mediated by macrofaunal activity, that is, bioturbation and bioirrigation, or macrophyte O2 input (see Sections 5.09.3, 5.09.4.1 and 5.09.4.5 for



Figure 1 (a) A schematic representation of the coupling between bacterial sulfate reduction and sulfide oxidation, such that in a net biogeochemical sense O_2 appears to be the electron acceptor used in the remineralization of sedimentary organic matter. (b) A more complex coupling of oxygen, manganese, sulfur, and organic carbon redox cycling that may occur in some marine sediments. Modified from Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton based on Aller, R.C., 1994b. The sedimentary Mn cycle in Long Island Sound: its role as intermediate oxidant and the influence of bioturbation, O_2 , and C_{org} flux on diagenetic reaction balances. Journal of Marine Research 52, 259–295 and Burdige, D.J., 1993. The biogeochemistry of manganese and iron reduction in marine sediments. Earth-Science Reviews 35, 249–284.

details). This coupling implies that redox-sensitive metals such as Mn and Fe can undergo repetitive oxidation–reduction cycles before ultimately being buried in sediments (see Section 5.09.7).

This type of sulfur redox cycling also has an important impact on the sulfur isotope record in sediments and sedimentary rocks (Canfield et al., 2005), as well as on the occurrence of carbonate dissolution or authigenic carbonate precipitation in estuarine and coastal sediments. In situations where there is no net sulfate reduction (i.e., sulfate reduction and sulfide oxidation are balanced as in **Figure 1(a)**), net CO₂ production occurs when the acid produced during sulfide oxidation protonates the bicarbonate produced by sulfate reduction, thus forming aqueous CO₂. This CO₂ neutralizes dissolved carbonate ion in the pore waters, according to,

$$CO_2 + H_2O + CO_3^{2-} \rightarrow 2HCO_3^{-}$$
 [2]

and lowers the saturation state of the pore waters with respect to the most soluble carbonate mineral present in the sediments. Once the pore waters become sufficiently undersaturated with respect to this phase, continued CO_2 production then drives the dissolution of sedimentary carbonate (Burdige and Zimmerman, 2002; Burdige et al., 2008).

At the other extreme are systems where net sulfate reduction occurs, or where the couplings between chemolithotrophic and suboxic or anoxic organic matter remineralization processes are more complex (see Figure 1(b) and discussions below). Here, bicarbonate (rather than aqueous CO_2) is the ultimate dissolved inorganic carbon end product. This then leads to an alkalinity buildup that eventually results in carbonate mineral precipitation (Sholkovitz, 1973; Suess, 1979; Aller and Rude, 1988; Canfield and Raiswell, 1991a; Aller et al., 1996).

This latter situation also illustrates how anaerobic chemolithotrophic may be coupled with aerobic chemolithotrophic processes as well as suboxic or anoxic organic matter remineralization (Figure 1(b)). Here, elements such as manganese and sulfur may actually undergo more complex internal redox cycling than that described above (i.e., compare Figures 1(a) and 1(b)). The occurrence of such interactions suggests that the sediment biogeochemical cycles of Mn, Fe, N, S, C, and O₂ may be closely coupled and more complex than previously thought. These processes not only have important implications on the internal dynamics of sediment processes, but may also impact net sediment (and perhaps global) budgets of these elements.

An important consequence of these interactions is that in many cases sediment oxygen uptake (also sometimes referred to as sediment oxygen demand) results not only from aerobic respiration but also from the oxidation of reduced compounds such as sulfide (dissolved and solid FeS and FeS₂), ammonium, Fe²⁺, and Mn²⁺ (Mackin and Swider, 1989; Sampou and Oviatt, 1991; Rysgaard et al., 1998; Berg et al., 2003; Glud et al., 2003). Furthermore, these processes are often tightly coupled, such that there is no significant net loss of reduced intermediates during this redox cycling by, for example, sulfide burial, benthic N₂ gas flux, or benthic fluxes of dissolved Mn²⁺. Under these circumstances, sediment oxygen uptake that occurs by a combination of aerobic respiration, chemolithotrophic reactions, and/or abiotic chemical oxidation still provides a reasonably good estimate of the total (depth-integrated) rate of sediment organic carbon oxidation (i.e., oxic, suboxic plus

anoxic remineralization; referred to here as R_{cox}), even though aerobic respiration is only a fraction of the total sediment oxygen uptake (Canfield et al., 1993a; Soetaert et al., 1996; Thamdrup et al., 2007). In situations where there is net loss of reduced metabolites (e.g., sulfide burial in sediments), these other processes can, however, also be quantified and incorporated (along with, e.g., sediment O₂ uptake measurements) into estimates of R_{cox} (Berelson et al., 1996; Thamdrup and Canfield, 2000). This may be particularly important in low salinity (or extremely organic-rich), estuarine sediments, where methane loss from the sediments (e.g., gas ebullition) uncouples sediment oxygen uptake and sediment organic carbon remineralization (Martens and Klump, 1984; Kelley et al., 1990; Middelburg et al., 1996a).

The discussion above pertains solely to sediments that are in steady state, based on the observation that under steadystate conditions, solute fluxes across the sediment-water interface are balanced by (i.e., are equal to) their depthintegrated rates of sediment production or consumption (see Burdige, 2006). However, the question of steady state can often be one of observational timescales, since estuarine and coastal sediments that clearly show seasonal variability in, for example, pore-water profiles or measured rates of sediment processes (i.e., the sediments are in non-steady-state on short timescales) may actually be in steady state on annual timescales (see the next section for more details as well as Klump and Martens, 1989; Martens et al., 1992; Soetaert et al., 1996; and Burdige and Zheng, 1998). One particular aspect of this behavior is that it can result in a temporal uncoupling of anoxic remineralization reactions and their corresponding (chemolithotrophic) re-oxidation reactions, leading to the seasonal in-growth and subsequent removal of reduced inorganic compounds in the sediments (Luther et al., 1991; Sampou and Oviatt, 1991; Therkildsen and Lomstein, 1993; Thamdrup et al., 1994a; Green and Aller, 1998; Thamdrup and Canfield, 2000).

In contrast, the non-steady-state conditions occurring in physically reworked margin sediments such as those on the Amazon shelf (Section 5.09.4.2) result in a more complex uncoupling of sediment O₂ uptake from the cycling of reduced metabolites. Here, estimates of sediment remineralization rates based on sediment oxygen uptake are likely to be lower limits of their true value (Aller et al., 1996; McKee et al., 2004), and such sediment systems with complex transport processes and non-steady-state diagenetic conditions require the independent determination of the rates of sediment processes to ascertain estimates of the total rates of sediment organic carbon remineralization.

5.09.2.4 Seasonality and Non-Steady-State Processes in Estuarine and Coastal Sediments

Seasonality in the rates of sediment biogeochemical processes are often observed in estuarine and coastal sediments in response to: annual changes in temperature; the composition, and reactivity of carbon deposited in the sediments; bottom water conditions; or the rates of benthic macrofaunal activity (which can themselves be, in part, a function of these other parameters; see the references cited below and, Klump and Martens, 1989; Sun et al., 1991; Therkildsen and Lomstein, 1993; Aller, 1994b; Roden et al., 1995; Canuel and Martens, 1996; Gerino et al., 1998; and Marvin-DiPasquale et al., 2003). For example, in temperate sediments an annual temperature changes of up to ~20 °C between winter and summer implies that remineralization rates may vary by up to a factor of ~8 over an annual cycle, based on the observed temperature dependence of sediment remineralization rates (Burdige, 2006). Phytoplankton production (e.g., Kemp et al., 1997) and growth of salt marsh plants (e.g., Hines, 2006) also tend to increase from spring to fall, and can result in, for example, seasonal changes in sediment POC deposition.

Seasonality in remineralization rates expresses itself in timevarying sediment pore-water profiles of remineralization end products (e.g., ammonium or phosphate). However, on these timescales diffusion damps out any seasonal periodicity in these profiles (Lasaga and Holland, 1976), and what is generally observed are smooth, asymptotic profiles whose concentrations simply grow in and out over the annual seasonal cycle (e.g., Aller, 1980a; Klump and Martens, 1989). Seasonal changes in sediment O₂ consumption also lead to the shoaling of the oxygen penetration depth during warmer months (Rysgaard et al., 1998; Glud et al., 2003; Glud, 2008; Martin, 2009). Likewise, in the organic-rich sediments of Cape Lookout Bight, NC (where sulfate reduction and methanogenesis dominate sediment organic matter remineralization), there is a seasonal shoaling in the depth of complete sulfate removal in the pore waters, from a depth of greater than ~20 cm during winter to ~10 cm (or less) during summer (Martens and Klump, 1984).

Despite non-steady-state seasonality in such sediments, under some circumstances it is possible to assume that porewater profiles are in quasi-steady-state over short time periods (Burdige, 2006). This then allows one to apply steady-state diagenetic models to individual, time-varying pore-water profiles to estimate instantaneous sediment remineralization rates for the times of sample collection (Martens and Klump, 1984; Soetaert et al., 1996). These are therefore snapshots of the timevarying rates at a particular point in time, and are not necessarily good estimates of the rates of sediment processes on more appropriate seasonal or annual timescales. Nevertheless, to determine seasonal- or annual-average rates, one can average these instantaneous rate estimates over a seasonal cycle (e.g., Burdige and Zheng, 1998). Alternately, one can obtain this annual average rate by fitting a temporal data set of, for example, individual pore-water profiles and/or rate measurement profiles with a time-dependent model for sediment diagenesis (Klump and Martens, 1989; Soetaert et al., 1996).

On shorter timescales, non-steady-state processes may affect biogeochemical processes in sediments in a number of different ways (see Sections 5.09.4.1, 5.09.4.3, and 5.09.4.5 for further details). Episodic (or event-driven) events such as flash floods, storms, or resuspension events (wind or current driven) can also have an impact on biogeochemical processes in sediments (Fanning et al., 1982; Chanton and Whiting, 1995; Cowan et al., 1996; Deflandre et al., 2002; Mucci et al., 2003; Tengberg et al., 2003; Kalnejais et al., 2007). For example, resuspension events which move reduced constituents from the sediments to the water column may be several times more effective at re-oxidizing these constituents than is re-oxidation mediated by diffusive benthic O_2 uptake (Glud, 2008).

5.09.3 Linkages of Biogeochemical Processes and Transport Processes and the Occurrence of Mixed Redox Conditions in Sediments

As noted above, transport processes such as bioturbation or bioirrigation (Section 5.09.4.1), permeability-driven porewater advection (Section 5.09.4.3), and solute release from macrophyte roots and rhizomes (Section 5.09.4.5) complicate the relatively simple picture of sediment biogeochemistry predicted by the biogeochemical zonation model. In terms of this solute transport, oxygen input to sediments is of particular importance, given the impact that it has on the two- and three-dimensional redox gradients that occur in sediments. In addition, a more general observation here is that these transport processes often lead to more active and dynamic cycling of redox-sensitive elements than would occur in their absence (i.e., in more strictly diffusive systems such as those described in Section 5.09.4.1). In particular, the linked chemolithotrophic and organic matter remineralization processes discussed in Section 5.09.2.3 are often an integral part of what is referred to here as mixed redox conditions. Furthermore, while all of the processes discussed here lead (in general) to more complex sediment redox cycling, differences in, for example, the spatial and temporal scales of sediment mixing, or the mechanism of oxidant input to the sediments, can lead to very different biogeochemical expressions of these forcings (Dellapenna et al., 1998; Hines, 2006). These considerations will be discussed later, in Sections 5.09.4.2 and 5.09.4.5.

When these transport processes are thought of in the context of the biogeochemical zonation model, the zonation predicted by this model may occur, for example, radially around cylindrical animal burrows (Figure 2(b)), or near-spherically around organic matter hot spots, for example, fecal pellets, in the sediments (Figure 2(c)), rather than as simple vertical zonation (Figure 2(a)). This transforms the sediment environment into



Figure 2 (a) The classical biogeochemical zonation that appears to occur in many marine sediments (e.g., based on the thermodynamic considerations in **Table 1**). Note the vertical redox zonation of electron acceptors that are used in these sediments. (b) A representation of the zonation of redox reactions in the microenvironment surrounding an irrigated macrofaunal burrow (or perhaps a macrophyte root). (c) A representation of the zonation of redox reactions in the microenvironment surrounding a fecal pellet in a marine sediment. From Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton. Modified from Aller, R.C., 1982. The effects of macrobenthos on chemical properties of marine sediments and overlying water. In: McCall, P.L., and Tevesz, M.J.S. (Eds.), Animal–Sediment Relations, the Biogenic Alteration of Sediments. Plenum Press, New York, NY, pp. 53–102.

one in which the full three-dimensional view of the sediments may need to be considered when attempting to study biogeochemical processes (e.g., Aller, 1988; Brandes and Devol, 1995).

However, even this conceptual model of mixed redox sediments is incomplete since particles and pore waters in these environments may actually be exposed to alternating oxic and anoxic condition as a result of these transport processes. In the simplest sense we think of these sediments as being periodically, or episodically, oxidized (or oxygenated), although the details of just how this occurs will vary among different environments. The timescales over which these redox oscillations occur are quite variable, ranging from minutes for bioirrigation to longer timescale characteristics of sediment mixing (also see discussions in Burdige, 2006). In some cases, these redox oscillations are asymmetrical in length, with anoxic conditions occurring for substantially longer times ($10 \times$ or greater) than oxic conditions (Aller, 1994a).

Oscillating (or mixed) oxic/anoxic sediment redox conditions occur, for example, when mobile deposit feeder oxygenate the sediments as they crawl through the sediments, when burrowing organisms pump fresh (oxygenated) seawater through their burrows (Kristensen, 2001), or when wave and/ or tidal action drives advective flow and exchange of anoxic pore water and oxic bottom water in permeable, sandy sediments (see Section 5.09.4.2 and Precht et al., 2004). Physical mixing (reworking) of sediments (Section 5.09.4.2) can also lead to the occurrence of such oscillating redox conditions, as can O_2 transport into sediments by the roots and rhizomes of macrophytes such as seagrasses (see Section 5.09.4.5 and e.g., Fredericksen and Glud, 2006).

Redox oscillations such as those described here also have the potential to greatly enhance organic carbon remineralization in sediments over that which might occur under more strict anoxic conditions, in spite of the often-times asymmetry between cycles or periods of anoxic versus oxic conditions (Aller, 1994a; Canfield, 1994; Hedges and Keil, 1995; Aller et al., 1996; Sun et al., 2002; Kristensen and Mikkelsen, 2003). The specific mechanisms by which this enhanced organic matter remineralization occurs under mixed redox conditions are not well characterized, although several possibilities have been discussed in the literature (see Burdige, 2007 for a recent review).

Based on all of these considerations, we see that mixed redox sediments are temporally varying, two- or three-dimensional sedimentary environments. Some degree of caution should therefore be taken in studying them assuming a one-dimensional, time-averaged perspective, as is often done in many sediment biogeochemical studies. A number of modeling approaches have begun to address these problems (Boudreau and Marinelli, 1994; Koretsky et al., 2002; Meysman et al., 2007; Miller et al., 2007; Cardenas et al., 2008), as have the use of two-dimensional optical imaging systems in laboratory mesocosm studies and in natural sediments *in situ* (e.g., Glud et al., 2001; Zhu et al., 2006).

5.09.4 The Classification of Estuarine and Coastal Sediments

Estuarine and coastal sediments can be classified into several functional types based on differences between the linkages of (and interactions between) biogeochemical processes and transport processes in the sediments. This approach has been used previously in the examination of continental shelf sediments (Aller, 1998, 2004; McKee et al., 2004), and will be adapted here for our discussion (also see related discussions in Middelburg et al., 2005).

In an attempt to put this classification scheme into a larger conceptual framework, I have also examined the distribution of these sediment types in terms of the physical energy of each sediment type and the average grain size of the sediments (**Figure 3**). This is far from a perfect approach since, for example, sediment grain size is a function of both sediment sources and the physical processes acting on the sediment system. Similarly, this figure shows that certain sediment types plot in the same region of this graph because they show strong

similarities based solely on these two parameters. Here, other controlling factors not incorporated into this plot lead to the divergence in the properties of these sediments. Nevertheless, in spite of the shortcomings of this approach, we believe that it will be useful in the discussion presented here.

For several sediment types, one key factor not explicitly indicated here is the differentiation between intertidal and subtidal sediments. Subtidal sediments, being completely submerged, can exchange solutes and dissolved gases with the overlying waters continuously. In contrast, intertidal sediments can only exchange solutes during times they are submerged, and gas exchange will vary between exchange directly with the atmosphere and exchange of dissolved gases with the overlying waters. The consequences of these differences are not well



Figure 3 (a) The approximate distribution of five of the sediments types discussed in Section 5.09.4 in energy vs. grain size space. (b) An overlay of the approximate distribution of intertidal sandflats and mudflats in energy vs. grain size space on the analogous distribution for other sediment types shown in (a). bt/bi=bioturbation/bioirrigation.

understood, although for reduced gases such as methane, direct exchange with the atmosphere bypasses their possible oxidation in the water column (see Section 5.09.9 for details).

5.09.4.1 Steadily Accumulating Sediments

These subtidal sediments contain fine-grained silts and muds, and the surface zone of the sediments can be bioturbated or bioirrigated. These sediments are often referred to as depositional systems, and have been well-studied for several decades (see Boudreau, 2000 and Aller, 2004 for historical perspectives). In fact, much of what we know about biogeochemical processes in sediments stems from studies of these types of sediments.

In sediments that are not bioturbated or bioirrigated, molecular diffusion, modified by the tortuosity of the sediment matrix (Burdige, 2006), is the major transport process for solutes. Particles are added to the sediments by both vertical and lateral transport from the overlying waters, and this process is referred to as sedimentation or sediment accumulation. The occurrence of sediment accumulation implies that particles at the sediment surface are eventually buried below the surface as new material continues to accumulate. Since the sediment– water interface is defined as the frame of reference for most geochemical studies, this then implies that there is a net downward flux (or burial) of material as a result of sediment accumulation.

In many estuarine and coastal sediments (both those that are depositional and nondepositional), the upper portion of the sediments is inhabited by higher organisms as well as bacteria. The activities of benthic macrofauna and meiofauna lead to the transport of solids and solutes in the surface sediments, and profoundly affect sediment redox conditions. These organisms therefore have a major impact on the sedimentary cycles of elements such as carbon, nitrogen, sulfur and redoxsensitive trace metals (e.g., Fe and Mn). Different aspects of benthic macrofaunal biogeochemistry from a more biological perspective are discussed in a number of monographs and review articles (Rhoads, 1974; Snelgrove and Butman, 1994; Kristensen, 2001; Schaffner et al., 2001; Kristensen et al., 2005; Dorgan et al., 2006; Levin et al., 2009), and the interested reader is urged to look in these references for this information.

There is a great variety in the types of macrofaunal activities (Figure 4). Some organisms simply crawl through the sediments or along the sediment surface in search of food, and mix sediment and sediment pore waters as a result of their activities. These organisms also ingest and subsequently defecate sediment materials. Such processes are commonly referred to as local mixing processes (Boudreau, 1986a; Meysman et al., 2005a) and more generally referred to as bioturbation. This process leads to the creation of a mixed (or bioturbated) layer in the surface sediments. (In some earlier works, the term 'bioturbation' is used in a much broader sense to describe all of the activities of benthic macrofauna in surface sediments. However, here (as in many more recent works) we will attempt to use this term in the more specific sense that is discussed in this section.) Bioturbation can, in principle, mix (and transport) both sediment particles and pore waters. However except in heavily bioturbated sediments (many of which are found in coastal and estuarine regions), rates of bioturbation are slow compared to molecular diffusion. Thus, bioturbation is largely of greatest impact in terms of particle mixing and transport in



Figure 4 A schematic representation of the major benthic macrofauna found in a typical coastal marine sediment. The organisms shown here include the deposit feeding bivalves *Nucula annulata* (A) and *Yoldia limulata* (C), the suspension feeding bivalve *Mulnia lateralis* (D), the mobile burrowing polychaete *Nephtys incisa* (B), the sedentary polychaetes *Spiochaetopterus oculatus* (E) and *Clymenella* sp. (F), and the deposit feeding gastropod *Nassarius trivitarus* (G). From Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton, based on an original figure in Aller, R.C., 1980a. Diagenetic processes near the sediment–water interface of Long Island Sound, I. Decomposition and nutrient element geochemistry (S, N, P). Advances in Geophysics 22, 235–348.

surface sediments. An exception to this may be bioturbation by benthic meiofauna, for example, nematodes or juvenile polychaetes, in addition to, or instead of, benthic macrofauna (Berg et al., 2001). Meiofaunal bioturbation may differentially affect solutes versus solids, because of the size of these organisms, and the fact that at the relatively small (submillimeter) scales at which these organisms operate it may be easier to move pore waters rather than solids (also see Aller and Aller, 1992).

Other deposit-feeding macrofauna live in fixed sediment burrows and move material from depths a few centimeters below the sediment surface to the surface, forming burrow mounds. This process is generally referred to as conveyor-belt transport (e.g., Boudreau, 1986b; Robbins, 1986). Reverse conveyor-belt feeders carry out a similar type of transport in the opposite direction, by taking surface sediment and moving it down to some depth in the sediments (e.g., Smith et al., 2001). Both of these types of bioturbation are also referred to as nonlocal mixing (Boudreau, 1986b; Meysman et al., 2003).

Finally, a last group of macrofaunal organisms are watercolumn suspension feeders and tubeworms. These organisms build sediment burrows and tubes, and flush these burrows with overlying bottom waters to cleanse the burrows of potentially harmful solutes that can accumulate in the burrow due to, for example, diffusion into the burrow from the surrounding sediments. This process is generally referred to as bioirrigation, in that it does not affect transport of the surrounding sediment particles, but does enhance the exchange of pore waters and bottom waters.

The impact of bioirrigation on sediment biogeochemistry results from the fact that the occurrence of macrofaunal tubes and burrows leads to a complex three-dimensional geometry in the zone of irrigation that significantly alters the spatial (and temporal) distribution of reactions occurring in these sediments. Bioirrigation of sediments creates diffusive sources and sinks for pore-water solutes within portions of the sediments that would otherwise be isolated from exchange with the overlying waters by diffusive processes alone. Bioirrigation is therefore also a nonlocal exchange process (Aller, 1980c, 2001; Boudreau, 1984; Grigg et al., 2005). Because burrows are flushed with oxygenated bottom water, bioirrigation enhances sediment oxygen input, thus altering sediment redox conditions, promoting redox oscillations, and allowing for the occurrence of a wide range of coupled oxidative and reductive processes associated with organic matter remineralization (see discussions in Sections 5.09.2.3 and 5.09.3, as well as in Aller et al., 2001 and Glud, 2008). The sediments immediately surrounding animal burrows and tubes are often sites of elevated microbial populations and enhanced activity (Aller, 1988). As a result, macrofaunal processes can have both direct as well as indirect effects on the rates of sediment organic matter remineralization (also see Glud et al., 2003).

Irrigation rates vary among different species of benthic macrofauna, and irrigation is often times a discontinuous process (i.e., there are periods of active irrigation followed by periods of rest). This can then lead to transient (non-steadystate or time-dependent) responses in burrow-water and porewater concentrations near the burrows. The impact this has on sediment processes and pore-water concentrations can vary depending on the type of sediment (e.g., sands vs. muds) along with the reaction kinetics (e.g., zero vs. first-order kinetics) affecting the production/consumption of a given solute (Boudreau and Marinelli, 1994; Marinelli and Boudreau, 1996; Meysman et al., 2005b, 2006).

A variety of factors control the species distribution and abundance of benthic macrofauna in sediments (also see more extensive reviews in Rhoads, 1974; Snelgrove and Butman, 1994; Herman et al., 1999; and Schaffner et al., 2001). As a general controlling factor bottom-water oxygen appears to be quite important, at least in terms of the presence/absence of benthic macrofauna. Long-term bottomwater O₂ concentrations that are less than ~5 µM generally lead to the complete absence of benthic macrofauna in the underlying sediments (Rhoads, 1974; Savrda et al., 1984; Levin et al., 2009). Studies also show that bottom water [O2] levels less than ~60-70 µM begin to adversely impact benthic communities (Diaz and Rosenberg, 1995; Levin et al., 2009). (Note that in the literature a wide range of units are still used to express oxygen concentrations. For comparison, an oxygen concentration (at STP) of $1 \text{ ml} \cdot l^{-1}$ is equivalent to 44.6 μ M, so 5 μ M is equivalent to ~0.1 ml · l⁻¹. In discussing low oxygen environments, the term hypoxia is also often used, and it is generally defined as oxygen concentrations below $2 \text{ ml} \cdot l^{-1}$, or ~90 µM (Diaz and Rosenberg, 1995.))

In contrast though, many estuarine and coastal sediments are subject to episodic, or seasonal, hypoxia (or anoxia) in the bottom waters. Its effect on the distribution, abundance, and activity of benthic macrofauna is a complex function of the level, duration, and physical extent of low oxygen conditions in the bottom waters, and the impact that this has on the geochemistry of the bottom sediments (Middelburg and Levin, 2009). It is difficult to generalize these effects, and Levin et al. (2009) summarized the results from studies of benthic macrofaunal responses to bottom-water hypoxia (and anoxia) across a wide range of estuarine and coastal settings, including Chesapeake Bay, the Baltic Sea, and the northern Gulf of Mexico.

Other factors that appear to control the occurrence of benthic macrofauna include sediment type, food source, sedimentation rate, and physical disturbance (again see any of the review articles cited above for details). In the latter cases, areas of high physical disturbance (in terms of both frequency and intensity), as well as areas of high sediment deposition generally show very low abundances of macrofauna, with bacteria dominating the total benthic biomass (e.g., Aller and Aller, 2004). Of the macrofauna that are present in these sediments, small, surface dwelling polychaetes and tube-dwelling amphipods are generally most important (also see Rhoads, 1974; Snelgrove and Butman, 1994; Schaffner et al., 2001).

Benthic macrofauna tend to inhabit the upper ~10–20 cm of sediment, and based on radio-geochemical analyses Boudreau (1998) has suggested that worldwide, the mixed layer depth in marine sediments has a mean value of ~10 (\pm 4.5) cm. The average thickness of the bioirrigation zone may be of similar magnitude (see Burdige, 2006 for a summary), although some studies do suggest that irrigation may occur to depths of ~30–50 cm (e.g., Martin et al., 2007; Jørgensen and Parkers, 2010). At the same time, many of the factors outlined in the previous paragraph may lead to bioturbation or bioirrigation depths that are shallower than 10 cm (e.g., Schaffner et al., 2001; Smith and Rabouille, 2004).

5.09.4.2 Sediments Subject to Extensive Physical Reworking

These sediments are highly mobile deposits, in which physical processes both resuspend sediments and redistribute them horizontally. As a result, the sediments appear to act as a massive, non-steady-state suboxic fluidized bed reactor (see Section 5.09.7 for details). The biogeochemical consequences of this type of large-scale sediment mixing contrast with the more small-scale, mixed redox conditions (oxic \rightarrow anoxic) associated with biological mixing, that is, bioturbation (see discussions in Dellapenna et al., 1998).

This type of physical reworking of sediments has been best studied in muddy continental margin deltas such as the Amazon continental shelf or the Gulf of Papua/Fly River complex (Papua New Guinea), as well as in mobile tropical mud belts such as the Amazon-Guianas mud belt of coastal French Guiana (Alongi, 1995; Aller, 1998; Zhu et al., 2002; Aller et al., 2004a, 2008; Corbett et al., 2004; McKee et al., 2004). On the Amazon shelf, the upper ~0.5–2 m of sediment is eroded by shelf currents and tides, re-oxidized by brief resuspension into the bottom waters, and then re-deposited (McKee et al., 2004). This physical reworking occurs on daily to seasonal timescales (also see DeMaster and Aller, 2001).

An important aspect of these continental margin settings is that they occur in association with major rivers that have relatively high freshwater discharge rates. As a result, estuarine processes, which would normally take place in a physically confined estuary, take place offshore, on the adjacent continental shelf. These large rivers also have high suspended sediment loads, and the river-dominated ocean margins receive large inputs of both lithogenic and biogenic material (McKee et al., 2004; Bianchi and Allison, 2009). These continental margin settings therefore represent the major repositories for the deposition and burial of the material transported by rivers to the oceans (also see Berner and Berner, 1996).

The physical reworking of these sediments has a profound impact on both sediment carbon cycling (i.e., preservation vs. remineralization) and nitrogen cycling on both local and global scales (see the references cited above, as well as Burdige, 2005). These processes also impact the oceanic cycles of elements such as U, F, and K, by, for example, promoting the occurrence of authigenic mineral formation, or reverse weathering (e.g., Michalopoulos and Aller, 2004).

Analogous types of physical reworking of sediments also appear to occur in estuarine environments, albeit on smaller spatial scales (Dellapenna et al., 1998; Schaffner et al., 2001; Galler and Allison, 2008). They may occur, for example, in the maximum turbidity zone of tidal estuaries, where there are finegrained materials and sufficient energy for physical reworking of the surface fluid muds. While the biogeochemical significance of this type of physical reworking in estuarine sediments is not as well documented as it is in deltaic or offshore continental margin mobile muds, results to date suggest some degree of commonality (Abril et al., 1999, 2000; Robert et al., 2004; Arzayus and Canuel, 2005; Palomo and Canuel, 2010).

5.09.4.3 High Permeability, Sandy Sediments

Because of their grain size these sediments have the capacity to allow advective flow through them in response to an external pressure gradient (e.g., Lerman, 1979). This phenomenon is referred to as sediment permeability. In contrast, more finegrained silts and muds have a low permeability, and allow for little advective flow; diffusion is therefore the dominant nonbiological transport process for solutes. High-permeability, sandy sediments occur on ~70% of all continental shelves globally, often as nonaccumulating, well-sorted relict sands that were deposited during the low stand of sea level during the last Ice Age (Emery, 1968). Sandy sediments also occur in some estuarine settings on-, and just off-shore, most beaches (see Sections 5.09.4.4–6), and on shallow water carbonate platforms such as the Bahamas Bank.

The near-seabed pressure gradients that drive advective flow in these sediments have several possible sources. In shallow waters and intertidal regions, pressure gradients may result from wave and tide action (Shum and Sundby, 1996; Falter and Sansone, 2000) that can be thought of as pumping overlying waters into the sediments. Tidal forcing such as that described here has also been examined in recent years in terms of submarine groundwater discharge to the oceans, based on the recognition that some fraction of what is referred to as groundwater being discharged into the coastal ocean is actually seawater that has been recirculated through the underlying sediments (e.g., Martin et al., 2007). By analogy with a surface estuary, Moore (1999) has referred to the reaction zone of fresh (terrestrial) ground water and recirculated seawater as a subterranean estuary. There is increasing recognition that reactions occurring in this subterranean estuary may be of geochemical significance (e.g., Charette and Sholkovitz, 2006; Johannesson and Burdige, 2007; Roy et al., 2010), although sampling difficulties coupled with the complexity of water circulation in these systems have made quantification of these processes difficult (Burnett et al., 2006).

Advective exchange in permeable sediments also occurs though interactions between bottom currents and surface structures, for example, physical surface roughness, sand ripples, and biogenic structures such as animal mounds (e.g., Huettel and Webster, 2001). Larger structures such as seagrass canopies (Section 5.09.4.5) can similarly impact bottom-water flow and lead to near-seabed pressure gradients (Jumars et al., 2001; Ghisalberti and Nepf, 2002; Koch et al., 2006). These latter interactions are complex and not fully understood, although past studies suggest that the most important factors include canopy architecture (shoot density and leaf length). Also important here is the type of flow (e.g., unidirectional tidal currents vs. wave-dominated flow) and its impact on the motion of individual leaves (also see Koch and Gust, 1999; Peterson et al., 2004). Wind-driven Langmuir cells that extend over the full depth of shallow water columns may similarly result in oscillatory bottom water flow that can generate large (and changing) areas of increased and decreased pressure on the seafloor (Gargett et al., 2004; Tejada-Martinez and Grosch, 2007). This then represents another physical process that may drive advective pore-water exchange across the sediment-water interface in shallow environments (see related discussion in Dierssen et al., 2009).

In examining the relative importance of different transport processes in sediments, it has been suggested that pressuredriven pore-water advection is likely to be important when permeabilities (*k*) exceed $\sim 1-2 \times 10^{-12}$ m² (see Figure 5, and Huettel and Gust, 1992 and Forster et al., 2003). Based on a



Figure 5 A schematic representation of the depth dependence of various pore-water transport processes (diffusion, bioturbation, and physical advection) as a function of sediment permeability. Note that there is some overlap in these three regions, such that diffusion and advection can also occur in the region where bioturbation occurs. Modified from Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton based on Huettel, M., and Webster, I.T., 2001. Porewater flow in permeable sediments. In: Boudreau, B.P., and Jørgensen, B.B. (Eds.), The Benthic Boundary Layer. Oxford University Press, Oxford, pp. 144–179.

log–log plot of measured permeability versus sediment grain size (data complied in Burdige, 2006), this cutoff occurs at a median grain size of ~70–80 µm, that is, sediments that are larger than a very fine sand (also see Lerman, 1979; Bennett et al., 1990). Using a slightly different data set, Wilson et al. (2008) observed what appears to be distinct trend lines for these parameters for continental margin and coastal sediments. Here *k* values greater than 2×10^{-12} m² occur for mean grain sizes greater than ~40 µm for coastal sediments and ~125 µm for continental margin sediments, values that span the grain size value estimated above. There is also some overlap in the regions illustrated in **Figure 5**, and many sandy sediments are also inhabited by benthic macrofauna, even at relatively high permeabilities (see below for details).

For typical bottom-water flows and surface structures, laboratory flume studies suggest that bottom waters can be forced several centimeters into the sediment, simultaneously drawing pore water from depths >~10 cm up to the sediment surface (Huettel et al., 1996). Field observations are generally consistent with these laboratory results (Marinelli et al., 1998; Rasheed et al., 2004; Reimers et al., 2004; Jahnke et al., 2005; Burdige et al., 2008). Changes in flow velocity (both magnitude and frequency) can have differing effects on oxygen uptake and the depth of oxygen penetration in permeable sediments, depending on the type of flow as well as the nature of the surface topography (Forster et al., 1996; Ziebis et al., 1996b; Berninger and Huettel, 1997; Precht et al., 2004).

In flume studies where the flow velocity was fixed (in a given experiment) but varied among different studies, oxygen penetration depth showed virtually no change with increasing flow velocity when the sediment surface was smooth (Forster

et al., 1996; Ziebis et al., 1996b). Such observations may be relevant to the effect of changing tidal flow velocities on advective pore-water exchange over tidal cycles, although this effect has not yet been examined *in situ*.

In contrast, flume studies in which short period (~1 s) water waves pass over a sediment surface rapidly lead to the formation of sand ripples which subsequently creates "an undulating oxic-anoxic boundary within the upper sediment layer, mirroring the topographical features of the sediment bed" (Precht et al., 2004). Here pore-water flow is also more complex, with oxygenated overlying water being forced into the sediments in the ripple troughs and deep anoxic pore water being upwelled to the surface under the ripple crests (e.g., Huettel et al., 1996). Similar effects also appear to impact processes such as denitrification (see Section 5.09.6) and the redox cycling of other solutes such as Fe²⁺ or Mn²⁺ (Huettel et al., 1998). Oxygenated bottom water that is advected into permeable sediments coming in contact with reduced pore waters should result in the types of linkages between oxic and anoxic organic matter remineralization and chemolithotrophic reactions associated with other mixed redox sediments (see Section 5.09.3).

Permeable sediments act as efficient sand filters because of these advective processes, analogous to those found in sewage treatment plants (Boudreau et al., 2001). Recent work has therefore shown that they are significantly more biogeochemically active than previously thought (also see Jahnke, 2005 for a summary). Rates of sediment oxygen uptake are greatly enhanced in permeable sediments due to increased oxygen input from bottom waters, coupled with the advective input of suspended organic particles (e.g., phytoplankton debris or bacteria) from the water column into the sediments, where they can subsequently undergo remineralization (Forster et al., 1996; Ziebis et al., 1996b; Huettel and Rusch, 2000; Rusch and Huettel, 2000).

Permeable sediments generally have very low organic carbon content, due, in part, to their nondepositional nature. However, the material that is input into the sediments appears to be relatively reactive, as compared to that found in many fine-grained depositional sediments (Boudreau et al., 2001). In part, this occurs because the input of reactive organic matter to the sediments is apparently not diluted by the accumulation of aged, and thus, less reactive, organic matter (Boudreau et al., 2001). Therefore, the low standing stock of organic carbon (TOC) in permeable sediments, and other sediment nutrients in general, may be controlled more by rapid turnover than by low input.

Finally, many permeable sediments occur in shallow waters, and recent estimates suggest that ~30% of the continental shelf seafloor receives sufficient light to support net primary production (Jahnke, 2005; Gattuso et al., 2006). Although estimates of benthic primary production in shallow water sands are subject to large uncertainties (Boudreau et al., 2001; Gattuso et al., 2006), in many cases, they appear to be comparable to rates of water-column primary productivity (Jahnke et al., 2000). Many permeable sediments are also inhabited by benthic macrofauna (Huettel and Gust, 1992; Ziebis et al., 1996a; D'Andrea et al., 2002; Meysman et al., 2005b). Their impact on biogeochemical processes in these sediments is several fold. First, the biogenic structures built by the macrofauna help create the near-seabed pressure gradients that drive advective pore-water exchange in these sediments. The activity of benthic macrofauna in permeable sediments also adds an additional level of complexity to solute transport in these sediments (Meysman et al., 2006, 2007), which may result in synergistic effects that enhance remineralization and solute fluxes across the sediment–water interface in these settings (e.g., Volkenborn et al., 2007).

5.09.4.4 Bypass Zone Sediments

In one sense, these sediments may be thought of as being transitional between steady accumulating, bioturbated/bioirrigated silts and muds and high-permeability sands (Figure 3(a)). In these settings, physical energy (e.g., near bottom tidal current, and wind-wave action) is great enough to winnow finegrained materials from the sediments (and advect this material out of the system as suspended sediments), but remains just above (or slightly below) critical levels that will cause net seabed erosion (e.g., Wright et al., 1997). These sediments often support active communities of bioturbating and bioirrigating macrofauna (Dellapenna et al., 1998; Schaffner et al., 2001), and organic matter remineralization processes in these sediments are strongly controlled by the mixed redox conditions that develop in the sediments as a result of this macrofaunal activity. Advective pore-water exchange is likely to occur in these sediments, given their grain size and, because they are bioturbated and bioirrigated, the presence of biogenic structures on the sediment surface that can contribute to nearseabed pressure gradients. These sediments may be nondepositional, but often have low rates of sediment accumulation as compared to other more typical estuarine settings. A well-studied sediment system acting as this type of a bypass zone is the bay stem plains in the lower Chesapeake Bay (Schaffner et al., 1987; Wright et al., 1987, 1997; Dellapenna et al., 1998; Schaffner et al., 2001).

As the physical energy in bypass sediments increases, the sediment will become increasingly inhospitable to macrofauna, and the sediments will move from one of little or no net deposition, to an erosional system. On the continental shelf, the processes that lead to highly mobile physically reworked sediments (Section 5.09.4.2) can also lead to erosional regions in which the surface sediments are scoured away and deeply buried materials are re-exposed (Kuehl et al., 1986; Sommerfield et al., 1995). This can then result in the nonsteady-state release and re-oxidation of reduced metabolites such as CH_4 , along with the oxic remineralization of organic carbon that had previously escaped remineralization and been buried in these sediments (Blair and Aller, 1995).

5.09.4.5 Macrophyte-Dominated Sediments

This group (also sometimes referred to as vegetated sediments) includes salt marshes, mangrove forests, and seagrass-dominated sediments. While the vegetations here include both emergent and submerged macrophytes, there is sufficient commonality among these sediment systems to consider them together.

Salt marshes are intertidal wetlands inhabited by salt tolerant grasses (nonwoody vascular plants), including species of *Spartina* (e.g., *S. alterniflora* and *S. patens*), *Juncus*, and *Salicornia*. They are common in low-energy estuarine and coastal settings in temperate, boreal, and arctic regions. In tropical and subtropical settings, salt marshes are replaced by mangrove forests, which contain salt-tolerant trees, shrubs, and other plants. In contrast, seagrasses are submerged vascular plants (angiosperms) that are found in all estuarine and coastal areas except in Antarctica. Most seagrasses live in subtidal environments, with their depth distribution controlled by light limitations (Hemminga and Duarte, 2000); on average, they require ~11% or more of surface irradiance, which limits them to a water depth of ~40 m or shallower in very clear ocean waters. Seagrasses can live in muddy to sandy sediments, limited here by sediment stability or current velocities.

Macrophyte-dominated sediments are also often extensively bioturbated. In salt marshes and mangrove forests burrowing crabs tend to dominate (Alongi, 1998), while seagrass meadows generally support a broader range of benthic macrofaunal species (Hemminga and Duarte, 2000).

In intertidal salt marsh and mangrove sediments, tidal forcing can play an important role in regulating biogeochemical processes in the sediments, and ultimately solute fluxes across the sediment–water interface (Lacerda et al., 1999; Bouillon et al., 2007; Taillefert et al., 2007). The factors controlling this subsurface flow are complex, and, for example, rising and falling tides can induce both vertical as well as horizontal subsurface flow in these sediments based on factors such as marsh (or forest) topography and sedimentology (i.e., grain size), as well as the importance of fresh groundwater flow from upland sources (also see Mitsch and Gosselink, 2000; Tobias et al., 2001; Jahnke et al., 2003).

In terms of their interaction with the surrounding physical environment, the seagrass canopy dissipates the energy of local currents and promotes the deposition of organic detritus and fine-grained inorganic suspended material from the water column to the underlying sediments (Hemminga and Duarte, 2000; Koch et al., 2006). Salt marsh plants and mangrove trees have a similar effect on particulate deposition from the tidal or creek waters that flow through (or over) the marsh or forest (Mitsch and Gosselink, 2000; Kristensen et al., 2008).

Salt marsh, seagrass, and mangrove ecosystems are highly productive, and macrophyte-derived organic matter plays an important role in organic carbon remineralization in the underlying sediments (Holmer et al., 2001; Jones et al., 2003; Kaldy et al., 2006; Kristensen et al., 2008). At the same time, the external input of organic detritus (due to the processes described in the previous paragraph) represents a variable, but sometimes large, input of organic carbon to macrophyte sediments, that can then similarly contribute to sediment organic carbon remineralization (Papadimitriou et al., 2005; Bouillon and Boschker, 2006; Kaldy et al., 2006). Recent studies have re-examined organic carbon budgets in vegetated sediments (Chmura et al., 2003; Duarte et al., 2005b, 2010; Kristensen et al., 2008) and have suggested that organic carbon burial in these sediments is comparable in magnitude to estimates of organic carbon burial in nonvegetated, subaqueous deltaic and continental margin sediments (which have previously been suggested to be the primary sites of organic carbon burial in the oceans; Berner, 1989; Hedges and Keil, 1995). Salt marshes and mangrove forests may also represent an intertidal pump in which globally significant amounts of dissolved inorganic carbon (and to a lesser extent dissolved organic carbon, or, DOC) derived from the remineralization

of plant-derived organic matter are exported to the coastal ocean (Bouillon et al., 2008; Jahnke, 2009).

Salt marsh grasses, mangroves, and seagrasses all actively transport atmospheric or photosynthetically produced O₂ in the plants to their belowground tissues, which then results in either active or passive oxygen transport from the roots and rhizomes into the surrounding sediments (see reviews, and references therein, in Alongi, 1998; Canfield et al., 2005; Lovell, 2005; and Borum et al., 2006). The impact of this oxygen transport away from belowground plant tissues can be viewed in the context of the model for mixed (or oscillating) redox conditions in Section 5.09.3, and will also have impacts that are somewhat analogous to those which occur as a result of macrofaunal bioirrigation (also see Vale and Sundby, 1998). However, the contrast between oxidant (O_2) input from roots and rhizomes versus oxidant input (e.g., Mn and Fe oxides) that occurs by physical mixing (local and nonlocal bioturbation) can, under some circumstances, result in the predominance of differing types of redox cycling being linked to O₂ reduction (i.e., sulfur redox cycling in the latter case and iron redox cycling in the former case; e.g., Hines, 2006). These effects are not universal, though, depending on, for example, the iron content of the sediment or the presence of bioturbating macrofauna in a macrophyte sediment.

Many macrophytes live in fine-grained, high TOC sediments. Because sulfate reduction is often an important organic carbon remineralization process in these sediments, sulfide levels in the sediment pore waters are often high, and oxygen input by the plants into the surrounding rhizosphere is thought to protect the plant from sulfide and other phytotoxins such as reduced metals that are also commonly found in these anoxic sediments (Alongi, 1998; Lee and Dunton, 2000; Lovell, 2005; Fredericksen and Glud, 2006; Hines, 2006; Marbá et al., 2006; Miller et al., 2007). Many such plants also have a thin iron oxide coating (generally referred to as an iron plaque) that forms on the root surface as a result of the oxidation of Fe²⁺ produced in the surrounding anoxic sediments (St-Cyr et al., 1993; Sundby et al., 1998; Povidisa et al., 2009). Oxygen input in macrophyte sediments therefore plays an important role in not only metal but also sulfur, nitrogen, and carbon redox cycling (also see Burdige et al., 2010; Holmer et al., 1999; Ku et al., 1999; Kristensen et al., 2000; Sundby et al., 2003; Hu and Burdige, 2008; Hyun et al., 2009).

The region surrounding macrophyte roots and rhizomes may also be a region of enhanced microbial activity (see general reviews in Alongi, 1998; Duarte et al., 2005a; and Hines, 2006), again as is seen around macrofaunal burrows and tubes (Aller, 1988). In part, this may be due to the release of reactive DOM from the plant roots into the surrounding sediments, which supports both aerobic and anaerobic (sulfate reducing) remineralization in the sediments (Wetzel and Penhale, 1979; Moriarty et al., 1986; Ziegler and Benner, 1999; Hines, 2006; Kaldy et al., 2006; Kristensen and Alongi, 2006). More specifically, studies of salt marsh plants have shown that plant roots release both acetate and ethanol (Mendelsson et al., 1981; Hines et al., 1994; Hines, 2006), which are fermentation end products (presumably produced by anaerobic metabolism in the plants) and excellent substrates for both aerobic and sulfate-reducing bacteria.

Many aspects of the biogeochemistry of vegetated sediments are inherently time dependent due to the diel variability in belowground input of photosynthetically produced O_2 , or temporal variability in the release of reactive DOC from roots (Fredericksen and Glud, 2006; Hebert et al., 2007; Miller et al., 2007). The factors that control the expression of this variability in, for example, pore-water sulfide profiles, are complex and in seagrass sediments may be a function of the seagrass density (Lee and Dunton, 2000). However, small-scale spatial variability may obscure temporal variability (Hebert et al., 2007), and developing appropriate sampling strategies that overcome these hurdles is an important consideration here (Morse et al., 2003).

For seagrasses living in more organic-poor carbonate sediments in tropical and subtropical regions, belowground O₂ input may also play an important role in P acquisition by the plants. In these setting, phosphorus is often a critical limiting nutrient due, in part, to the strong tendency of phosphate to adsorb to carbonate minerals. O2 input by the plants into the sediments results in the dissolution of sedimentary carbonates as a result of CO₂ production due to sediment aerobic respiration (Burdige and Zimmerman, 2002), and this seagrassmediated carbonate dissolution then liberates carbonatebound phosphate (Jensen et al., 1998). This process also has important positive feedback characteristics since seagrass growth and the associated belowground O₂ input enhances sediment carbonate dissolution, which then further liberates phosphate from the sediments and promotes additional seagrass growth and carbonate dissolution (see Burdige et al., 2008 for further discussions).

5.09.4.6 Intertidal Mudflats and Sandflats

These sediments show a range of characteristics that have similarities to the sediment types discussed in the previous sections. As with salt marsh and mangrove sediments, tidal processes play an important role in the dynamics of these macrophytefree intertidal sediments (Alongi, 1998). The fact that intertidal mudflats and sandflats are alternately exposed and submerged (versus permanently submerged subtidal sediments) also has an impact on the exchange of solutes and gases from these sediments (see Section 5.09.4).

High-energy sandy beaches show some commonality with subtidal high-permeability sediments (Section 5.09.4.3), and they are often discussed together (e.g., Boudreau et al., 2001; Huettel and Webster, 2001). In a similar fashion, intertidal mud banks are often discussed in the context of biogeochemical processes associated with physically reworked tropical mud belt sediments (e.g., Alongi, 1991; Aller et al., 2004b).

In contrast to these high-energy environments, tidal sandflats and intertidal mudflats are lower-energy environments that are usually actively bioturbated and/or bioirrigated, oftentimes by economically valuable organisms such as soft-shell or hard-shell clams (D'Andrea et al., 2002; Green et al., 2004; Hyun et al., 2009). As sediment grain size increases, porewater advective exchange becomes increasingly important in these sediments. The surfaces of mudflats and sandflats are also generally covered by photosynthetic microalgal or macroalgal mats (e.g., Stephens et al., 2003; Sundback and McGlather, 2005). The presence of this microphytobenthos can often help stabilize the sediments against resuspension, by producing a mucilaginous biofilm (Miller et al., 1996; Decho, 2000).

5.09.5 Sediment Oxygen Consumption

In most estuarine and coastal sediments the large flux of reactive organic carbon to the sediments results in high potential rates of sediment O_2 consumption relative to O_2 transport into the sediments by processes such as molecular diffusion. Oxygen in the pore waters therefore generally goes to zero within a centimeter or less below the sediment–water interface, oftentimes in upper few millimeters (see summaries in Burdige, 2006 and Glud, 2008). In contrast, nonlocal transport processes (e.g., bioirrigation or macrophyte root and rhizome oxygen input) can input oxygen into deeper portions of many estuarine and coastal sediments. Nevertheless, the tight coupling of O_2 input and consumption results in the near-zero to zero O_2 concentrations in the bulk of the pore waters, except in a region immediately adjacent to the particular nonlocal source (Berg et al., 2003; König et al., 2005; Fredericksen and Glud, 2006; Glud, 2008).

Estimates of aerobic respiration as a percentage of total sediment organic carbon remineralization (R_{cox}) in estuarine and coastal sediments range from 0 to ~60% with an average of ~28% (see Burdige, 2006 for a summary). Where macrofauna are present, their respiration (vs. microbial respiration) is ~15–20% of R_{cox} (for a review, see Herman et al., 1999). A direct comparison of these two percentages is equivocal given the way these results are presented in the literature, although they do suggest that when macrofauna are present, they could account for perhaps half of the total sediment aerobic respiration.

As discussed in Section 5.09.2.3, much of the oxygen uptake in estuarine and coastal sediments occurs at the expense of the oxidation of reduced sulfide, ammonium, Mn^{2+} , and Fe^{2+} (Mackin and Swider, 1989; Sampou and

Oviatt, 1991; Rysgaard et al., 1998; Berg et al., 2003; Glud et al., 2003). Where quantified, these re-oxidation reactions represent ~60–70% of the total sediment oxygen uptake (or TOU). The remaining oxygen uptake occurs by either macrofaunal or microbial respiration, since meiofaunal respiration is generally insignificant (Middelburg et al., 2005). Canfield et al. (2005) estimated that macrofaunal respiration accounts for ~5–30% of TOU. This estimate of macrofaunal respiration as a percentage of TOU does not appear to be inconsistent with the discussion above of macrofaunal respiration as a percentage of $R_{\rm coxy}$ although a direct comparison of these results is not possible.

Oxygen consumption during the oxidation of these reduced solutes then results in low concentrations of these solutes near the sediment–water interface due to their oxidative removal (see Figure 6 and, e.g., Burdige, 1993 and Martin, 2009). It impacts benthic fluxes of these solutes to the overlying water column and also has a larger-scale effect on sediment trace metal cycling (see Section 5.09.10 for details). The Mn and Fe oxides produced by such reactions are also important since the mixing of the oxides into deeper portions of the sediments can drive a number of other biogeochemically important processes (see Sections 5.09.2.3).

5.09.6 Organotrophic Denitrification, Anammox and Anoxic Nitrification

Previous estimates (Burdige, 2006) suggest that organotrophic denitrification accounts for \sim 10–20% of $R_{\rm cox}$ in estuarine and coastal sediments. These estimates have generally been based on diagenetic models applied to pore-water nitrate profiles,



Figure 6 Pore-water profiles (collected 10/94) for dissolved Mn^{2+} , total dissolved Fe (\approx [Fe²⁺]), total dissolved sulfide and sulfate at a site in the mesohaline region of Chesapeake Bay. Note that the manganese profile has an expanded depth scale as compared to the other three profiles. The curves through the data are meant to better illustrate the trends in the data; they are not model fits and are not meant to imply any functional relationships. Given the importance of sulfate reduction in these sediments (see sulfate profile and data in Marvin-DiPasquale and Capone, 1998), Mn and Fe reduction here is likely coupled to sulfide oxidation, although this has not been definitively demonstrated (also see related discussions in Burdige, 1993). The accumulation of iron in the pore waters is caused by a balance between production from the reduction of iron oxides, and consumption during manganese reduction (see reaction 9 in **Table 2**) and precipitation of iron sulfides. Although sulfate reduced, as evidenced here by the draw down of all iron in the pore waters. Finally, note that removal of Mn^{2+} from the pore waters occurs at deeper sediment depths (>-10 cm) than does Fe removal, presumably due to the requirement of higher pore-water sulfide and/or alkalinity levels (found deeper in the pore waters) for the precipitation of the more soluble Mn carbonate and/or sulfide phases (Holdren et al., 1975; Suess, 1979; Aller, 1980b; Middelburg et al., 1987). Redrawn from Skrabal, S.A., Donat, J.R., and Burdige, D.J., 2000. Pore water distributions of dissolved copper and copper-complexing ligands in estuarine and coastal marine sediments. Geochimica et Cosmochimica Acta 64, 1843–1857.

sediment incubations (often using ¹⁵N-tracers), and benthic lander determinations of sediment nitrate uptake or N₂ gas efflux. However, recent studies have shown that nitrate loss or N₂ production in sediments does not exclusively occur by organotrophic denitrification (as illustrated in **Table 1**), but can also occur by processes such as anammox (ammonium oxidation by nitrite to produce N₂ gas), dissimilatory nitrate reduction to ammonium, or denitrification coupled to sulfide, Mn^{2+} and Fe²⁺ oxidation (see **Table 2**, Section 5.09.2.2.2, and the discussion in the next section). More work is therefore needed to elucidate carefully the different processes that impact nitrogen transformations in sediments, and to verify the role of organotrophic denitrification in sediment organic carbon remineralization (also *see* Chapter **5.08**).

Recent studies in permeable sediments have also provided evidence for the occurrence of aerobic denitrification, that is, N_2 production in the presence of up to ~100 μ M O₂ (Rao et al., 2007; Gao et al., 2009). While this type of denitrification has been observed in pure culture laboratory studies with a broad range of bacteria (also see Zehr and Ward, 2002; Rao et al., 2008), studies of denitrification in other more muddy marine sediments and oxygen-deficient water-column environments are consistent with the conventional view of denitrification as an anaerobic processes that does not occur until dissolved oxygen is essentially completely zero (less than $\sim 2-4 \,\mu$ M; Brandes et al., 2007). Although the exact mechanism(s) by which aerobic denitrification in permeable sediments occurs is not well understood, one possible explanation is that organisms capable of carrying out this process have adapted to the varying redox conditions associated with the advective flow and exchange of anoxic pore water and oxic bottom water that occurs in permeable sediments (see Gao et al. (2009), and references cited therein). An alternate explanation is that the process actually occurs under anoxic conditions through closely coupled nitrification/denitrification in microenvironments isolated from the bulk pore waters (Jenkins and Kemp, 1984; Brandes and Devol, 1995; Rao et al., 2008). In this sense, this latter explanation shows some similarities to a process referred to oxygen-limited autotrophic nitrification-denitrification, or OLAND (Brandes et al., 2007). At the present time, the

evidence favoring any of these mechanisms over the others is equivocal.

Interest in denitrification in sediments stems, in part, from the fact that it represents the major removal process for fixed nitrogen in the oceans (Devol, 2008). Global-scale model calculations (Middelburg et al., 1996b) indicated that much of this removal (~60%) occurs in continental slope sediments (defined here as sediments in water depths >150 m), with the remaining removal occurring primarily in shallower continental shelf sediments (also see Burdige, 2006). However until recently, the overwhelming majority of studies of sedimentary denitrification has been carried out in relatively fine-grained, muddy sediments, and these results largely formed the database used in the Middelburg et al. (1996b) calculations.

Estimates of denitrification rates in permeable sands are shown in Table 3, along with the model estimates from Middelburg et al. (1996b). At first glance, it would appear that estimates of denitrification in permeable sediments are slightly lower than the Middelburg et al. (1996b) model estimates. However, this conclusion must be tempered by the observation that when the occurrence of advective processes in permeable sediments is not incorporated into the experimental design, estimates of rates of biogeochemical processes in permeable sediments may be inaccurate and not representative of the true in situ rate (Huettel and Gust, 1992; Jahnke et al., 2000; Reimers et al., 2001; Berg and Huettel, 2009). Similarly, given the complexity of advective flow in permeable sediments, rate measurements (or model estimates) that do incorporate the occurrence of this flow may be difficult to interpret, or scale up, to obtain a larger-scale understanding of the rates of biogeochemical processes in permeable sediments (e.g., Cardenas et al., 2008).

However since sandy continental shelf sediments comprise ~70% of all continental shelves globally, if denitrification rates in high-permeability, sandy sediments are less than those in muddy sediments at similar water depths, then previous global estimates of marine sediment denitrification (Middelburg et al., 1996b) may have overestimated the magnitude of this loss term. Such considerations have implications regarding questions of whether or not the oceanic nitrogen budget is in

Table 3 Denitrification rates in continental shelf sediments

Site/methodology	Advection ^a	Denitrification rate (mmol m ^{−2} day ^{−1})
Permeable sediments		
Static core incubation, South Atlantic Bight (Vance-Harris and Ingall, 2005)	No	0.02
Benthic flux chamber incubation, mid-Atlantic Bight (Laursen and Seitzinger, 2002)	No	1.7
Static core incubation (stirred overlying water), North Sea (Lohse et al., 1996)	No	0.28
Benthic flux chamber incubations, Washington (USA) shelf (Devol, 1991; Devol and Christensen, 1993)	No	~2.3–3.2
Flow-through sand columns, South Atlantic Bight (Rao et al., 2007)	Yes	0.25-1.46
Core incubations using continuous advection chambers ^b , Gulf of Mexico (Gihring et al., 2010)	No	0.13
	Yes	0.26-0.87
Numerical simulations with continental shelf parameters using coupled hydrodynamic and sediment biogeochemical models (Cardenas et al., 2008)	Yes	<0.24 to ~0.72
Data-driven, model estimate for North Atlantic continental shelf sediments (Seitzinger and Giblin, 1996)	N/A	0.69
Middelburg et al. (1996b) model estimate for continental margin sediments	No	~1.0–1.7

^a This column indicates whether pore-water advective processes were incorporated into the experimental (or modeling) techniques used here to determine denitrification rates. ^b By varying the stirring rate in these chambers, one can go from diffusion-controlled chambers with no pore-water advection across the sediment–water interface, to chambers with varying amounts of pore-water advection (also see Huettel and Rusch, 2000, for details). steady state with regard to input and removal terms (see Cardenas et al., 2008 for details), and will require further study.

Denitrification (in general) accounts for the removal of up to ~50% of the reactive nitrogen input to estuarine sediments (see compilation in Burdige, 2006). In whole estuary N budgets, denitrification, which largely occurs in the sediments (Boynton and Kemp, 2008), represents 2–3% to up to ~75% of the total nitrogen input to the estuary, and is a strong function of the water residence time in the estuary (Nixon et al., 1996). In shallow estuarine systems (i.e., those with significant benthic primary production), factors associated with this primary production may also control the rates of sedimentary denitrification (Joye and Anderson, 2008). This topic is discussed in greater detail in Chapters **5.07** and **5.08**.

5.09.6.1 Anammox

In recent years, it has been shown that denitrification can be combined with the oxidation of substrates other than organic matter (i.e., see reactions 1–4 in Table 2), and one such reaction that has been of particular interest in recent years is the anammox reaction (reaction 1 in Table 2). Strictly speaking, this reaction involves ammonium oxidation at the expense of nitrite (rather than nitrate) reduction. However, in nature, nitrite is generally observed at much lower concentrations than nitrate, and is often a transient intermediate in oxidative and reductive processes in the nitrogen cycle (e.g., nitrification and denitrification; Hulth et al., 2005). Furthermore, because nitrate is readily reduced to nitrite in environments where anammox occurs (Dalsgaard et al., 2005), the biogeochemical significance of anammox is often considered from the perspective of nitrate reduction coupled to ammonium oxidation, or

$$5NH_4^+ + 3NO_3^- \rightarrow 4N_2 + 9H_2O + 2H^+$$
 [3]

In fact, early evidence for the occurrence of anammox was based on thermodynamic calculations and field observations that suggested the occurrence of this specific reaction (for this historical perspective, see Dalsgaard et al., 2005; Hulth et al., 2005; and Devol, 2008).

Anammox is expected to occur in O₂-deficient sedimentary environments similar to those in which organotrophic denitrification occurs. However, anammox is presumably carried out by autotrophic organisms (who fix carbon using the energy gained from reaction 3), whereas organotrophic denitrifiers are heterotophs. This difference is not well discussed in literature in terms of the relative importance of the two processes (although do see discussions in Ward et al., 2009), and it may be important in terms of the overall controls on the occurrence of these two N removal processes. Anammox also requires a source of both nitrite and ammonium, and this too may play a role in regulating the occurrence of this process in sediments (see below for details).

Rates of anammox have been determined in a range of estuarine, coastal, and continental margin sediments, from the intertidal down to water depths of ~700 m (see Dalsgaard et al., 2005 for a review). As a percentage of overall N₂ production in sediments, the importance of anammox increases with increasing water depth (Dalsgaard et al., 2005), from less than ~5% in intertidal sediments to ~60–70% in Danish continental

margin sediments of the Skagerrak (700 m water depth). In this data set, this trend is largely controlled by a strong decrease in the rate of organotrophic denitrification (roughly 2 orders of magnitude) with water depth. Absolute anammox rates are perhaps higher at shallower depths, although the overall magnitude of change in the observed rates is one order of magnitude or less.

More specifically though, rates of organotrophic denitrification in sediments are positively correlated with somewhat interconnected parameters such as sediment organic matter reactivity or reductant availability (Thamdrup and Dalsgaard, 2002; Trimmer et al., 2003; Dalsgaard et al., 2005; Engström et al., 2005). In contrast, the anammox rate shows no such behavior (Dalsgaard et al., 2005; Engström et al., 2005). This could then explain the observed trend of an increase in the relative importance of anammox with increasing water depth since, for example, sediment organic matter remineralization rates show a significant inverse correlation with water-column depth (Dalsgaard et al., 2005; Burdige, 2006). Ultimately, however, nitrite availability may play a key role in regulating anammox activity in sediments (e.g., Hulth et al., 2005), although how this occurs is not well understood. Finally, the issue of sediment stability (as discussed, for example, in the next section regarding anoxic nitrification) may be important in regulating anammox activity in sediments (Dalsgaard et al., 2005; Hulth et al., 2005), although again this suggestion is not well tested.

In estuarine settings, the relative importance of anammox to total N₂ production generally decreases with increasing salinity (Trimmer et al., 2003; Rich et al., 2008; Koop-Jakobsen and Giblin, 2009), although, consistent with the discussion above, in all of these studies anammox was ~10% or less of total N₂ production in estuarine sediments, except in tidal freshwater sediments in the upper Chesapeake Bay where this value was ~20% (Rich et al., 2008). It is unknown at the present time whether the observed trend in relative anammox production rates with salinity is the result of a direct salt effect on microbial processes affecting nitrogen cycling, or is the result of an indirect effect that simply covaries with salinity.

5.09.6.2 Anoxic Nitrification

The anammox reaction represents one of a number of alternate pathways for anoxic ammonium oxidation in nature. Other such reactions that have generated interest in recent years include anoxic nitrification (reaction 5 in Table 2) as well as Mn reduction coupled to oxidative N2 production (reaction 6 in Table 2). A wide range of seemingly contradictory observations are reported in the literature for and against the occurrence of these reactions and/or biological involvement in these reactions (see recent discussions in Hulth et al., 2005 and Bartlett et al., 2008). For anoxic nitrification, Bartlett et al. (2008) have attempted to reconcile these observations by suggesting that the reaction is strongly dependent on sediment stability (defined in a very broad sense to include physical processes such as flood events or sediment slumping, sediment mixing by bioturbation, as well as perturbations to the sediments that occur during experimental setup in manipulative studies examining anoxic nitrification). Future work will be needed to test this suggestion and thus better understand the

role (and significance) of these processes in Mn and N dynamics in marine sediments.

5.09.6.3 Nitrogen Loss from Sediments: Closing Thoughts

The identification of a number of new nitrogen reactions has indicated that oceanic and sedimentary nitrogen cycling is more complex than previously thought (e.g., Hulth et al., 2005). Interest in nitrogen cycling in estuarine and coastal sediments stems, in part, from this fact, and the fact that these sediments represent the major sites for the removal of fixed nitrogen in the oceans. Historically, this removal has been thought to occur primarily by N₂ production associated with anaerobic organotrophic denitrification (Devol, 2008). However in this regard, we know very little about the qualitative and quantitative significance of aerobic denitrification, anammox, and other pathways for nitrogen removal with more direct linkages to the Mn and Fe redox cycling (e.g., see Table 2), and more work here is clearly needed (also see similar discussions in Hulth et al., 2005 and Chapter 5.08). These new linkages of N cycling to Fe and Mn redox cycling further suggest that suboxic sediments such as physically reworked deltaic shelf sediments (e.g., the Amazon shelf; see Section 5.09.4.2) may be more important sites for nitrogen cycling (and perhaps fixed nitrogen loss) than previously thought (e.g., Hulth et al., 2005).

5.09.7 Iron and Manganese Reduction

Unlike the other respiratory processes discussed here, organotrophic manganese and iron reduction use solid, rather than dissolved, electron acceptors. (Note that here metal oxide reduction coupled to organic carbon remineralization will be referred to as organotrophic metal oxide reduction, to differentiate it clearly from abiotic (or microbial) reduction coupled to the oxidation of other chemical species (e.g., sulfide).) The most reactive forms (with respect to abiotic or microbial reductive dissolution) are generally poorly crystalline, amorphous oxides, or oxyhydroxides (Burdige et al., 1992; Thamdrup, 2000), referred to here, in a general sense, as metal (or Fe or Mn) oxides. In sediments these oxides are often found as coatings on inorganic or biogenic particles, such as clays or siliceous tests (for a more detailed discussion of the mineralogy of these oxide phases, see Burns and Burns, 1981; Cornell and Schwertmann, 1996; and Post, 1999). Microbes can also reduce other forms of oxidized manganese and iron, including more crystalline oxide phases, as well as structural Fe(III) in clay minerals such as smectite (e.g., Burdige et al., 1992; Roden and Zachara, 1996; Kostka et al., 2002a; Dong et al., 2003). More detailed discussions of the mechanisms of microbial Fe and Mn reduction and the organisms involved can be found elsewhere (Nealson and Saffarini, 1994; Lovley, 2000).

An important aspect of Fe and Mn biogeochemistry is that these elements undergo redox transformations at Earth surface conditions, which change their solubility. Reduced forms can, under some conditions, be soluble, while oxidized forms (solid oxides or oxyhydroxides) are highly insoluble. The occurrence of iron and manganese reduction in sediments is often consistent with the concept of biogeochemical zonation discussed earlier. However Mn and Fe oxides can also be reduced by inorganic compounds such as sulfide, and a number of organic acids and ligands; furthermore, Fe²⁺ itself can reduce manganese oxides (Burdige and Nealson, 1986; Myers and Nealson, 1988b: Yao and Millero, 1993: Stone et al., 1994: Sunda and Kieber, 1994; Yao and Millero, 1996). In particular, since sulfate reduction is an important process in estuarine and coastal sediments, upward fluxes of sulfide from deeper sediments can lead to Fe²⁺ and Mn²⁺ production in more shallow regions of the sediments through sulfide mediated metal oxide reduction (i.e., not directly coupled to bacterial respiration), despite the fact that organotrophic metal oxide reduction might be predicted to occur there. The same is also likely true in the upper portion of the sulfate reducing zone of such sediments, particularly when sulfate reduction begins essentially at the sediment-water interface (e.g., see Figure 6 and relevant discussions in Thamdrup, 2000). Not only does this confound, at times, the ability to estimate rates of organotrophic Fe and Mn reduction in sediments (Kostka et al., 2002c), but these interactions between the Mn and Fe redox cycles and the sulfur cycle likely play a role in regulating the relative importance of organotrophic Fe and Mn reduction in overall sediment organic carbon remineralization.

Compared to other organic matter remineralization processes, far fewer studies have directly examined rates of organotrophic metal oxide reduction in sediments. There are a number of reasons for this (Thamdrup, 2000; Thamdrup and Canfield, 2000), including the fact that during sediment incubations the production of chemical reductants of these oxides (e.g., sulfide production by bacterial sulfate reduction) can make it difficult to distinguish between microbial (organotrophic) Mn and Fe reduction and abiotic, chemical reduction. Rates of organotrophic Mn and Fe reduction can also be obtained by modeling pore-water profiles (Berg et al., 2003), and, somewhat more indirectly by subtracting the rates of other remineralization processes (e.g., sulfate reduction or aerobic respiration) from estimates of the rate of total sediment carbon oxidation (Canfield et al., 1993a, 1993b; Kristensen et al., 2000; Thamdrup, 2000; Jensen et al., 2003; Hyun et al., 2009).

Where rates of metal oxide reduction have been examined in the context of total organic carbon remineralization in estuarine and coastal sediments (Figure 7), we see that the relative rates range from 0% to almost 100%, with a large number of sites having relative rates of ~20-50%. Although not explicitly shown here, under almost all circumstances Fe reduction, rather than Mn reduction, is the predominant organotrophic metal oxide reduction process (see below for details). Canfield et al. (2005) similarly suggested (using much of the same data shown here) that where rates of iron reduction have been directly determined, iron reduction is, on average, 17% of total sediment organic carbon remineralization and 26% of anaerobic organic carbon remineralization. In past studies, the failure to explicitly account for organotrophic metal oxide reduction potentially leads to an overestimate of the importance of aerobic respiration relative to sulfate reduction (Jensen et al., 2003).

The results in Figure 7 also show that there does not appear to be any consistent trend in the relative rates of organotrophic metal reduction with water-column depth, beyond the fact that the process becomes insignificant in sediments in water depths greater than ~1000–2000 m. In these deep-water sediments,



Figure 7 The relative rate of organotropic Fe and Mn reduction (as compared to total sediment organic carbon oxidation or R_{cox}) vs. water depth. At all sites except those where the data points are annotated by '* (and where Mn reduction dominates) Fe reduction is the predominant (and almost the exclusive) organotrophic metal oxide reduction process. Low bottom-water oxygen sites are indicated by red triangles. Modified from Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton where references to most of the original data can be found. New data shown here are from Rysgaard, S., Thamdrup, B., Risgaard-Petersen, N., Fossing, H., Berg, P., Christensen, P.B., Dalsgaard, T., 1998. Seasonal carbon and nutrient mineralization in a high-Arctic coastal marine sediment, Young Sound, Northeast Greenland. Marine Ecology Progress Series 175, 261-276; Kristensen, E., Andersen, F.Ø., Holmboe, N., Holmer, M., Thongtham, N., 2000, Carbon and nitrogen mineralization in sediments of the Bangrong mangrove area. Phuket, Thailand. Aquatic Microbial Ecology 22, 199-213; Thamdrup, B., 2000. Bacterial manganese and iron reduction in aquatic sediments. Advances in Microbial Ecology 16, 41-83; Thamdrup, B., Canfield, D.E., 2000. Benthic respiration in aquatic sediments. In: Sala. O.E., Jackson, R.B., Mooney, H.A., Howarth, R.A. (Eds.), Methods in Ecosystem Science. Springer, New York, NY, pp. 86-102; Kostka, J.E., Gribsholt, B., Petrie, E., Dalton, D., Skelton, H., Kristensen, E., 2002b. The rates and pathways of carbon oxidation in bioturbated saltmarsh sediments. Limnology and Oceanography 47, 230-240; Kostka, J.E., Roychoudhury, A., Van Cappellen, P., 2002c. Rates and controls of anaerobic microbial respiration across spatial and temporal gradients in saltmarsh sediments. Biogeochemistry 60, 49-76; Berg, P., Rysgaard, S., Thamdrup, B., 2003. Dynamic modeling of early diagenesis and nutrient cycling. A case study in an Arctic marine sediment. American Journal of Science 303, 905-955; Glud, R.N., Gundersen, J.K., Røy, H., Jørgensen, B.B., 2003. Seasonal dynamics of benthic O₂ uptake in a semienclosed bay: importance of diffusion and faunal activity. Limnology and Oceanography 48, 1265-1276; Jensen, M.M., Thamdrup, B., Rysgaard, S., Holmer, M., Fossing, H., 2003. Rates and regulation of microbial iron reduction in sediments of the Baltic-North Sea transition. Biogeochemistry 65, 295-317.

sediment O_2 uptake keeps up with the low organic carbon flux to the sediments and aerobic respiration dominates sediment organic matter remineralization (Emerson et al., 1985; Burdige, 2006).

Despite the fact that organotrophic Mn reduction is generally not a significant organic carbon remineralization process in most marine sediments, Mn redox cycling is often significant in estuarine and coastal (and other continental margin) sediments, and can play an important role as an electron shuttle in the sediments (see Section 5.09.2.3, and Burdige and Nealson, 1986; Canfield et al., 1993b; and Aller, 1994b). For example, in Long Island Sound sediments, re-oxidation of Mn^{2+} annually accounts for ~40% of the sediment oxygen uptake, while sulfate reduction is the dominant organic matter remineralization process (Aller, 1994b). The proposed coupling of such redox reactions in these sediments is illustrated in Figure 1(b).

In examining metal oxide reduction rates in sediments (regardless of whether or not the process is microbially catalyzed), it has been noted that high rates of metal oxide reduction require internal processes that continually replenish the reactive metal oxides, since the needed supply of oxides greatly exceeds their input by sedimentation (Burdige, 1993; Canfield et al., 1993b; Thamdrup, 2000). This implies that there must be significant internal redox cycling between the reduced and oxidized forms of these metals, through processes described in Sections 5.09.2.3 and 5.09.3, to sustain high rates of metal oxide reduction. The initial oxides that form in sediments from Fe²⁺ and Mn²⁺ oxidation are amorphous, poorly crystalline phases (Burdige, 1993), which given time will slowly transform into more stable crystalline forms. This rapid redox cycling therefore maintains sediment metal oxides in their most reactive form, which may act as a positive feedback on organotrophic metal oxide reduction (Thamdrup, 2000).

The specific transport processes associated with this redox cycling vary among different sediments. Internal Mn and Fe redox cycling can be driven by relatively slow processes such as molecular diffusion and sediment burial (e.g., Burdige and Gieskes, 1983), although an examination of the sites summarized in Figure 7 suggests that these processes do not significantly contribute to the high relative rates of metal reduction shown in this figure. In many sediments (particularly finegrained subtidal sediments in estuaries and on the continental shelf), bioturbation or bioirrigation play an important role in this redox cycling. They may provide the needed oxidant (e.g., O₂) input for the re-oxidation of reduced metals, and bioturbation also transports particulate oxides that are diagenetically produced in surface sediments to deeper depths where they are then consumed by microbial or chemical metal oxide reduction (Sundby and Silverberg, 1985; Aller, 1990; Canfield et al., 1993b; Aller, 2001; Berg et al., 2003). In intertidal (e.g., salt marsh) sediments, the activities of fiddler crabs and similar benthic organisms can play an analogous role in this redox cycling (e.g., Kostka et al., 2002b), while in vegetated subtidal and intertidal sediments (e.g., salt marshes, mangrove forests, seagrass beds) the input of photosynthetically produced O₂ also is important (Holmer et al., 1999; Kristensen et al., 2000; Hyun et al., 2009).

Oxygen input by pore-water advective exchange in permeable sediments (Section 5.1.2.1) can also promote the redox cycling of Mn and Fe, as well as other redox-sensitive trace metals. Such Mn and Fe redox cycling has been demonstrated in laboratory flume studies with natural permeable sands (Huettel et al., 1998), although the mechanism(s) and biogeochemical significance of this metal redox cycling have to date not been examined *in situ*.

Physical reworking of mobile tropical muds and deltaic sediments (Section 5.09.4.2) has similarly been shown to mediate Fe (and to a lesser extent Mn) redox cycling. The frequency and intensity of physical reworking lead to non-steady-state conditions in the sediments (Mackin et al., 1988) and the continual re-oxidation of reduced Fe or Mn (or S) that is produced in the sediments. Thus, in comparison to sediments in low-energy environments which have comparable rates of net sediment accumulation and organic carbon remineralization (see Section 5.09.4.1), these reworked sediments are, roughly speaking, poised at a redox state such that in a net sense iron reduction, rather than sulfate reduction, dominates organic matter remineralization (e.g., Aller et al., 1986).

The term net is used here, since it has not been definitively demonstrated that the observed iron reduction in these sediments is indeed directly coupled to organic carbon remineralization. Rather it may involve a coupling with sulfur redox cycling in which bacterial sulfate reduction is actually the predominant respiratory process in the sediments, which is coupled to iron reduction through sulfide oxidation (Aller et al., 1986; Madrid et al., 2001). However, calculations presented by Aller (2004) suggest that if the biogeochemical zonation inferred by Table 1 occurs in physically reworked sediments not in vertical sense (as it is normally considered), but in a temporal sense (i.e., during the non-steady-state diagenesis of the sediments as a whole after a resuspension/ reoxidation event), then the availability of oxygen, nitrate, and reactive Fe and Mn oxides is such that sulfate reduction should not occur in these sediments in less than ~1 year. Therefore, because the occurrence of physical reworking has a frequency of less than $\sim 1 \text{ yr}^{-1}$ (e.g., McKee et al., 2004), the sediments should be poised at suboxic conditions with no net (and minimal gross) sulfate reduction.

As noted earlier (Section 5.09.4.2), this type of physical reworking of sediments and its associated metal redox cycling are not restricted to continental shelf mobile muds, and can also occur in certain high-energy regions of estuaries, such as the estuarine turbidity maximum (Dellapenna et al., 1998; Abril et al., 1999; Schaffner et al., 2001; Galler and Allison, 2008). Studies of the estuarine turbidity maximum in the Gironde estuary (between salinity 0.15 and 6) clearly demonstrate that fluid muds in this region are sites for intense metal cycling (Robert et al., 2004). However, studies have been unable to differentiate whether the observed Mn and Fe reduction occurs through organotrophic metal oxide reduction, or, as discussed above, through a coupling with sulfate reduction and sulfide oxidation (Audry et al., 2006; Audry et al., 2007a, 2007b; also see Section 5.09.10).

Organotrophic iron reduction shows some dependence on the concentration of reactive Fe(III) oxides (Thamdrup, 2000; Gribsholt et al., 2003; Jensen et al., 2003), and the importance of organotrophic iron reduction (relative to total anaerobic remineralization) generally begins to significantly decline when the concentration of highly reactive Fe(III) drops below \sim 30–40 µmol \cdot cm⁻³ (Figure 8). Since anaerobic remineralization generally dominates organic carbon remineralization in the largely estuarine and coastal sediments shown in this figure (Section 5.09.5), these observations further point to the importance of organotrophic iron reduction in shallow water sediments. It can also be seen in Figure 8 that there is a fair bit of scatter around the curve here. In part, this may be the results of some of the inherent difficulties (and large uncertainties) in determining rates of organotrophic metal oxide reduction (Thamdrup, 2000; Thamdrup and Canfield, 2000). However, these results may also point to the possibility that high rates of internal Fe redox cycling can stimulate high rates of organotrophic iron reduction in a manner that is not simply



Figure 8 The relative rate of organotrophic Fe reduction (as compared to anaerobic sediment organic carbon oxidation) in marine and estuarine sediments vs. the concentration of reactive Fe(III) oxides in the sediments. The empirical best fit curve through the data is %FeR = 90% (1 $-e^{-0.069\mbox{ Fe(III)}}$: see Jensen et al., 2003) for a very similar best-fit curve obtained with a subset of these results). Reactive Fe is defined here as the amount of Fe in oxides that is available for reductive dissolution by sulfide or organotrophic metal oxide reduction. Many of the analytical schemes used to examine iron speciation in sediments refer to this as 'poorly crystalline' or 'highly reactive' Fe(III). From Kristensen, E., Andersen, F.Ø., Holmboe, N., Holmer, M., Thongtham, N., 2000. Carbon and nitrogen mineralization in sediments of the Bangrong mangrove area, Phuket, Thailand. Aquatic Microbial Ecology 22, 199-213; Thamdrup, B., 2000. Bacterial manganese and iron reduction in aquatic sediments. Advances in Microbial Ecology 16, 41-83; Gribsholt, B., Kostka, J.E., Kristensen, E., 2003. Impact of fiddler crabs and plant roots on sediment biogeochemistry in a Georgia saltmarsh. Marine Ecology Progress Series 259, 237-251; Jensen, M.M., Thamdrup, B., Rysgaard, S., Holmer, M., Fossing, H., 2003. Rates and regulation of microbial iron reduction in sediments of the Baltic-North Sea transition. Biogeochemistry 65, 295-317; and Hyun, J.-H., Mok, J.-S., Cho, H.-Y., Kim, S.-H., Lee, K., Kostka, J., 2009. Rapid organic matter mineralization coupled to iron cycling in intertidal mud flats of the Han River estuary, Yellow Sea. Biogeochemistry 92, 231-245.

dependent on the sediment concentration (pool size) of reactive Fe(III).

Assuming that a threshold Mn concentration comparable to that seen in Figure 8 also holds for organotrophic manganese reduction (Thamdrup et al., 2000), it suggests that the low concentration of manganese in most estuarine and continental margin sediments (Table 4) may limit organotrophic manganese reduction, at the expense of either sulfate reduction or organotrophic iron reduction. Such a situation is also expected to have a negative feedback on organotrophic manganese reduction, since the end products of these other metabolic processes (sulfide and Fe²⁺) rapidly reduce manganese oxides (Figure 6). Differences in the rates of reduced metal re-oxidation $(Fe^{2+} > Mn^{2+} at natural pH values)$ and in the solubility of reduced metal phases such as sulfides or carbonates (Fe(II) < Mn(II)) may play some role here by allowing many sediments to better retain reduced Fe versus Mn during metal redox cycling (e.g., see benthic flux chamber results in Balzer, 1982 and Sundby et al., 1986). Therefore, factors that enhance the retention of Mn in the sediments (a relatively deep oxygen penetration depth as in the Skaggerak sediments) or lead to

Table 4 Representative values of Fe and Mn in differing marine sediments

	Concentration ^a		
Location	(μmol gdw ⁻¹)	(μmol cm ⁻³)	
Reactive Fe ^b			
Mesohaline Chesapeake Bay ^c	150-200	~20–50	
Continental margin sediments ^d	~50–200	~25–100	
N. American temperate nearshore and continental margin sediments ^e	~100–200	~25–100	
Selected sites where Fe reduction appears to be a major component of sediment organ	ic carbon remineralization		
Gulf of Papua ^f	390–450	~240–270	
Amazon continental shelf ^g	~280	~80–200	
Georgia (US) salt marsh sediments ^h		100-125	
Baltic-North Sea transition sediments ⁱ		25-40	
Intertidal mangrove sediments, Thailand [/]	~15–30	~9–16	
Reactive Mn ^b			
Mesohaline Chesapeake Bay ^k	~5–30	~1–16	
Nearshore muddy sediments'	<15	<8	
Sites where Mn reduction appears to be a major component of sediment organic carbo	n remineralization		
Panama Basin (3890 m water depth) ^m	180–730	70-290	
Skaggerak (Danish continental margin sediments; 695 m water depth) ⁿ		~100–150	
Oxic Black Sea sediments (surface sediments; 62–100 m water depth) ^o		~20–120	

^{*a*} Concentrations reported in the literature as μ mol gdw⁻¹ (or wt.%) were converted to μ mol cm⁻³ (total volume of wet sediment) for ease of comparison among the entire compilation of data presented here. This conversion is very sensitive to the assumed porosity of the sediment (ϕ) since the calculation depends on the term (1- ϕ). Thus, for example, a range in assumed porosities of 0.8–0.9 (–11% difference) results in a factor of two difference in concentrations expressed per cm³ of total sediment (for further details, see Burdige, 2006). ^{*b*} Reactive Fe or Mn is defined here as the amount of Fe and Mn in oxides that is available for reductive dissolution by sulfide or organotrophic metal oxide reduction. For iron, many of the analytical schemes used to examine iron speciation in sediments refer to this as poorly crystalline or highly reactive Fe(III) oxides.

^c From Cornwell and Sampou (1995). Porosity was assumed to range from 0.9 to 0.95.

^d From results in Canfield et al. (1992) and Raiswell and Canfield (1998). These workers report a range of 1–4 wt.% total Fe in these sediments, and based on their extraction techniques they estimate that 25–28% of the total iron is highly reactive iron. Porosity was assumed to be 0.8.

^e From a summary presented in (Aller et al., 2004a). Porosity was assumed to range from 0.8 to 0.9.

⁷ From Aller et al. (2004a). Porosity was assumed to be 0.75.

⁹ From Aller and Blair (1996). Based on a porosity range of 0.72–0.9, using % water data in Aller and Aller (1986).

^h From Kostka et al. (2002b).

ⁱ From Jensen et al. (2003).

^{*j*} From Kristensen et al. (2000).

^k From Holdren et al. (1975). Porosity was assumed to range from 0.9 to 0.95.

¹ From Chester (2000). Since this is the total Mn content of the sediment, this is an upper limit of the Mn that is found in reactive oxides. Porosity was assumed to be 0.8.

^m From Aller (1990). Porosity was assumed to be 0.85. For comparison, these concentrations are 1–4 wt% Mn.

ⁿ From Canfield et al. (1993b) and Canfield (1993).

^o From Thamdrup et al. (2000).

elevated Mn input to sediments (hydrothermal inputs to Panama Basin sediments) appear to be important for organotrophic manganese reduction to be a significant remineralization process. However, why extraordinarily high concentrations of sedimentary manganese oxides are required for organotrophic Mn reduction to be quantitatively significant is not well understood (see Thamdrup, 2000 for further discussions).

To the best of my knowledge, there have been no studies that have specifically addressed the importance of organotrophic metal oxide reduction as a function of salinity in an estuary. Organotrophic iron reduction is important in freshwater wetlands (Roden and Wetzel, 1996), and inhibits the occurrence of methanogenesis (also see Lovley and Phillips, 1987), consistent with the biogeochemical zonation and anaerobic food chain models discussed earlier. Historically, the distribution of remineralization processes in estuarine sediments has been examined in terms of the relative roles of sulfate reduction versus methanogenesis (e.g., Martens and Goldhaber, 1978), based on sulfate availability in an estuary as a function of salinity, and the competition between sulfate reducers and methanogens for key carbon substrates (also see Capone and Kiene, 1988). However, organotrophic iron reduction is both thermodynamically (Table 1) and kinetically (Section 5.09.2.1.1) favored over sulfate reduction and methanogenesis. Thus, the relative importance of organotrophic iron reduction should not be directly affected by the increasing importance of sulfate reduction and the decreasing importance of methanogenesis as one moves down an estuary. The relative importance of organotropic iron reduction may be indirectly affected by other factors that vary through an estuary, such as the intensity of iron redox cycling in the sediments (to supply the needed reactive oxides), as well as competition between iron reduction mediated by sulfide oxidation versus organotrophic iron reduction. Based on the discussions above, these factors themselves may be related to, for example, changes in the quality and quantity of estuarine organic matter, changes in the intensity and types (e.g., bioturbation versus bioirriagation) of benthic macrofaunal processes, as well as changes in the importance of sulfate reduction in estuarine sediments. More work is needed to examine critically these suggestions.

Finally, Mn and Fe redox cycling in general, regardless of the exact mechanisms by which it occurs, plays an important role in the biogeochemical cycles of other elements (e.g., phosphorus and other trace metals) in estuarine and coastal sediments and water columns. This will be discussed in Section 5.09.11 (also see Ruttenberg, 2003 and Chapter 5.06).

5.09.8 Sulfate Reduction and Sulfide Mineral Formation

Sulfate reduction plays an important role in organic matter remineralization in estuarine and coastal sediments, generally accounting for ~50% (to up to ~100%) of R_{cox} (Canfield et al., 2005; Burdige, 2006). When compared to other anaerobic remineralization processes, the large concentration of sulfate in seawater (~28 mM) versus that of other oxidants such as nitrate or oxygen plays a role. However, as one moves into the low-salinity reaches of an estuary, this general observation can break down (see the end of this section for details).

Low relative rates of sulfate reduction can also be observed in estuarine and coastal sediments, suggesting that a number of factors regulate its occurrence in these environments. For example, discussions in Section 5.09.7 have shown that organotropic iron reduction can be a major component of anaerobic remineralization in many coastal sediments. Similarly, the occurrence of sulfate reduction requires a relatively large flux of reactive organic matter to the sediments, as compared to the input of oxidants such as O_2 or nitrate. Thus nearshore sediments with low carbon inputs are likely to be sites where sulfate reduction is of minor importance.

The sulfur cycle in estuarine and coastal sediments (and continental margin sediments in general) is also of interest and importance for other reasons. These regions represent the dominant sites of sulfur removal from the modern oceans due to burial here of pyrite (FeS₂) or diagentically produced organic sulfur (Vairavamurthy et al., 1995; Berner and Berner, 1996; Goldhaber, 2003). Both of these are products of sulfate reduction (Figure 9). Long-term changes (>100 million years) in this burial rate are linked to changes in the global redox state of the Earth's surface, that is, this plays a role in controlling atmospheric oxygen levels on these long timescales (Berner, 1999; Hedges, 2002; Berner, 2004). This topic is beyond the scope of this article, although the interested reader is urged to look at the references cited here for further details. Finally, in estuarine and coastal sediments, the cycling of trace metals (many of which may be found at relatively high levels as a result of anthropogenic inputs) is strongly linked to sulfur redox cycling (uptake in sulfide phases such as pyrite, release during pyrite oxidation). The topic is discussed in Section 5.09.10.

The conceptual model in Figure 9 that is used to describe sulfur cycling in recent marine sediments includes many processes that have been discussed previously, including bacterial sulfate reduction and sulfide oxidation. One set of processes that have not been discussed are sulfur disproportionation



Figure 9 A schematic representation of the sediment sulfur cycle. Note that red arrows indicate reductive processes, blue arrows indicate oxidative process, and black arrows indicate no redox transformation. Modified from Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton.

reactions involving elemental sulfur, thiosulfate, and sulfite. These reactions are the inorganic equivalent of organic fermentation reactions (see Section 5.09.2.1.1) and result in the production of reduced sulfide and oxidized sulfate from these intermediate oxidation state sulfur species (Jørgensen and Bak, 1991). For example, thiosulfate disproportionation can be written as

$$S_2O_3^{2-} + H_2O \rightarrow H_2S + SO_4^{2-}$$
 [4]

In addition, analogous reactions can be written for S° and sulfite disproportionation. An important aspect of sulfur disproportionation is that it results in significant sulfur isotope fractionation. In particular, the sulfide that is produced by this disproportionation can be depleted in ³⁴S by ~3–28‰ relative to the starting intermediate oxidation state sulfur compound (Canfield and Thamdrup, 1994; Habicht et al., 1998; Habicht and Canfield, 2001). The significance of this is discussed below.

In examining Figure 9 we start with bacterial sulfate reduction, which produces dissolved sulfide as one of its end products. In nearshore sediments with TOC contents of ~1-3%, sulfate reduction occurs over the upper tens of centimeters to several meters of sediment. Depending on whether sulfate or reactive organic carbon is limiting, sulfate concentrations either decrease to zero, oftentimes in an exponential-like fashion (see Figures 6 and 10), implying sulfate limitations, or reach some asymptotic nonzero concentration at depth, implying limitations by reactive organic carbon. The dissolved sulfide produced by bacterial sulfate reduction in sediments has several possible fates: reaction with iron oxides or Fe²⁺ to form FeS (see below), incorporation into sediment organic matter (formation of organo-sulfur compounds), reaction with FeS to form pyrite, and oxidation to form S° or other more oxidized sulfur species, up to, and including, sulfate.

The reaction of sulfide with iron oxides is an important process in sediments and plays a key role in explaining why sulfide does not accumulate in pore waters until some depth below that where sulfate reduction (sulfide production) begins to occur (**Figure 6**). In such regions of the sediment, reactive iron oxides (Section 5.09.7), if present, are rapidly reduced by dissolved sulfide, effectively removing sulfide from the pore waters. Once these reactive oxides are exhausted, and Fe^{2+} in the pore waters precipitates out as FeS, more refractory forms of oxidized iron (e.g., more crystalline oxides, magnetite, and iron in silicates) are less efficient at removing (i.e., reacting with) sulfide. As a result, sulfide then begins to accumulate in the pore waters (Canfield et al., 1992; Raiswell and Canfield, 1998).

5.09.8.1 Pyrite Formation in Sediments

In sediments where sulfate reduction occurs, pyrite (FeS₂) is the dominant form of solid-phase sulfur. Pyrite has a range of crystal morphologies, although in marine sediments only a limited number are known to occur, with single euhedra and framboids generally dominating (Canfield and Raiswell, 1991b; Goldhaber, 2003; Rickard and Luther, 2007).

Other sulfide minerals found in marine sediments include amorphous-FeS and mackinawite (FeS_{1-x} $x \le -0.1$). Both are commonly referred to in the literature as FeS. Another metastable sulfide phase is greigite (Fe₃S₄), the sulfur analog to magnetite (Fe₃O₄). All of these phases are soluble in nonoxidizing mineral acids such as HCl and are often referred to as acid volatile sulfide (AVS). Pyrite is not soluble in HCl and requires more stringent (i.e., oxidizing or reducing) conditions to be solubilized. These differences in solution chemistry have led to the development of selective leaching techniques, which at least operationally allow one to separately quantify AVS and pyrite in marine sediments (Canfield et al., 1986; Cornwell and Morse, 1987; Morse et al., 1987).



Figure 10 Pore-water sulfate and methane depth profiles from sediments in Saanich Inlet, British Columbia. From Devol, A.H., Anderson, J.J., Kuivala, K., Murray, J.W., 1984. A model for coupled sulfate reduction and methane oxidation in the sediments of Saanich Inlet. Geochimica et Cosmochimica Acta 48, 933–1004.

The black color that is common in many organic-rich coastal sediments is due to the presence of these sulfide phases. AVSs are metastable phases that will eventually transform into pyrite, the thermodynamically stable sulfide phase under early diagenetic conditions (e.g., Rickard and Luther, 2007). Another form of sulfur found in marine sediments is solid elemental S°, which forms during sulfide oxidation, and also plays an important role in pyrite formation.

Because FeS is metastable, once it is produced it is either reoxidized (see below) or transformed into pyrite. The oxidation of FeS in sediments, along with dissolved sulfide and pyrite, can occur either chemically or microbially, using oxidants such as O_{2i} nitrate or metal oxides (see Section 5.09.2.2). In mixed redox sediments, these processes are mediated by a whole host of transport processes (e.g., bioturbation, bioirrigation, or macrophyte O_2 input; e.g., see Section 5.09.3); in their absence (e.g., Section 5.09.4.1) these oxidation reactions may occur as a result of the downward diffusion of oxygen or nitrate into the sulfidic sediments. The occurrence of these reactions plays a role in the extent to which sulfides produced by bacterial sulfate reduction are retained in the sediments, that is, buried as reduced sulfur. At the same time though, because the disulfide in pyrite (FeS₂) is more oxidized than its initial source, for example, sulfide in either dissolved sulfide or FeS, these oxidative processes also play a role in pyrite formation, and hence have an opposite effect on sulfur burial in sediments.

Experimental studies examining pyrite formation suggest three possible mechanisms for pyrite formation in marine sediments. In all three cases, FeS is the initial precursor for pyrite formation, because of the ease with which it forms by the reaction of dissolved sulfide with either Fe²⁺or iron oxides, and because direct precipitation of pyrite from solution appears to be kinetically inhibited (Schoonen and Barnes, 1991). A detailed discussion of these mechanisms can be found in recent reviews (Goldhaber, 2003; Rickard and Luther, 2007).

One mechanism for pyrite formation involves the addition of the zero-valent sulfur in dissolved polysulfides (S_x^{2-}) to FeS by a dissolution/reprecipitation reaction (Luther et al., 1991; Rickard et al., 1995). A second mechanism can be thought of as pyrite formation through iron loss (coupled to FeS oxidation), as opposed to sulfur addition, through a greigite intermediate,

$$4\text{FeS} \rightarrow \text{Fe}_3\text{S}_4 \rightarrow 2\text{FeS}_2$$
 [5]

(Sweeney and Kaplan, 1973; Schoonen and Barnes, 1991; Wilkin and Barnes, 1996; Morse and Wang, 1997; Benning et al., 2000). A number of possible oxidizing agents, in addition to zero-valent forms of sulfur, may play a role in this type of pyrite formation. These include other sulfur species with intermediate oxidation states (e.g., thiosulfate) as well as O₂ itself. The factors controlling the occurrence of these two possible modes of pyrite formation are not well understood (see discussions in Schoonen and Barnes, 1991; Wang and Morse, 1996; and Goldhaber, 2003).

Given the potential oxidants required in these two modes of pyrite formation, their occurrence is consistent with the fact that much of the pyrite that forms in marine sediments does so close to the sediment–water interface, where rates of sulfate reduction are usually the highest and the availability of reactive iron is likely to be the greatest, that is, in association with mixed redox conditions in these surface sediments (Chanton et al., 1987; Canfield et al., 1992; Goldhaber, 2003). Evidence in support of these oxidative processes and their role in pyrite formation comes, in part, from a wide range of sulfur isotope studies in the laboratory and in the field (Canfield and Thamdrup, 1994; Habicht and Canfield, 1997; Habicht et al., 1998; Habicht and Canfield, 2001; Sørensen and Canfield, 2004).

Pyrite formation also occurs deeper in sediments, where dissolved sulfide levels are high (Figure 6) and the oxidants discussed above (i.e., O_2 , intermediate oxidation state sulfur species, or reactive Fe(III) oxides) are likely to be in very low concentrations or nonexistent. Here a third mechanism of pyrite formation may be

$$FeS(s) + H_2S \rightarrow FeS_2(s) + H_2(g)$$
[6]

where H_2S now serves as the oxidizing agent of FeS, and reduced hydrogen gas is produced as an end product (Rickard, 1997). Although this reaction has a negative ΔG° (Wilkin and Barnes, 1996), experimental evidence in support and against its occurrence has been presented (Rickard et al., 1995; Wilkin and Barnes, 1996; Rickard, 1997; Benning et al., 2000). It is interesting to note, however, that H_2 is an excellent substrate for both sulfate reducers and methanogens, and is generally found at nM levels in anoxic marine sediments (Hoehler et al., 1998). Therefore in the region of the sediments where reaction (6; **Table 2**) is likely to be most important, the low hydrogen levels observed there could enhance the thermodynamic driving force for the occurrence of this reaction.

5.09.8.2 Pyrite Burial and Sulfur Burial Efficiency

Much of the pyrite found in marine sediments appears to form as a result of the interaction between oxidative, or mixed redox, processes and bacterial sulfate reduction. The occurrence of these same processes also leads to only a small fraction of the total sulfide produced by sulfate reduction eventually being buried as pyrite. Expressed another way, sulfur (~pyrite) burial at depth is generally much smaller than the integrated gross rate of sulfate reduction; the ratio of the two is thus defined as the sulfur burial efficiency. Historically, oxygen has been assumed to be the predominant oxidant of pyrite (Morse, 1991). However, recent studies have also shown that other substances, such as nitrate, manganese, and iron oxides, may oxidize pyrite anaerobically by both biotic and abiotic processes (Luther and Church, 1992; Luther et al., 1992; Schippers and Jørgensen, 2002). As is also the case for anaerobic FeS oxidation (see Section 5.09.2.2), evidence to date regarding chemical versus microbial mediation of anaerobic pyrite oxidation is equivocal.

In most sediments sulfur burial efficiency is generally less than ~10-20% (Jørgensen, 1982b; Berner and Westrich, 1985; Chanton et al., 1987; Canfield, 1989; Swider and Mackin, 1989), although there are examples of sediments with much higher burial efficiencies (e.g., in Cape Lookout Bight sediments sulfur burial efficiency is 77%). Sediments with such high sulfur burial efficiencies are often sites with high sedimentation rates, high rates of sulfate reduction, and are also generally not extensively bioturbated or bioirrigated (Canfield, 1994; Goldhaber, 2003; also see Section 5.09.4.1). A high sulfur burial efficiency can also occur at low rates of sulfate reduction if large amounts of reactive iron are present in the sediments (Marvin-DiPasquale and Capone, 1998). This observation points to the fact that the availability of sulfate, reactive iron, and metabolizable organic matter are all important parameters that control sulfur burial efficiency, and that of these, the availability of these last two quantities appears to be most critical. A detailed discussion of this topic is beyond the scope of this article, although any of the references cited above provide a good discussion of the topic (also see Berner, 1970; Morse and Berner, 1995).

In estuarine sediments, sulfate availability has the potential to limit sulfate reduction as one moves upstream into the low salinity reaches of an estuary. However, in actuality this is often not the case. Sulfate-reducing bacteria (both in lab and field studies) have a high affinity (low half saturation constant or K_m) for sulfate (Boudreau and Westrich, 1984; Ingvorsen and Jørgensen, 1984; Roden and Tuttle, 1993), suggesting that sulfate reduction will be limited by sulfate availability only at concentrations below ~2–3 mM, which will roughly occur at $S \approx 2.5-4$. Nevertheless, as one moves into the very low salinity upper reaches of an estuary, the importance of sulfate reduction decreases, with a concomitant increase in the importance of methane production (Kelley et al., 1990; Middelburg et al., 1996a). This point is discussed in more detail in the next section.

In contrast, other factors such as organic carbon availability and reactivity do appear to play a role in regulating sulfate reduction in estuarine and coastal sediments (Westrich and Berner, 1984; Berner and Westrich, 1985; Marvin-DiPasquale and Capone, 1998). In sediments with a low sulfur burial efficiency, the effective recycling of sulfate in these sediments further minimizes the possibility of sulfate limitations on sulfate reduction. As discussed in Section 5.09.2.3 and illustrated in Figure 1(a), here sulfate largely serves as an electron shuttle for transporting oxidation equivalents into the sediments that ultimately are used to remineralize sedimentary organic carbon. At the same time though, in extremely organic-rich sediments sulfate reduction can become sulfate limited at depths where sulfate concentrations go to zero and methanogenesis begins to occur (Reeburgh, 1969; Martens and Berner, 1977; Devol et al., 1984; Martens and Klump, 1984; Martens et al., 1998). The depth in the sediments at which this occurs plays an important role in regulating methane gas loss from these sediments (see the next section for details).

5.09.9 Methanogenesis, Methane Oxidation, and Methane Fluxes to the Atmosphere

Upon complete sulfate depletion in sediments, methanogenesis occurs below the zone of sulfate reduction, consistent with the concepts of biogeochemical zonation discussed in Section 5.09.2.1. Given the high concentration of sulfate in seawater, the occurrence of methanogenesis in shallow marine sediments requires a relatively large flux of reactive organic matter to the sediments. As a result, methanogenesis is generally responsible for a small amount of sediment carbon remineralization. Globally, organic carbon remineralization by methanogenesis in sediments is ~12% that of organic carbon remineralization by sulfate reduction (Henrichs and Reeburgh, 1987), although in some sediments methanogenesis can be more important. For example, in Cape Lookout Bight sediments, organic carbon remineralization by methanogenesis is ~43% that of remineralization by sulfate reduction (Martens and Klump, 1984).

In estuaries, the salinity decrease that occurs as one moves up the estuary also decreases the estuarine sulfate concentration. With this, the necessary carbon flux to the sediments required to achieve complete sulfate depletion also decreases. Thus as one moves into the very low salinity upper reaches of an estuary, there is an increase in the importance of methanogenesis to total sediment organic carbon remineralization (e.g., Kelley et al., 1990; Middelburg et al., 1996a).

In many marine sediments, much of the methane produced during methanogenesis is oxidized anaerobically, apparently using sulfate as the electron acceptor. This reaction, termed anaerobic oxidation of methane (AOM), can be written as

$$CH_4 + SO_4^{2-} \rightarrow HCO_3^{-} + HS^{-} + H_2O$$
^[7]

which is thermodynamically favored under typical *in situ* conditions ($\Delta G = -25$ kJ mol⁻¹; Martens and Berner, 1977). While geochemical evidence since the 1970s has been consistent with the occurrence of AOM (Reeburgh, 1982; Alperin and Reeburgh, 1985), a complete understanding of the microbial processes involved in AOM remained elusive for a number of years (for a discussion of this history, see Canfield et al., 2005). However, it now appears that AOM is mediated by a consortium of sulfatereducing bacteria and methanogens, through a process that has been termed 'reverse methanogenesis' (Hoehler et al., 1994; Boetius et al., 2000; Orphan et al., 2001; Strous and Jetten, 2004; Alperin and Hoehler, 2009).

In marine and estuarine sediments, AOM occurs in the transition zone between sulfate reduction and methanogenesis, as sulfate concentrations approach zero, and it often leads to characteristic concave-up methane pore-water profiles (see Figure 10; Martens and Berner, 1977; Reeburgh, 1982; Devol et al., 1984; Alperin and Reeburgh, 1985; Iversen and Jørgensen, 1985). In these situations, AOM serves as a quantitative sink for virtually all of the methane that is produced in the sediments (Henrichs and Reeburgh, 1987; Reeburgh, 2007). The process therefore significantly limits the global importance of the oceans as a source of methane to the atmosphere.

Methane produced in sediments that escapes oxidation by sulfate can also be oxidized aerobically by methanotrophic bacteria (Hanson and Hanson, 1996), and in low salinity or tidal freshwater systems this process can similarly play an important role in regulating the methane flux from these sediments. A microbial consortium capable of anaerobic methane oxidation at the expense of denitrification has also been described (Raghoebarsing et al., 2006). Like AOM coupled to sulfate reduction this reaction is thermodynamically favored, although virtually nothing is known about the biogeochemical significance of this process.

Although sediment systems are, for the most part, very efficient, at recycling methane that is produced at depth, methane can escape sediments by gas bubble ebullition, transport mediated by macrophytes, or upward diffusion across the sediment–water interface. Given the importance of methane as a greenhouse gas, an understanding of the factors controlling methane fluxes to the atmosphere is important for both quantifying these fluxes along with improving our understanding of how these gas fluxes may respond to climate change (Whiting and Chanton, 2001; Middelburg et al., 2002; also *see* Chapter 5.04). Diffusion can play an important role in methane loss from sediments in conjunction with either of the other two transport processes discussed here (Chanton et al., 1989; Kelley et al., 1990). However, in strictly diffusive systems (i.e., those without plants or gas ebullition), there appears to be a fairly efficient coupling between methane production and oxidation in the sediments, with small net methane losses from the sediments (Martens et al., 1998). This is controlled largely by the interplay between the microbial and physical processes occurring in such sediments (also see Chanton and Dacey, 1991).

In sediments where large pore-water methane concentrations build up, gas bubble can form in the sediments when the sum of the partial pressures of all dissolved gases exceeds the local hydrostatic pressure (Chanton and Whiting, 1995). This often, but not always, occurs when the sediment pore waters become saturated with respect to methane. Not only do these bubbles cause what is referred to as acoustic turbidity (Wever et al., 1998), but the bubbles can rise through the sediments (ebullition), resulting in a flux of gaseous methane from the sediments. In the process of gas bubble formation, growth, and ebullition, the bubbles strip other dissolved gases out of the pore waters, including ²²²Rn (which can be used as a tracer for bubble formation), CO2, N2, and H2S and other reduced sulfur gases (Chanton and Whiting, 1995). In subtidal sediments, methane gas ebullition forms cylindrical bubble tubes that lead to enhanced rates of sediment-water exchange through the creation of additional sediment surface area for benthic exchange (Martens et al., 1980). Gas ebullition from methanogenic sediments is of importance beyond its effect on sediment-water exchange because it transports methane, and possibly other reduced gases produced in anoxic sediments, directly to the atmosphere, bypassing their possible oxidation in the sediments or water column (Kipput and Martens, 1982; Martens and Chanton, 1989; see Chapter 5.04).

Gas ebullition is common in many tidal freshwater and wetland soils and sediments, as well as in some subtidal coastal sediments, such as Cape Lookout Bight and Long Island Sound (Martens and Berner, 1977; Kipput and Martens, 1982; Chanton et al., 1989; Chanton and Dacey, 1991; Albert et al., 1998; Martens et al., 1998). In many of these environments, gas ebullition occurs seasonally, during warmer months when there are higher rates of sediment organic carbon remineralization, and concomitantly higher inventories of gas in the sediments (Chanton et al., 1989). More specifically, bubble ebullition appears to be driven by a decrease in the local hydrostatic pressure associated with low tides (Martens and Klump, 1980; Chanton et al., 1989). However, a more detailed explanation of the linkages here has not yet been presented (see recent discussions in Algar and Boudreau, 2009). Gas ebullition from sediments can also be triggered by discrete wind events or physical disturbance of the sediments (Chanton and Whiting, 1995).

In vegetated sediments, the presence of plants can impact methane fluxes from the sediments in that the plants serve as conduits for methane exchange with the atmosphere, using many of the same mechanisms that allow them to transport O_2 into the sediments (see Section 5.09.4.5 and Chanton and Dacey, 1991). Plant O_2 input also allows for the occurrence of belowground aerobic methane oxidation, while belowground organic carbon production and root exudation by the plants can stimulate methanogenesis (again see related discussions in Section 5.09.4.5). Many of these same processes can similarly stimulate iron redox cycling in freshwater-vegetated wetland sediments (Roden and Wetzel, 1996; van der Nat and Middelburg, 1998), and the resulting stimulation of organotrophic iron reduction then inhibits methanogenesis in the sediments (Section 5.09.7) and lowers the methane flux from these sediments. In general, the positive factors (plant gas exchange and stimulated belowground production) appear to more important than the negative factors (belowground methane oxidation, the inhibition of methanogenesis by iron redox cycling), since methane fluxes often show a positive correlation with plant biomass or photosynthetic activity (Chanton and Whiting, 1995). However, as Roden and Wetzel (1996) also noted, the impact of these different controlling factors (and the relevant times scales of operation) may vary among different types of wetland sediments.

5.09.10 Trace Metal Cycling

The geochemistry of many trace metals and trace elements in estuarine and coastal marine sediments is strongly influenced by many of the processes discussed throughout this article. In particular, the redox cycling of iron and manganese oxides is of importance here, as is the redox cycling of iron sulfide minerals (e.g., FeS or pyrite).

Linkages between Fe and Mn redox cycling and trace metal cycling exist in part because Fe and Mn oxides and iron sulfides are enriched in a wide range of trace metals (e.g., Cu, Ni, Zn, Cd, Mo, and Co). The processes that lead to these enrichments vary, and include simple adsorption, direct incorporation of these metals into crystal structures, and coprecipitation, for example, of trace metal sulfides in conjunction with iron sulfides (Balistrieri and Murray, 1986; Morse and Luther, 1999; Glasby, 2000; Li, 2000). Furthermore, many Mn and Fe oxides occur either as coatings on other particles or as fine-grained amorphous solids, and therefore tend to have large specific surface areas. Thus for surface processes such as adsorption, small amounts of these oxides can play an important role in controlling the geochemical cycling of trace metals.

For many of the trace elements discussed here, their association with Mn and Fe oxides can lead to these elements undergoing indirect (or passive) redox cycling in association with Mn and Fe redox cycling (Elderfield et al., 1981b; Salwan and Murray, 1983; Westerlund et al., 1986; Shaw et al., 1990; Shimmield and Pedersen, 1990; Audry et al., 2006; and others). Here, changes in redox conditions do not generally change the oxidation state or speciation of these trace elements, nor directly affect their solubility.

In contrast, there are other trace elements (e.g., V, Re, U) that do have multiple oxidation states and different solubilities associated with their different redox states. In general, the oxidized forms are soluble in seawater while the reduced forms are not. Furthermore, the more active redox cycling of these elements in sediments shows some broad similarities to that seen for sediment Mn and Fe redox cycling, that is, the redox transformations occur when O_2 concentrations equal zero, at redox potentials consistent with suboxic or anoxic conditions (Shaw et al., 1990; Morford and Emerson, 1999;

McManus et al., 2006; Morford et al., 2009). Sediment profiles of these redox-sensitive metals have the potential to be used as tracers (across a range of timescales) of past changes in sediment redox conditions due to variations in either bottomwater oxygen concentration or the organic carbon flux to the sediments (Calvert and Pedersen, 1993; Adelson et al., 2001; Crusius and Thomson, 2003; McManus et al., 2006).

The uptake of trace metals into AVS and pyrite also plays an important role in their cycling in estuarine and coastal sediments given the importance of sulfate reduction and sulfide production in these sediments (as compared to, e.g., freshwater or deep-sea sediments). Examining the uptake of an operationally defined reactive metal fraction into pyrite has been done by examining what is termed the degree of trace metal pyritization, or DTMP (Huerta-Diaz and Morse, 1992), a parameter defined in the same manner as the degree of pyritization (DOP) is for iron. DOP is defined as

$$DOP = \frac{pyrite-Fe}{pyrite-Fe + 'reactive'-Fe}$$
[8]

where reactive-Fe is iron that is soluble in HCl (i.e., pyrite is not soluble in HCl), and it is likely to include iron in phases whose reactivity toward sulfide ranges from hours to perhaps $\sim 10^4 - 10^5$ years (Raiswell and Canfield, 1998; Goldhaber, 2003). DTMP is defined in an analogous sense based on the concentration of metals co-extracted with the reactive-Fe and pyrite-Fe fractions.

Comparisons of DTMP and DOP in a large number of sediments (also see recently Scholz and Neumann, 2007) have proved to be useful in terms of understanding the degree to which various metals are incorporated into pyrite relative to the formation of pyrite itself (Figure 11). Such studies have observed the following general pattern of DTMP values for these metals:

$$\begin{split} Hg > As = Mo > Cu = Fe > Co > Ni >>Mn > Zn > Cr \\ = Pb > Cd \end{split}$$
 [9]



Figure 11 A schematic representation of DTMP (degree of trace metal pyritization) vs. DOP (degree of pyritization) for selected trace metal (see the text for further details). Modified from Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton based on results presented in Huerta-Diaz, M.A., Morse, J.W., 1992. Pyritization of trace metals in anoxic marine sediments. Geochimica et Cosmochimica Acta 56, 2681–2702 and Morse, J.W., Luther, G.W., III, 1999. Chemical influences on trace metal–sulfide interaction in anoxic sediments. Geochimica et Cosmochimica Acta 63, 3373–3378.

While sulfidic sediments can be an important sink for all of these trace metals (see discussions below), DTMP values indicate whether this uptake occurs in metal (Me) monosulfide phases (MeS) versus pyrite phases, or phases that are operationally extracted with pyrite. This has some impact on the stability of these metals in anoxic or mixed redox sediments (see discussions at the end of this section for further details).

Morse and Luther (1999) have examined the thermodynamic and kinetic controls on the observed trends in eqn [9] using the rate of water exchange for soluble water-metal complexes as a first-order proxy for the relative reactivity of a metal to form a MeS phase. This approach explains many, but not all, of the observations in eqn [8] and in Figure 11. For elements such as As and Mo, which show high DTMP values that are almost independent of DOP, they are first reduced by sulfide (i.e., As⁵⁺ to As³⁺ and Mo⁶⁺ to Mo⁴⁺) before forming sulfide phases. They then undergo a more complex set of redox reactions than the simple substitution reactions discussed above, and these reactions appear to favor strongly their co-precipitation with pyrite. High DTMP values for Hg may be a function of the more complicated Hg solution chemistry that appears to favor Hg incorporation into pyrite. At the same time, because HgS is extremely insoluble in HCl (Cooper and Morse, 1998b), it will also operationally be extracted with the pyrite fraction.

Near the sediment-water interface in subtidal sediments with oxic bottom waters, trace metal dynamics are largely controlled by processes such as regeneration from biogenic (organic matter) carrier phases, uptake into (or in association with) diagenetically produced Mn and Fe oxides, and release from (or in association with) the reductive dissolution of Mn and Fe oxides (see discussions, and summaries, most recently in Burdige, 2006 and Martin, 2009). In many cases, this may lead to the release (benthic flux) of dissolved metals from sediments, which can play an important role in affecting dissolved metal concentrations in shallow water coastal ecosystems (Westerlund et al., 1986; Skrabal et al., 1997; Colbert et al., 2001; Warnken et al., 2001; Caetano et al., 2007; Point et al., 2007 and references therein). At the high anthropogenic metal levels that are often be found in some nearshore sediments, benthic metal fluxes may play some role in controlling the toxicity and bioavailability of these metals to water-column organisms (Riedel et al., 1997; Shine et al., 1998). This observation is in contrast to that observed in open ocean settings, where the availability of certain trace metals can limit phytoplankton growth (Morel et al., 2003).

The processes discussed above can also be examined in the broader context of a conceptual model of trace metal and trace element cycling in continental margin sediments (including estuarine and coastal sediments) that is shown in Figure 12. The figure is presented in a generic sense and thus different aspects of the processes shown here may be more or less important for a specific trace metal. In addition, in many cases the reactions of interest are actually redox reactions directly between Mn and Fe oxides and iron sulfides, which can be mediated by macrofaunal activity under mixed redox conditions (see Section 5.09.2.2).

Since most of the reactions in this figure are dissolution/ reprecipitation processes, the cycling of these metals generally goes through a dissolved phase. Furthermore, as trace metals move between oxide and sulfide phases during sediment redox cycling (or are regenerated from organic carrier phases), the



Figure 12 A conceptual model for trace metal cycling in coastal and nearshore sediments, presented in terms of the major sediment carrier phases of these metals (e.g., oxides and sulfides). Although not explicitly shown here, some of the relevant redox reactions these sulfides and oxides undergo involve both as reactants (e.g., see reaction 7 in Table 2). Modified from Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton.

release from one phase may be slower than uptake by the other phase, or the release and uptake processes may be separated from one another within the sediment column. Under these conditions, metals can accumulate in sediment pore waters. Such considerations will then also play a role in regulating the magnitude and direction of benthic fluxes of trace metals from sediments. Metals in sediments can also be returned to the water column by sediment resuspension (Saulnier and Mucci, 2000; Kalnejais et al., 2007). However, the bioavailability of particle-bound metals to water-column organisms likely depends on the release into solution (e.g., desorption, solubilization, dissolution) of the metals from the solid phase (Chen and Mayer, 1999).

Organic complexation of trace metals is also likely to occur in pore waters (Elderfield, 1981; Skrabal et al., 2000; Skrabal et al., 2006). This complexation might be expected to be of greatest importance in anoxic sediments where sulfate reduction occurs, due to the high DOC levels generally observed in such sediments (Burdige, 2002). The potential importance of this metal complexation in pore waters has been discussed in terms of its impact on metal benthic fluxes (also see Skrabal et al., 1997), although it also appears to play a role in the solubilization and precipitation of metals in sediments (Luther et al., 1992).

Bottom-water oxygen concentration, carbon rain rate, and benthic macrofaunal activity all play important roles in regulating benthic trace metal fluxes, in part, because of their impact on iron, manganese, and sulfur redox cycling in sediments (see references discussed below and Sundby et al., 1986; Shaw et al., 1990; Aller, 1994b; and Riedel et al., 1997). In particular, factors such as these control the sediment oxygen penetration depth (e.g., Glud, 2008), and the relationship between the timescale of transport though this oxic surface layer, versus the timescale of either oxidative removal (for Mn and Fe) or uptake onto freshly formed oxides (for the other trace metals) will then determine the extent to which metals regenerated at depth are retained in the sediments, or released as a benthic flux. Because Fe oxidation is much faster than that of Mn (Section 5.09.2.1), such processes are more likely to lead to Mn versus Fe benthic fluxes (e.g., Balzer, 1982). This would also suggest that there could be some divergence in the

behavior of trace elements more strongly associated with Fe versus Mn oxides.

Remobilization of metals near the sediment surface can also lead to the downward diffusive flux of regenerated metals into the deeper, anoxic sediments (**Figure 6**; also see, e.g., Cu profiles in Skrabal et al., 2000). Here the uptake of some of these metals into sulfide phases (as outlined above) or sometimes carbonate phases (mostly Mn; see Holdren et al., 1975; Suess, 1979; Aller, 1980b; Middelburg et al., 1987) will be important. In the absence of bioturbation, these sulfides will not be reoxidized and will simply be buried in the sediments.

The relative balance of the diagentic processes discussed above can also, under some circumstances, lead to sediments taking up certain metals from the water column. For example, Colbert et al., 2001 demonstrated with benthic flux measurements that coastal and continental margin sediments go from being a source of Cd to the water column to being a sink for water-column Cd with increasing rates of sediment sulfate reduction. The sediment uptake of Cd at high sulfate reduction rates appears to be related to the precipitation of the highly insoluble CdS in the sediments. At low rates of sulfate reduction, other oxidative processes apparently contribute to the net release of dissolved Cd from the sediments. Earlier studies by Elderfield et al. (1981a, 1981b) similarly suggested that anoxic sediments could be sinks for both Cd and Ni (which also forms a highly insoluble sulfide), based on both pore-water profiles and measured benthic fluxes.

The occurrence of bioturbation and/or bioirrigation, and the resulting mixed redox conditions, has a more complex effect on sediment trace metal cycling and benthic fluxes. Metals that are sequestered as sulfides may be mixed upward into surficial portions of the sediments, where they can be reoxidized by a number of dissolved and solid oxidants, for example, O_2 , nitrate, and manganese and iron oxide. Since this oxidation liberates, at least in a transient sense, metals into the pore waters, their potential flux out of the sediments may be enhanced. The diffusive openness of bioirrigated sediments (Section 5.09.4.1) also provides an means of enhancing the transport of dissolved oxidants into the sediments and, as a result, allows metals that are remobilized/solubilized by oxidative processes a greater likelihood of escaping the sediments as a benthic flux (e.g., Emerson et al., 1984). Consistent with these suggestions, in coastal sediments the magnitude of benthic fluxes of metals such as Cd, Cu, Zn, and Ni (and in some cases also the direction of the flux) is strongly affected by the occurrence of these oxidative processes (Westerlund et al., 1986; Gobeil et al., 1987; Riedel et al., 1997; Skrabal et al., 1997; Shine et al., 1998; Colbert et al., 2001). Of equal importance is the fact that metal sulfide production at depth in sediments does not necessarily imply permanent burial of metals.

However, O_2 input through the burrow walls of irrigating macrofauna, and macrophyte roots as well, results in localized precipitation of Mn and Fe iron oxides in regions immediately adjacent to these subsurface O_2 sources (Aller and Yingst, 1978; Aller, 1988; St-Cyr et al., 1993; Sundby et al., 1998; Vale and Sundby, 1998). Thus, in a similar fashion to oxides which form in nonirrigated sediments near the sediment–water interface and may act as a barrier for benthic fluxes of trace metals out of the sediments (i.e., see the discussion above and compare **Figures 2(a)** and **2(b)**, the same may be true here for trace metal loss in association with macrofaunal burrow irrigation (Berg et al., 2003; Meile et al., 2005). (Recall based on previous discussions that in the context of these figures, metal oxide precipitation is likely to occur in (or near) the regions marked O_2 and NO_3^- .)

Metals that are remobilized and solubilized at depth in association with the reduction of manganese and iron oxides show somewhat differing behaviors under mixed redox conditions. For Mn, for example, it has been observed that while mixed redox conditions promote greater internal redox cycling of manganese (e.g., **Figure 1(b)**), benthic Mn fluxes do not appear to be greatly enhanced (Sundby et al., 1986; Aller, 1990, 1994b; Aller and Aller, 1998).

Low dissolved oxygen in bottom waters and/or increased carbon fluxes to the sediments will enhance benthic Mn fluxes along with the fluxes of other metals (e.g., Co) that are associated with both Mn and Fe oxides (Sundby et al., 1986). Seasonal studies of benthic Mn fluxes also show that benthic Mn fluxes increase with increasing temperature (Aller and Benninger, 1981; Elderfield et al., 1981a; Hunt, 1983). However, identifying the causes of such seasonal trends is hampered by the fact that changes in temperature, bottomwater oxygen, organic carbon input to the sediments, and sediment O₂ penetration depth often co-occur, although not always by direct causality (see discussion in Aller, 1994b). In a related fashion, studies of iron fluxes from California continental margin sediments suggest that these fluxes are only significant for sediments with shallow oxygen penetration depths, that is, less than ~20 mm, and/or those underlying low bottom-water O_2 concentrations of less than ~20 μ M (McManus et al., 1997).

Deposit feeding organisms living in sediments are also capable of solubilizing metals such as Cu, Cd, and Pb from contaminated sediments during their passage through macrofaunal digestive systems (Mayer et al., 1996; Chen and Mayer, 1998). This process is mediated by soluble proteins in the animal's digestive fluids, and thus a large fraction of these solubilized metals are also complexed. This process increases the bioavailability of the metals to the deposit feeding organism itself, and can therefore lead to metal accumulation in the organism; at the same time, it may also lead to the release of these metals into the pore waters.

Seasonal variations of redox conditions in coastal sediments affect the sediment cycling and benthic fluxes of many other trace metals in addition to Mn (e.g., Hines et al., 1984; Aller, 1994b; Shine et al., 1998). The results of Cooper and Morse (1998a) further suggest that for metals which form highly insoluble sulfides (e.g., Cu and Zn in their study), both the amounts and seasonal changes in the sediment AVS content play an important role in regulating seasonal sediment metal cycling. Based on comparative studies of two sites in Chesapeake Bay, they suggest that if sediment Fe in AVS is less than ~20% of HCl-soluble Fe, then seasonal changes in AVS content can affect the cycling of these other metals. In contrast, when Fe in AVS is greater than ~20% of the reactive Fe, there appears to be enough sulfide available in the sediments to trap effectively these metals and buffer temporal changes in metal dynamics.

Such factors may also play a role in explaining metal toxicity to organisms living in coastal sediments. For metals that show a low DTMP, and which form highly insoluble monosulfides (e.g., Cd, Ni, Zn, Pb, and Cu), the molar ratio of the acid (HCl)-extractable metal to AVS is strongly correlated with sediment metal toxicity (see DiToro, 2001 and references therein). When this ratio is less than 1, the excess AVS in the sediment apparently titrates all of the available metal, such that no free metal is available to sediment infauna. In contrast, as this ratio increases, sulfide availability limits metal uptake as a sulfide precipitate and thus makes metals more available to organisms.

Although pyrite is the ultimate burial phase for most sulfur in estuarine and coastal sediments, the low sulfur burial efficiency (Section 5.09.8.2) implies that both pyrite and AVS undergo extensive cycling in surface sediments. While AVS appears to be more reactive than pyrite with respect to oxidation (Morse, 1991), pyrite oxidation does occur on early diagenetic timescales (again see Section 5.09.8.2). Laboratory studies by Morse (1994) have shown that pyrite oxidation releases significant amounts of metals such as As, Cu, Hg, and Mo, all of which generally show high values of DTMP. More importantly, the percentage loss of these metals from the pyrite fraction is similar to or greater than the percentage loss of Fe during pyrite oxidation. Thus, pyrite oxidation, driven by sediment resuspension or *in situ* sediment redox processes, also has the potential to increase the bioavailability of pyritized metals.

5.09.11 Concluding Remarks

A variety of physical and biological forcing functions act on estuarine and coastal sediments across a broad range of time and space scales, and can have a profound effect on the processes associated directly or indirectly with the remineralization of sedimentary organic carbon. These interactions exert a strong influence on the cycling of biogeochemically reactive elements, and result in linkages between and within elemental cycles (e.g., between the Mn and N cycle, or simply with the N cycle itself) that are far more complex than previously envisioned. There is still much to be learned here, and many important questions await answers. The author's hope is that this chapter accurately presents the current state of knowledge and sets the stage for this future work. There has also been the increased recognition that biogeochemical processes in estuaries and the coastal ocean (and their underlying sediments) can have large-scale impacts on global ocean processes. Thus, studies of these processes have implications not only in terms of the internal dynamics and elemental budgets of estuarine and coastal sediments, but also on benthic-pelagic coupling, and perhaps global oceanic cycles and budgets of biogeochemically active elements. Again, future studies in this area will better address these topics as well.

References

- Abril, G., Etcheber, H., Hir, P.L., Bassoullet, P., Boutier, B., Frankignoulle, M., 1999. Oxic/anoxic oscillations and organic carbon mineralization in an estuarine maximum turbidity zone (The Gironde, France). Limnology and Oceanography 44, 1304–1315.
- Abril, G., Riou, S.A., Etcheber, H., Frankignoulle, M., de Wit, R., Middelburg, J.J., 2000. Transient, tidal time-scale, nitrogen transformations in an estuarine turbidity maximum – fluid mud system (the Gironde, South-west France). Estuarine, Coastal and Shelf Science 50, 703–715.
- Adelson, J.M., Helz, G.R., Miller, C.V., 2001. Reconstructing the rise of recent coastal anoxia; molybdenum in Chesapeake Bay sediments. Geochimica et Cosmochimica Acta 65. 237–252.
- Albert, D.B., Martens, C.S., Alperin, M.J., 1998. Biogeochemical processes controlling methane in gassy coastal sediments – part 2: groundwater flow control of acoustic turbidity in Eckernförde Bay sediments. Continental Shelf Research 18, 1771–1793.
- Algar, C.K., Boudreau, B.P., 2009. Transient growth of an isolated bubble in muddy, finegrained sediments. Geochimica et Cosmochimica Acta 73, 2581–2591.
- Aller, J.Y., Aller, R.C., 1986. General characteristics of benthic faunas on the Amazon inner continental shelf with comparison to the shelf off the Changjiang River, East China Sea. Continental Shelf Research 6. 291–310.
- Aller, J.Y., Aller, R.C., 2004. Physical disturbance creates bacterial dominance of benthic biological communities in tropical deltaic environments of the Gulf of Papua. Continental Shelf Research 24, 2395–2416.
- Aller, R.C., 1980a. Diagenetic processes near the sediment–water interface of Long Island Sound, I. Decomposition and nutrient element geochemistry (S, N, P). Advances in Geophysics 22, 235–348.
- Aller, R.C., 1980b. Diagenetic processes near the sediment–water interface of Long Island Sound, II. Fe and Mn. Advances in Geophysics 22, 351–415.
- Aller, R.C., 1980c. Quantifying solute distributions in the bioturbated zone of marine sediments by defining an average microenvironment. Geochimica et Cosmochimica Acta 44, 1955–1965.
- Aller, R.C., 1982. The effects of macrobenthos on chemical properties of marine sediments and overlying water. In: McCall, P.L., Tevesz, M.J.S. (Eds.), Animal– Sediment Relations, the Biogenic Alteration of Sediments. Plenum, New York, NY, pp. 53–102.
- Aller, R.C., 1988. Benthic fauna and biogeochemical processes in marine sediments: the role of burrow structures. In: Blackburn, T.H., Sørensen, J. (Eds.), Nitrogen Cycling in Coastal Marine Environments. Wiley, Chichester, pp. 301–338.
- Aller, R.C., 1990. Bioturbation and manganese cycling in hemipelagic sediments. Philosophical Transactions of the Royal Society A 331, 51–68.
- Aller, R.C., 1994a. Bioturbation and remineralization of sedimentary organic matter: effects of redox oscillation. Chemical Geology 114, 331–345.
- Aller, R.C., 1994b. The sedimentary Mn cycle in Long Island Sound: its role as intermediate oxidant and the influence of bioturbation, O₂, and C_{org} flux on diagenetic reaction balances. Journal of Marine Research 52, 259–295.
- Aller, R.C., 1998. Mobile deltaic and continental shelf muds as suboxic, fluidized bed reactors. Marine Chemistry 61, 143–155.
- Aller, R.C., 2001. Transport and reactions in the bioirrigated zone. In: Boudreau, B.P., Jorgensen, B.B. (Eds.), The Benthic Boundary Layer. Oxford University Press, Oxford, pp. 269–301.
- Aller, R.C., 2004. Conceptual models of early diagenetic processes: the muddy seafloor as an unsteady, batch reactor. Journal of Marine Research 62, 815–835.
- Aller, R.C., Aller, J.Y., 1992. Meiofauna and solute transport in marine muds. Limnology and Oceanography 37, 1018–1033.
- Aller, R.C., Aller, J.Y., 1998. The effect of biogenic irrigation intensity and solute exchange on diagenetic reaction rates in marine sediments. Journal of Marine Research 56, 905–936.
- Aller, R.C., Aller, J.Y., Kemp, P.F., 2001. Effects of particle and solute transport on rates and extent of remineralization in bioturbated sediments. In: Aller, J.Y., Woodin, S.A.,

Aller, R.C. (Eds.), Organism–Sediment Interaction. University of South Carolina Press, Columbia, pp. 315–333.

- Aller, R.C., Benninger, L.K., 1981. Spatial and temporal patterns of dissolved ammonium, manganese, and silica fluxes from the bottom sediments of Long Island Sound, U.S.A. Journal of Marine Research 39, 295–314.
- Aller, R.C., Blair, N.C., Xia, Q., Rude, P.D., 1996. Remineralization rates, recycling and storage of carbon in Amazon shelf sediments. Continental Shelf Research 16, 753–786.
- Aller, R.C., Blair, N.E., 1996. Sulfur diagenesis and burial on the Amazon shelf: major control by physical sedimentation processes. Geo-Marine Letters 16, 3–10.
- Aller, R.C., Blair, N.E., Brunskill, G.J., 2008. Early diagenetic cycling, incineration, and burial of sedimentary organic carbon in the central Gulf of Papua (Papua New Guinea). Journal of Geophysical Research 113, doi:10.1029/2006JF000689.
- Aller, R.C., Hall, P.O.J., Rude, P.D., Aller, J.Y., 1998. Biogeochemical heterogeneity and suboxic diagenesis in hemipelagic sediments of the Panama Basin. Deep-Sea Research 45, 133–165.
- Aller, R.C., Hannides, A., Heilbrun, C., Panzeca, C., 2004a. Coupling of early diagenetic processes and sedimentary dynamics in tropical shelf environments: the Gulf of Papua deltaic complex. Continental Shelf Research 24, 2455–2486.
- Aller, R.C., Heilbrun, C., Panzeca, C., Zhu, Z., Baltzer, F. 2004b. Coupling between sedimentary dynamics, early diagenetic processes and biogeochemical cycling in the Amazon–Guianas mobile mud belt: coastal French Guiana. Marine Geology 208, 331–360.
- Aller, R.C., Mackin, J.E., 1984. Preservation of reactive organic matter in marine sediments. Earth and Planetary Science Letters 70, 260–266.
- Aller, R.C., Mackin, J.E., Knox, R.T., Jr., 1986. Diagenesis of Fe and S in Amazon inner shelf muds: apparent dominance of Fe reduction and implications for the genesis of ironstones. Continental Shelf Research 6, 263–289.
- Aller, R.C., Rude, P.D., 1988. Complete oxidation of solid phase sulfides by manganese and bacteria in anoxic marine sediments. Geochimica et Cosmochimica Acta 52, 751–765.
- Aller, R.C., Yingst, J.Y., 1978. Biogeochemistry of tube-dwellings study of sedentary polychaete *Amphitrite ornata* (Leidy). Journal of Marine Research 36, 201–254.
- Alongi, D.M., 1991. The role of intertidal mudbanks in the diagenesis and export of dissolved and particulate materials from the Fly Delta, Papua New Guinea. Journal of Experimental Marine Biology and Ecology 149, 81–107.
- Alongi, D.M., 1995. Decomposition and recycling of organic matter in muds of the Gulf of Papua, northern Coral Sea. Continental Shelf Research 15, 1319–1337.
- Alongi, D.M., 1998. Coastal Ecosystem Processes. CRC Press, Boca Raton, FL
- Alperin, M.J., Hoehler, T.M., 2009. Anaerobic methane oxidation by archaea/sulfatereducing bacteria aggregates: 1. Thermodynamic and physical constraints. American Journal of Science 309, 869–957.
- Alperin, M.J., Reeburgh, W.S., 1985. Inhibition experiments on anaerobic methane oxidation. Applied and Environmental Microbiology 50, 940–945.
- Anschutz, P., Sundby, B., Lefrançois, L., Luther, G.W., III, Mucci, A., 2000. Interactions between metal oxides and species of nitrogen and iodine in bioturbated marine sediments. Geochimica et Cosmochimica Acta 64, 2751–2763.
- Arzayus, K.A., Canuel, E.A., 2005. Organic matter degradation in sediments of the York River estuary: effects of biological vs. physical mixing. Geochimica et Cosmochimica Acta 69, 455–463.
- Audry, S., Blanc, G., Schäfer, J., Chaillou, G., Robert, S., 2006. Early diagenesis of trace metals (Cd, Cu, Co, Ni, U, Mo, and V) in the freshwater reaches of a macrotidal estuary. Geochimica et Cosmochimica Acta 70, 2264–2282.
- Audry, S., Blanc, G., Schäfer, J., Guérin, F., Masson, M., Robert, S., 2007a. Budgets of Mn, Cd and Cu in the macrotidal Gironde estuary (SW France). Marine Chemistry 107, 433–448.
- Audry, S., Blanc, G., Schäfer, J., Robert, S., 2007b. Effect of estuarine sediment resuspension on early diagenesis, sulfide oxidation and dissolved molybdenum and uranium distribution in the Gironde estuary, France. Chemical Geology 238, 149–167.
- Balistrieri, L.S., Murray, J.W., 1986. The surface chemistry of sediments from the Panama Basin: the influence of Mn oxides on metal adsorption. Geochimica et Cosmochimica Acta 50, 235–2243.
- Balzer, W., 1982. On the distribution of iron and manganese at the sediment/water interface: thermodynamic versus kinetic control. Geochimica et Cosmochimica Acta 46, 1153–1161.
- Bartlett, R., Mortimer, R.J.G., Morris, K., 2008. Anoxic nitrification: evidence from Humber Estuary sediments (UK). Chemical Geology 250, 29–39.
- Bender, M., Jahnke, R., Weiss, R., Martin, W., Heggie, D.T., Orchardo, J., Sowers, T., 1989. Organic carbon oxidation and benthic nitrogen and silica dynamics in San Clemente Basin, a continental borderland site. Geochimica et Cosmochimica Acta 53, 685–697.
- Bennett, R.H., Li, H., Lambert, D.N., Fisher, K.M., Walter, D.J., Hickox, C.E., Hulbert, M.H., Yamamoto, T., Badiey, M., 1990. *In situ* porosity and permeability of selected

carbonate sediment: Great Bahama Bank Part 1: measurements. Marine Geotechnology 9, 1–28.

- Benning, L.G., Wilkin, R.T., Barnes, H.L., 2000. Reaction pathways in the Fe–S system below 100 °C. Chemical Geology 167, 25–51.
- Berelson, W.M., McManus, J., Kilgore, T., Coale, K., Johnson, K.S., Burdige, D., Pilskaln, C., 1996. Biogenic matter diagenesis on the sea floor: a comparison between two continental margin transects. Journal of Marine Research 54, 731–762.
- Berg, P., Huettel, M., 2009. Monitoring the seafloor using the noninvasive eddy correlation technique: integrated benthic exchange dynamics. Oceanography 21, 164–167.
- Berg, P., Rysgaard, S., Funch, P., Sejr, M.K., 2001. Effects of bioturbation on solutes and solids in marine sediments. Aquatic Microbial Ecology 26, 81–94.
- Berg, P., Rysgaard, S., Thamdrup, B., 2003. Dynamic modeling of early diagenesis and nutrient cycling. A case study in an Arctic marine sediment. American Journal of Science 303, 905–955.
- Berner, E.K., Berner, R.A., 1996. Global Environment: Water, Air, and Geochemical Cycles. Prentice Hall, Upper Saddle River, NJ.
- Berner, R.A., 1970. Sedimentary pyrite formation. American Journal of Science 268, 1-23.
- Berner, R.A., 1989. Biogeochemical cycles of carbon and sulfur and their effect on atmospheric oxygen over Phanerozoic time. Paleogeography Paleoclimatology Paleoecology 75, 97–122.
- Berner, R.A., 1999. Atmospheric oxygen over Phanerozoic time. Proceedings of the National Academy of Sciences 96, 10955–10957.
- Berner, R.A., 2004. The Phanerozoic Carbon Cycle. Oxford University Press, Oxford.
- Berner, R.A., Westrich, J.T., 1985. Bioturbation and the early diagenesis of carbon and sulfur. American Journal of Science 285, 193–206.
- Berninger, U.G., Huettel, M., 1997. Impact of flow on oxygen dynamics in photosynthetically active sediments. Aquatic Microbial Ecology 12, 291–300.
- Bianchi, T.S., Allison, M.A., 2009. Large-river delta-front estuaries as natural "recorders" of global environmental change. Proceedings of the National Academy of Sciences 106, 8085–8092.
- Blair, N.E., Aller, R.C., 1995. Anaerobic methane oxidation on the Amazon shelf. Geochimica et Cosmochimica Acta 59, 3707–3715.
- Boetius, A., Ravenschlag, K., Schubert, C.J., Rickert, D., Widdel, F., Gieseke, A., Amann, R., Jorgensen, B.B., Witte, U., Pfannkuche, O., 2000. A marine microbial consortium apparently mediating anaerobic oxidation of methane. Nature 407, 623–626.
- Borum, J., Sand-Jensen, K., Binzer, T., Pedersen, O., Greve, T., 2006. Oxygen movement in seagrasses. In: Larkum, A., Orth, R., Duarte, C. (Eds.), Seagrasses: Biology, Ecology and Conservation. Springer, Dordrecht, pp. 255–270.
- Boudreau, B.P., 1984. On the equivalence of nonlocal and radial-diffusion models for porewater irrigation. Journal of Marine Research 42, 731–735.
- Boudreau, B.P., 1986a. Mathematics of tracer mixing in sediments: I. Spatiallydependent, diffusive mixing. American Journal of Science 286, 199–238.
- Boudreau, B.P., 1986b. Mathematics of tracer mixing in sediments: II. Nonlocal mixing and biological conveyor-belt phenomena. American Journal of Science 286, 161–198.
- Boudreau, B.P., 1998. Mean mixed depth of sediments: the wherefore and the why. Limnology and Oceanography 43, 524–526.
- Boudreau, B.P., 2000. The mathematics of early diagenesis: from worms to waves. Reviews of Geophysics 38, 389–416.
- Boudreau, B.P., Huettel, M., Forster, S., Jahnke, R.A., McLachlan, A., Middelburg, J.J., Nielsen, P., Sansone, F., Taghon, G., Van Raaphorst, W., Webster, I., Marcin, J., Wiberg, P., Sundby, B., 2001. Permeable marine sediments, overturning an old paradigm. EOS, Transactions, American Geophysical Union 82, 133.
- Boudreau, B.P., Marinelli, R.L., 1994. A modelling study of discontinuous biological irrigation. Journal of Marine Research 52, 947–968.
- Boudreau, B.P., Westrich, J.T., 1984. The dependence of bacterial sulfate reduction on sulfate concentrations in marine sediments. Geochimica et Cosmochimica Acta 48, 2503–2516.
- Bouillon, S., Borges, A.V., Castañeda-Moya, E., Diele, K., Dittmar, T., Duke, N.C., Kristensen, E., Lee, S.Y., Marchand, C., Middelburg, J.J., Rivera-Monroy, V.H., Smith, T.J., III, Twilley, R.R., 2008. Mangrove production and carbon sinks: a revision of global budget estimates. Global Biogeochemal Cycles 22, doi:10.1029/2007GB003052.
- Bouillon, S., Boschker, H.T.S., 2006. Bacterial carbon sources in coastal sediments: a cross-system analysis based on stable isotope data of biomarkers. Biogeosciences 3, 175–185.
- Bouillon, S., Middelburg, J.J., Dehairs, F., Borges, A.V., Abril, G., Flindt, M.R., Ulomi, S., Kristensen, E., 2007. Importance of intertidal sediment processes and porewater exchange on the water column biogeochemistry in a pristine mangrove creek (Ras Dege, Tanzania). Biogeosciences 4, 311–322.
- Boynton, W.R., Kemp, W.M., 2008. Estuaries. In: Capone, D.G., Bronk, D.A., Mulholland, M.R., Carpenter, E.J. (Eds.), Nitrogen in the Marine Environment, Second ed. Academic Press, San Diego, CA, pp. 809–866.

Brandes, J.A., Devol, A.H., 1995. Simultaneous nitrate and oxygen respiration in coastal sediments: evidence for discrete diagenesis. Journal of Marine Research 53, 771–797.

- Brandes, J.A., Devol, A.H., Deutsch, C., 2007. New developments in the marine nitrogen cycle. Chemical Reviews 107, 577–589.
- Burdige, D.J., 1993. The biogeochemistry of manganese and iron reduction in marine sediments. Earth-Science Reviews 35, 249–284.
- Burdige, D.J., 2001. Dissolved organic matter in Chesapeake Bay sediment pore waters. In: Canuel, E., Bianchi, T. (Eds.), Organic Geochemical Tracers in Estuaries. Special Issue: Organic Geochemistry. Elsevier, vol. 32, pp. 487–505.
- Burdige, D.J., 2002. Sediment pore waters. In: Hansell, D.A., Carlson, C.D. (Eds.), Biogeochemistry of Marine Dissolved Organic Matter. Academic Press, San Diego, CA, pp. 611–663.
- Burdige, D.J., 2005. The burial of terrestrial organic matter in marine sediments: a reassessment. Global Biogeochemical Cycles 19, doi:10.1029/2004GB002368.
- Burdige, D.J., 2006. Geochemistry of Marine Sediments. Princeton University Press, Princeton.
- Burdige, D.J., 2007. The preservation of organic matter in marine sediments: controls, mechanisms and an imbalance in sediment organic carbon budgets? Chemical Reviews 107, 467–485.
- Burdige, D.J., Dhakar, S.P., Nealson, K.H., 1992. Effects of manganese oxide mineralogy on microbial and chemical manganese reduction. Geomicrobiology Journal 10, 27–48.
- Burdige, D.J., Gieskes, J.M., 1983. A porewater/solid phase diagenetic model for manganese in marine sediments. American Journal of Science 283, 29–47.
- Burdige, D.J., Hu, X., Zimmerman, R.C., 2008. Rates of carbonate dissolution in permeable sediments estimated from pore water profiles: the role of seagrasses. Limnology and Oceanography 53, 549–565.
- Burdige, D.J., Hu, X., Zimmerman, R.C., 2010. The widespread occurrence of coupled calcium carbonate dissolution and reprecipitation in shallow water carbonate platform sediments. American Journal of Science 310, 492–521.
- Burdige, D.J., Nealson, K.H., 1986. Chemical and microbiological studies of sulfidemediated manganese reduction. Geomicrobiology Journal 4, 361–387.
- Burdige, D.J., Zheng, S., 1998. The biogeochemical cycling of dissolved organic nitrogen in estuarine sediments. Limnology and Oceanography 43, 1796–1813.
- Burdige, D.J., Zimmerman, R.C., 2002. Impact of seagrass density on carbonate dissolution in Bahamian sediments. Limnology and Oceanography 47, 1751–1763.
- Burnett, W.C., Aggarwal, P.K., Aureli, A., Bokuniewicz, H., Cable, J.E., Charette, M.A., Kontar, E., Krupa, S., Kulkarni, K.M., Loveless, A., Moore, W.S., Oberdorfer, J.A., Oliveira, J., Ozyurt, N., Povinec, P., Privitera, A.M.G., Rajar, R., Ramessur, R.T., Scholten, J., Stieglitz, T., Taniguchi, M., Turner, J.V., 2006. Quantifying submarine groundwater discharge in the coastal zone via multiple methods. Science of the Total Environment 367, 498–543.
- Burns, R.G., Burns, V.M., 1981. Authigenic oxides. In: Emiliani, C. (Ed.), The Sea. Wiley-Interscience, New York, NY, pp. 875–914.
- Caetano, M., Madureira, M.J., Vale, C., 2007. Exchange of Cu and Cd across the sediment–water interface in intertidal mud flats from Ria Formosa (Portugal). Hydrobiologia 587, 147–155.

Calvert, S.E., Pedersen, T.F., 1992. Organic carbon accumulation and preservation in marine sediments: how important is anoxia? In: Whelan, J.K., Farrington, J.W. (Eds.), Productivity, Accumulation, and Preservation of Organic Matter in Recent and Ancient Sediments. Columbia University Press, New York, NY, pp. 231–263.

- Calvert, S.E., Pedersen, T.F., 1993. Geochemistry of recent oxic and anoxic marine sediments: implications for the geological record. Marine Geology 113, 67–88.
- Canfield, D., Thamdrup, B., 1994. The production of ³⁴S-depleted sulfide during bacterial disproportionation of elemental sulfur. Science 266, 1973–1975.
- Canfield, D., Thamdrup, B., Kristensen, E., 2005. Aquatic Geomicrobiology. Advances in Marine Biology. Elsevier, San Diego, CA, vol. 48.
- Canfield, D.E., 1989. Reactive iron in marine sediments. Geochimica et Cosmochimica Acta 53, 619–632.
- Canfield, D.E., 1993. Organic matter oxidation in marine sediments. In: Wollast, R., Chou, L., Mackenzie, F. (Eds.), Interactions of C, N, P and S Biogeochemical Cycles. NATO-ARW. Springer, Berlin, pp. 333–363.
- Canfield, D.E., 1994. Factors influencing organic matter preservation in marine sediments. Chemical Geology 114, 315–329.
- Canfield, D.E., Jorgensen, B.B., Fossing, H., Glud, R., Gundersen, J., Ramsing, N.B., Thamdrup, B., Hansen, J.W., Nielsen, L.P., Hall, P.O.J., 1993a. Pathways of organic carbon oxidation in three continental margin sediments. Marine Geology 113, 27–40.
- Canfield, D.E., Raiswell, R., Westrich, J.T., Reaves, C.M., Berner, R.A., 1986. The use of chromium reduction in the analysis of reduced inorganic sulfur in sediments and shales. Chemical Geology 54, 149–155.
- Canfield, D.E., Raiswell, R., 1991a. Carbonate dissolution and precipitation: its relevance to fossil preservation. In: Allison, P.A., Briggs, D.E.G. (Eds.), Taphony: Releasing the Data Locked in the Fossil Record. Plenum, New York, NY, pp. 411–453.

Canfield, D.E., Raiswell, R., 1991b. Pyrite formation and fossil preservation. In: Allison, P.A., Briggs, D.E.G. (Eds.), Taphonomy: Releasing the Data Locked in the Fossil Record. Plenum, New York, NY, pp. 337–387.

Canfield, D.E., Raiswell, R., Bottrell, S., 1992. The reactivity of sedimentary iron minerals toward sulfide. American Journal of Science 292, 659–683.

Canfield, D.E., Thamdrup, B., 2009. Towards a consistent classification scheme for geochemical environments, or, why we wish the term 'suboxic' would go away. Geobiology 7, 385–392.

Canfield, D.E., Thamdrup, B., Hansen, J.W., 1993b. The anaerobic degradation of organic matter in Danish coastal sediments: Fe reduction, Mn reduction, and sulfate reduction. Geochimica et Cosmochimica Acta 57, 2563–2570.

Canuel, E.A., Martens, C.S., 1996. Reactivity of recently deposited organic matter: degradation of lipid compounds near the sediment–water interface. Geochimica et Cosmochimica Acta 60, 1793–1806.

Capone, D.G., Kiene, R.P., 1988. Comparison of microbial dynamics in marine and freshwater sediments: contrasts in anaerobic carbon catabolisms. Limnology and Oceanography 33, 725–749.

Cardenas, M.B., Cook, P.L.M., Jiang, H., Traykovski, P., 2008. Constraining denitrification in permeable wave-influenced marine sediment using linked hydrodynamic and biogeochemical modeling. Earth and Planetary Science Letters 275, 127–137.

Chanton, J.P., Dacey, J.W.H., 1991. Effects of vegetation on methane flux, reservoirs, and carbon isotopic composition. In: Sharkey, T.D., Holland, E.A., Mooney, H.A. (Eds.), Trace Gas Emissions by Plants. Academic Press, San Diego, CA, pp. 65–92.

Chanton, J.P., Martens, C.S., Goldhaber, M.B., 1987. Biogeochemical cycling in an organic-rich marine basin – 7. Sulfur mass balance, oxygen uptake and sulfide retention. Geochimica et Cosmochimica Acta 51, 1187–1199.

Chanton, J.P., Martens, C.S., Kelley, C.A., 1989. Gas transport from methane-saturated, tidal freshwater and wetland sediments. Limnology and Oceanography 34, 807–819.

Chanton, J.P., Whiting, G.J., 1995. Trace gas exchange in freshwater and coastal marine environments: ebullition and transport by plants. In: Matson, P.A., Harriss, R.C. (Eds.), Biogenic Trace Gases: Measuring Emissions from Soil and Water. Blackwell, Oxford. pp. 98–125.

Charette, M.A., Sholkovitz, E.R., 2006. Trace element cycling in a subterranean estuary: part 2. Geochemistry of the pore water. Geochimica et Cosmochimica Acta 70, 811–826.

Chen, Z., Mayer, L.M., 1998. Mechanisms of Cu solubilization during deposit feeding. Environmental Science and Technology 32, 770–775.

Chen, Z., Mayer, L.M., 1999. Sedimentary metal bioavailability determined by the digestive constraints of marine deposit feeders: gut retention time and dissolved amino acids. Marine Ecology Progress Series 176, 139–151.

Chester, R., 2000. Marine Geochemistry. Blackwell, Oxford.

Chmura, G.L., Anisfeld, S.C., Cahoon, D.R., Lynch, J.C., 2003. Global carbon sequestration in tidal, saline wetland soils. Global Biogeochemal Cycles 17, doi:10.1029/2002GB001917.

Claypool, G.E., Kaplan, I.R., 1974. The origin and distribution of methane in marine sediments. In: Kaplan, I.R. (Ed.), Natural Gases in Marine Sediments. Plenum, New York, NY, pp. 99–139.

Clement, B.G., Luther, G.W., III, Tebo, B.M., 2009. Rapid, oxygen-dependent microbial Mn(II) oxidation kinetics at sub-micromolar oxygen concentrations in the Black Sea suboxic zone. Geochimica et Cosmochimica Acta 73, 1878–1889.

Colbert, D., Coale, K.H., Berelson, W.M., Johnson, K.S., 2001. Cadmium flux in Los Angeles/Long Beach harbours and at sites along the California continental margin. Estuarine, Coastal and Shelf Science 53, 169–180.

Cooper, D.C., Morse, J.W., 1998a. Biogeochemical controls on trace metal cycling in anoxic marine sediments. Environmental Science and Technology 32, 327–330.

Cooper, D.C., Morse, J.W., 1998b. Extractability of metal sulfide minerals in acidic solutions: application to environmental studies of trace metal contamination. Environmental Science and Technology 32, 1076–1078.

Corbett, D.R., McKee, B., Duncan, D., 2004. An evaluation of mobile mud dynamics in the Mississippi River deltaic region. Marine Geology 209, 91–112.

Cornell, R.M., Schwertmann, U., 1996. The Iron Oxides. VCH, Weinheim.

Cornwell, J.C., Morse, J.W., 1987. The characterization of iron sulfide minerals in anoxic marine sediments. Marine Chemistry 22, 193–206.

Cornwell, J.C., Sampou, P.A., 1995. Environmental controls on iron sulfide mineral formation in a coastal plain estuary. In: Vairavamurthy, M.A., Schoonen, M.A.A. (Eds.), Geochemical Transformations of Sedimentary Sulfur. American Chemical Society, Washington, DC, pp. 224–242.

Cowan, J.L.W., Pennock, J.R., Boynton, W.R., 1996. Seasonal and interannual patterns of sediment–water nutrient and oxygen fluxes in Mobile Bay, Alabama (USA): regulating factors and ecological significance. Marine Ecology Progress Series 141, 229–245.

Crill, P.M., Martens, C.S., 1986. Methane production from bicarbonate and acetate in an anoxic marine sediment. Geochimica et Cosmochimica Acta 50, 2089–2097.

- Crusius, J., Thomson, J., 2003. Mobility of authigenic rhenium, silver, and selenium during postdepositional oxidation in marine sediments. Geochimica et Cosmochimica Acta 67, 265–273.
- D'Andrea, A.F., Aller, R.C., Lopez, G.R., 2002. Organic matter flux and reactivity on a South Caroina sandflat: the impacts of pore water advection and macrobiological structures. Limnology and Oceanography 47, 1056–1070.

Dalsgaard, T., Thamdrup, B., Canfield, D.E., 2005. Anaerobic ammonium oxidation (anammox) in the marine environment. Research in Microbiology 156, 457–464.

Decho, A.W., 2000. Microbial biofilms in intertidal systems: an overview. Continental Shelf Research 20, 1257–1273.

Deflandre, B., Mucci, A., Gagne, J., Guignard, C., Sundby, B., 2002. Early diagenetic processes in coastal marine sediments disturbed by a catastrophic sedimentation event. Geochimica et Cosmochimica Acta 66, 2547–2558.

Dellapenna, T.M., Kuehl, S.A., Schaffner, L.C., 1998. Sea-bed mixing and particle residence time in biologically and physically dominated estuarine systems: a comparison of lower Cheapeake Bay and the York River subestuary. Estuarine, Coastal and Shelf Science 46, 777–795.

DeMaster, D.J., Aller, R.C., 2001. Biogeochemical processes on the Amazon shelf: changes in dissolved and particulate fluxes during river/ocean mixing. In: McClain, M.E., Victoria, R.L., Richey, J.E. (Eds.), The Biogeochemistry of the Amazon Basin. Oxford University Press. New York. NY, pp. 328–357.

Devol, A.H., 1991. Direct measurement of nitrogen gas fluxes from continental margin sediments. Nature 349, 319–321.

Devol, A.H., 2008. Denitrification including anammox. In: Capone, D.G., Bronk, D.A., Mulholland, M.R., Carpenter, E.J. (Eds.), Nitrogen in the Marine Environment, Second ed. Academic Press, San Diego, CA, pp. 263–301.

Devol, A.H., Anderson, J.J., Kuivala, K., Murray, J.W., 1984. A model for coupled sulfate reduction and methane oxidation in the sediments of Saanich Inlet. Geochimica et Cosmochimica Acta 48, 933–1004.

Devol, A.H., Christensen, J.P., 1993. Benthic fluxes and nitrogen cycling in sediments of the continental margin of the eastern North Pacific. Journal of Marine Research 51, 345–372.

Diaz, R.J., Rosenberg, R., 1995. Marine benthic hypoxia: a review of its ecological effects and the behavioural responses of benthic macrofauna. Oceanography and Marine Biology: An Annual Review 33, 245–303.

Dierssen, H.M., Zimmerman, R.C., Drake, L.A., Burdige, D.J., 2009. Potential transport of unattached benthic macroalgae to the deep sea through wind-driven Langmuir circulation. Geophysical Research Letters 36, L04602. doi:10.1029/2008GL036188.

DiToro, D.M., 2001. Sediment Flux Modeling. Wiley-Interscience, New York, NY.

Dong, H., Kukkadapu, R.K., Fredrickson, J.K., Zachara, J.M., Kennedy, D.W., Kostandarithes, H.M., 2003. Microbial reduction of structural Fe(III) in illite and geothite. Environmental Science and Technology, 37, 1268–1276.

Dorgan, K.M., Jumars, P.A., Johnson, B.D., Boudreau, B.P., 2006. Macrofaunal burrowing: the medium is the message. Oceanography and Marine Biology: An Annual Review 44, 85–121.

Duarte, C.M., Holmer, M., Marba, N., 2005a. Plant-microbe interactions in seagrass meadows. In: Kristensen, E., Haese, R.R., Kostka, J.E. (Eds.), Interactions Between Macro- and Microorganisms in Marine Sediments. American Geophysical Union, Washington, DC, pp. 31–60.

Duarte, C. M., Marbà, N., Gacia, E., Fourqurean, J.W., Beggins, J., Barrón, C., Apostolaki, E.T., 2010. Seagrass community metabolism: Assessing the carbon sink capacity of seagrass meadows, Global Biogeochemistry Cycles 24, GB4032. DOI: 10.1029/2010GB003793.

Duarte, C.M., Middelburg, J.J., Caraco, N., 2005b. Major role of marine vegetation on the oceanic carbon cycle. Biogeosciences 2, 1–8.

Elderfield, H., 1981. Metal-organic associations in interstitial waters of Narragansett Bay sediments. American Journal of Science 281, 1184–1196.

Elderfield, H., Luedtke, N., McCafrey, R.J., Bender, M., 1981a. Benthic flux studies in Narragansett Bay. American Journal of Science 281, 768–787.

- Elderfield, H., McCafrey, R.J., Luedtke, N., Bender, M., 1981b. Chemical diagenesis in Narragansett Bay sediments. American Journal of Science 281, 1021–1055.
- Emerson, S., Fischer, K., Reimers, C., Heggie, D., 1985. Organic carbon dynamics and preservation in deep-sea sediments. Deep-Sea Research 32: 1–21.
- Emerson, S., Jahnke, R., Heggie, D., 1984. Sediment–water exchange in shallow water estuarine sediments. Journal of Marine Research 42, 709–730.

Emery, K.O., 1968. Relict sediments on continental shelves of the world. AAPG Digital Bulletin 52, 445–464.

Engström, P., Dalsgaard, T., Hulth, S., Aller, R.C., 2005. Anaerobic ammonium oxidation by nitrite (anammox): implications for N₂ production in coastal marine sediments. Geochimica et Cosmochimica Acta 69, 2057–2065.

Falter, J.L., Sansone, F.J., 2000. Hydraulic control of pore water geochemistry within the oxic– suboxic zone of a permaeable sediment. Limnology and Oceanography 45, 550–557.

Fanning, K.A., Carder, K.L., Betzer, P.R., 1982. Sediment resuspension by coastal waters: a potential mechanism for nutrient re-cycling on the ocean's margins. Deep-Sea Research Part A. Oceanographic Research Papers 29, 953–965. Fauque, G.D., 1995. Ecology of sulfate-reducing bacteria. In: Barton, L.L. (Ed.), Sulfate-Reducing Bacteria. Plenum, New York, NY, pp. 217–241.

- Fenchel, T., King, G.M., Blackburn, T.H., 1998. Bacterial Biogeochemistry: The Ecophysiology of Mineral Cycling. Academic Press, San Diego, CA.
- Forster, S., Bobertz, B., Bohling, B., 2003. Permeability of sands in the coastal areas of the southern Baltic Sea: mapping a grain-size related sediment property. Aquatic Geochemistry 9, 171–190.
- Forster, S., Huettel, M., Ziebis, W., 1996. Impact of boundary layer flow velocity on oxygen utilisation in coastal sediments. Marine Ecology Progress Series 143, 173–185.
- Fredericksen, M.S., Glud, R.N., 2006. Oxygen dynamics in the rhizosphere of *Zostera marina*: a two-dimensional planar optode study. Limnology and Oceanography 51, 1072–1083.
- Froelich, P.N., Klinkhammer, G.P., Bender, M.L., Luedtke, N.A., Heath, G.R., Cullen, D., Dauphin, P., Hammond, D., Hartman, B., Maynard, V., 1979. Early oxidation of organic matter in pelagic sediments of the eastern equatorial Atlantic: suboxic diagenesis. Geochimica et Cosmochimica Acta 43, 1075–1090.
- Galler, J.J., Allison, M.A., 2008. Estuarine controls on fine-grained sediment storage in the Lower Mississippi and Atchafalaya Rivers. GSA Bulletin 120, 369–398.
- Gao, H., Schreiber, F., Collins, G., Jensen, M.M., Kostka, J.E., Lavik, G., de Beer, D., Zhou, H.-Y., Kuypers, M.M.M., 2009. Aerobic denitrification in permeable Wadden Sea sediments. ISME Journal 4, 417–426.

Gargett, A., Wells, J., Tejada-Martinez, A.E., Grosch, C.E., 2004. Langmuir supercells: a dominant mechanism for sediment resuspension and transport. Science 306, 1925–1928.

- Gattuso, J.P., Gentili, B., Duarte, C.M., Kleypas, J.A., Middelburg, J.J., Antoine, D., 2006. Light availability in the coastal ocean: impact on the distribution of benthic photosynthetic organisms and their contribution to primary production. Biogeosciences 3, 489–513.
- Gerino, M., Aller, R.C., Lee, C., Cockran, J.K., Aller, J.Y., Green, M.A., Hirschberg, D., 1998. Comparison of different tracers and methods used to quantify bioturbation during a spring bloom: 234-thorium, luminophores and chlorophyll *a*. Estuarine, Coastal and Shelf Science 46, 531–547.
- Ghisalberti, M., Nepf, H.M., 2002. Mixing layers and coherent structures in vegetated aquatic flows. Journal of Geophysical Research 107, doi:10.1029/2001JC000871.
- Gihring, T.M., Canion, A., Riggs, A., Huettel, M., Kostka, J.E., 2010. Denitrification in shallow, sublittoral Gulf of Mexico permeable sediments. Limnology and Oceanography 55, 43–54.
- Glasby, G., 2000. Manganese: predominant role of nodules and crusts. In: Schulz, H.D., Zabel, M. (Eds.), Marine Geochemistry. Springer, New York, NY, pp. 335–372.
- Glud, R.N., 2008. Oxygen dynamics of marine sediments. Marine Biology Research 4, 243–289.
- Glud, R.N., Gundersen, J.K., Røy, H., Jørgensen, B.B., 2003. Seasonal dynamics of benthic O₂ uptake in a semienclosed bay: importance of diffusion and faunal activity. Limnology and Oceanography 48, 1265–1276.
- Glud, R.N., Tengberg, A., Kuhl, M., Hall, P.O.J., Klimant, I., Holst, G., 2001. An *in situ* instrument for planar O₂ optode measurements at benthic interfaces. Limnology and Oceanography 46, 2073–2080.
- Gobeil, C., Silverberg, N., Sundby, B., Cossa, D., 1987. Cadmium diagenesis in Laurentian Trough sediments. Geochimica et Cosmochimica Acta 51, 589–596.
- Goldhaber, M.B., 2003. Sulfur-rich sediments. In: Mackenzie, F.T. (Ed.), Treatise on Geochemistry. Elsevier, Amsterdam, pp. 257–268.
- Green, M.A., Aller, R.C., 1998. Seasonal patterns of carbonate diagenesis in nearshore terrigenous muds: relation to spring phytoplankton bloom and temperature. Journal of Marine Research 56, 1097–1123.
- Green, M.A., Jones, M.E., Boudreau, C.L., Moore, R.L., Westman, B.A., 2004. Dissolution mortality of juvenile bivalves in coastal marine deposits. Limnology and Oceanography 49, 727–734.
- Gribsholt, B., Kostka, J.E., Kristensen, E., 2003. Impact of fiddler crabs and plant roots on sediment biogeochemistry in a Georgia saltmarsh. Marine Ecology Progress Series 259, 237–251.
- Grigg, N., Boudreau, B.P., Webster, I.T., Ford, P.W., 2005. The nonlocal model of porewater irrigation: limits to its equivalence with a cylinder diffusion model. Journal of Marine Research 63, 437–455.
- Habicht, K.S., Canfield, D.E., 1997. Sulfur isotope fractionation during bacterial sulfate reduction in organic-rich sediments. Geochimica et Cosmochimica Acta 61, 5351–5361.
- Habicht, K.S., Canfield, D.E., 2001. Isotope fractionation by sulfate-reducing natural populations and the isotopic composition of sulfide in marine sediments. Geology 29, 555–558.
- Habicht, K.S., Canfield, D.E., Rethmeier, J., 1998. Sulfur isotope fractionation during bacterial reduction and disproportionation of thiosulfate and sulfite. Geochimica et Cosmochimica Acta 62, 2585–2595.
- Hanson, R.S., Hanson, T.E., 1996. Methanotrophic bacteria. Microbiological Reviews 60, 439–471.

- Hebert, A., Morse, J., Eldridge, P., 2007. Small-scale heterogeneity in the geochemistry of seagrass vegetated and non-vegetated estuarine sediments: causes and consequences. Aquatic Geochemistry 13, 19–39.
- Hedges, J.I., 2002. Sedimentary organic matter preservation and atmospheric O₂ regulation. In: Gianguzza, A., Pelizzetti, E., Sammartano, S. (Eds.), Chemistry of Marine Water and Sediments. Springer, Berlin, pp. 105–123.
- Hedges, J.I., Keil, R.G., 1995. Sedimentary organic matter preservation: an assessment and speculative synthesis. Marine Chemistry 49, 81–115.
- Hemminga, M., Duarte, C., 2000. Seagrass Ecology. Cambridge University Press, Cambridge.
- Henrichs, S.M., Reeburgh, W.S., 1987. Anaerobic mineralization of marine sediment organic matter: rates and the role of anaerobic processes in the oceanic carbon economy. Geomicrobiology Journal 5, 191–237.
- Herman, P.M.J., Middelburg, J.J., Koppel, J.V.D., Heip, C., 1999. The ecology of estuarine macrobentos. Advances in Ecological Research 29, 195–240.
- Hines, M., 2006. Microbially mediated redox cycling at the oxic–anoxic boundary in sediments: comparison of animal and plants habitats. Water, Air, and Soil Pollution: Focus 6, 523–536.
- Hines, M.E., Banta, G.T., Giblin, A.E., Hobbie, J.E., Tugel, J.B., 1994. Acetate concentrations and oxidation in salt marsh sediments. Limnology and Oceanography 39, 140–148.
- Hines, M.E., Lyons, W.B., Armstrong, P.A., Orem, W.H., Spencer, M.J., Gaudette, H.E., 1984. Seasonal metal remobilization in the sediments of Great Bay, New Hampshire. Marine Chemistry 15, 173–187.
- Hoehler, T.M., Alperin, M.J., Albert, D.B., Martens, C.S., 1994. Field and laboratory studies of methane oxidation in an anoxic marine sediment: evidence for a methanogen-sulfate reducer consortium. Global Biogeochemical Cycles 8, 451–463.
- Hoehler, T.M., Alperin, M.J., Albert, D.B., Martens, C.S., 1998. Thermodynamic control on hydrogen concentrations in anoxic sediments. Geochimica et Cosmochimica Acta 62, 1745–1756.
- Holdren, G.R., Jr., Bricker, O.P., III, Matisoff, G., 1975. A model for the control of dissolved manganese in the interstitial water of Chesapeake Bay. In: Church, T.M. (Ed.), Marine Chemistry in the Coastal Environment. ACS Symposium Series #18. American Chemical Society, Washington, DC, pp. 364–381.
- Holmer, M., Andersen, F.Ø., Holmboe, N., Kristensen, E., Thongtham, N., 1999. Transformation and exchange processes in the Bangrong mangrove forest-seagrass bed system, Thailand. Seasonal and spatial variations in benthic metabolism and sulfur biogeochemistry. Aquatic Microbial Ecology 20, 203–212.
- Holmer, M., Andersen, F.Ø., Nielsen, S.L., Boschker, H.T.S., 2001. The importance of mineralization based on sulfate reduction for nutrient regeneration in tropical seagrass sediments. Aquatic Botany 71, 1–17.
- Hu, X., Burdige, D.J., 2008. Shallow marine carbonate dissolution and early diagenesis – implications from an incubation study. Journal of Marine Research 66, 489–527.
- Huerta-Diaz, M.A., Morse, J.W., 1992. Pyritization of trace metals in anoxic marine sediments. Geochimica et Cosmochimica Acta 56, 2681–2702.
- Huettel, M., Gust, G., 1992. Impact of bioroughness on interfacial solute exchange in permeable sediments. Marine Ecology Progress Series 89, 253–267.
- Huettel, M., Rusch, A., 2000. Transport and degradation of phytoplankton in permeable sediment. Limnology and Oceanography 45, 534–549.
- Huettel, M., Webster, I.T., 2001. Porewater flow in permeable sediments. In: Boudreau, B.P., Jørgensen, B.B. (Eds.), The Benthic Boundary Layer. Oxford University Press, Oxford, pp. 144–179.
- Huettel, M., Ziebis, W., Forster, S., 1996. Flow-induced uptake of particulate matter in permeable sediments. Limnology and Oceanography 41, 309–322.
- Huettel, M., Ziebis, W., Forster, S., Luther, G.W., III, 1998. Advective transport affecting metal and nutrient distributions and interfacial fluxes in permeable sediments. Geochimica et Cosmochimica Acta 62, 613–631.
- Hulth, S., Aller, R.C., Canfield, D.E., Dalsgaard, T., Engström, P., Gilbert, F., Sundbäck, K., Thamdrup, B., 2005. Nitrogen removal in marine environments: recent findings and future research challenges. Marine Chemistry 94, 125–145.
- Hulth, S., Aller, R.C., Gilbert, F., 1999. Coupled anoxic nitrification/manganese reduction in marine sediments. Geochimica et Cosmochimica Acta 63, 49–66.
- Hunt, C.D., 1983. Variability in the benthic Mn flux in coastal marine ecosystems resulting from temperature and primary production. Limnology and Oceanography 28, 913–923.
- Hyun, J.-H., Mok, J.-S., Cho, H.-Y., Kim, S.-H., Lee, K., Kostka, J., 2009. Rapid organic matter mineralization coupled to iron cycling in intertidal mud flats of the Han River estuary, Yellow Sea. Biogeochemistry 92, 231–245.
- Ingvorsen, K., Jørgensen, B.B., 1984. Kinetics of sulfate uptake by freshwater and marine species of Desulfovibrio. Archives of Microbiology 139, 61–66.
- Iversen, N., Jørgensen, B.B., 1985. Anaerobic methane oxidation at the sulfate-methane transition in marine sediments from the Kattegat and Skagerrak (Denmark). Limnology and Oceanography 30, 944–955.

Jahnke, R., Richards, M., Nelson, J., Roberton, C., Rao, A., Jahnke, D., 2005. Organic matter remineralization and porewater exchange rates in permeable South Atlantic Bight continental shelf sediments. Continental Shelf Research 25, 1433–1452.

Jahnke, R.A., 2005. Transport processes and organic matter cycling in coastal sediments. In: Robinson, A.R., Brink, K.H. (Eds.), The Sea, Volume 13: The Global Coastal Ocean Multiscale Interdisciplinary Processes. Harvard University Press, Cambridge, MA, pp. 163–191.

Jahnke, R.A., 2009. Maybe it's not just about air-water gas exchange. Oceanography 21, 42-43.

Jahnke, R.A., Alexander, C.R., Kostka, J.E., 2003. Advective pore water input of nutrients to the Satilla River estuary, Georgia, USA. Estuarine, Coastal and Shelf Science 56, 641–653.

Jahnke, R.A., Nelson, J.R., Marinelli, R.L., Eckman, J.E., 2000. Benthic fluxes of biogenic elements on the southeastern US continental shelf: influence of pore water advective transport and benthic microalgae. Continental Shelf Research 20, 109–127.

Jenkins, M.C., Kemp, W.M., 1984. The coupling of nitrification and denitrification in two estuarine sediments. Limnology and Oceanography 29, 609–619.

Jensen, H.S., McGlathery, K.J., Marino, R., Howarth, R.W., 1998. Forms and availability of sediment phosphorus in carbonate sands of Bermuda seagrass beds. Limnology and Oceanography 43, 799–810.

Jensen, M.M., Thamdrup, B., Rysgaard, S., Holmer, M., Fossing, H., 2003. Rates and regulation of microbial iron reduction in sediments of the Baltic–North Sea transition. Biogeochemistry 65, 295–317.

Johannesson, K.H., Burdige, D.J., 2007. Balancing the global neodymium budget: evaluating the role of groundwater. Earth and Planetary Science Letters 253, 129–142.

Jones, W.B., Cifuentes, L.A., Kaldy, J.E., 2003. Stable carbon isotope evidence for coupling between sedimentary bacteria and seagrasses in a sub-tropical lagoon. Marine Ecology Progress Series 255, 15–25.

Jørgensen, B.B., 1982a. Ecology of the bacteria of the sulfur cycle with special reference to anoxic–oxic interface environments. Philosophical Transactions of the Royal Society B 25, 799–822.

Jørgensen, B.B., 1982b. Mineralization of organic matter in the sea bed – the role of sulphate reduction. Nature 296, 643–645.

Jørgensen, B.B., Bak, F., 1991. Pathways and microbiology of thiosulfate transformations and sulfate reduction in a marine sediment (Kattegat, Denmark). Applied and Environmental Microbiology 57, 847–856.

Jørgensen, B.B., Parkes, R.J., 2010. Role of sulfate reduction and methane production by organic carbon degradation in eutrophic fjord sediments (Limfjorden, Denmark). Limnology and Oceanography 55 (3), 1338–1352.

Joye, S.B., Anderson, I.C., 2008. Nitrogen cycling in coastal sediments. In: Capone, D.G., Bronk, D.A., Mulholland, M.R., Carpenter, E.J. (Eds.), Nitrogen in the Marine Environment, Second ed. Academic Press, San Diego, CA, pp. 867–915.

Jumars, P.A., Eckman, J.E., Koch, E., 2001. Macroscopic plants and animals in benthic flows. In: Boudreau, B.P., Jørgensen, B.B. (Eds.), The Benthic Boundary Layer. Oxford University Press, Oxford, pp. 320–347.

Kaldy, J.E., Eldridge, P.M., Cifuentes, L.A., Jones, W.B., 2006. Utilization of DOC from seagrass rhizomes by sediment bacteria: ¹³C-tracer experiments and modeling. Marine Ecology Progress Series 317, 41–55.

Kalnejais, L.H., Martin, W.R., Signell, R.P., Bothner, M.H., 2007. Role of sediment resuspension in the remobilization of particulate-phase metals from coastal sediments. Environmental Science and Technology 41, 2282–2288.

Kelley, C.A., Martens, C.S., Chanton, J.P., 1990. Variations in sedimentary carbon remineralization rates in the White Oak River estuary, North Carolina. Limnology and Oceanography 35, 372–383.

Kemp, W.M., Smith, E.M., Marvin-DiPasquale, M.C., Boynton, W.R., 1997. Organic carbon balance and net ecosystem metabolism in Chesapeake Bay. Marine Ecology Progress Series 150, 229–248.

Kipput, G.W., Martens, C.S., 1982. Biogeochemical cycling in an organic-rich marine basin – 3. Dissolved gas transport in methane-saturated sediments. Geochimica et Cosmochimica Acta 46, 2049–2060.

Klump, J.V., Martens, C.S., 1989. The seasonality of nutrient regeneration in an organic rich coastal sediment: kinetic modelling of changing pore water-nutrient and sulfate distributions. Limnology and Oceanography 34, 559–577.

Koch, E., Gust, G., 1999. Water flow in tide- and wave-dominated beds of the seagrass *Thalassia testudinum*. Marine Ecology Progress Series 184, 63–72.

Koch, E.W., Ackerman, J.D., Verduin, J.J., van Keulen, M., 2006. Fluid dynamics in seagrass ecology – from molecules to ecosystems. In: Larkum, A., Orth, R., Duarte, C. (Eds.), Seagrasses: Biology, Ecology and Conservation. Springer, Dordrecht, pp. 193–225.

König, B., Kohls, O., Holst, G., Glud, R.N., Kühl, M., 2005. Fabrication and test of sol-gel based planar oxygen optodes for use in aquatic sediments. Marine Chemistry 97, 262–276.

Koop-Jakobsen, K., Giblin, A., 2009. Anammox in tidal marsh sediments: the role of salinity, nitrogen loading, and marsh vegetation. Estuaries and Coasts 32, 238–245.

Koretsky, C.M., Meile, C., Van Cappellen, P., 2002. Quantifying bioirrigation using ecological parameters: a stochastic approach. Geochemical Transactions 3, 17–30. Kostka, J.E., Dalton, D.D., Skelton, H., Dollhopt, S., Stucki, J.W., 2002a. Growth of iron (III)-reducing bacteria on clay minerals as the sole electron acceptor and comparison of growth yields on a variety of oxidized iron forms. Applied and Environmental Microbiology 68, 6256–6262.

Kostka, J.E., Gribsholt, B., Petrie, E., Dalton, D., Skelton, H., Kristensen, E., 2002b. The rates and pathways of carbon oxidation in bioturbated saltmarsh sediments. Limnology and Oceanography 47, 230–240.

Kostka, J.E., Roychoudhury, A., Van Cappellen, P., 2002c. Rates and controls of anaerobic microbial respiration across spatial and temporal gradients in saltmarsh sediments. Biogeochemistry 60, 49–76.

Kristensen, E., 2001. Impact of polychaetes (*Nereis* spp. and *Arenicola marina*) on carbon biogeochemistry in coastal marine sediments. Geochemical Transactions 2, 92–103.

Kristensen, E., Alongi, D.M., 2006. Control by fiddler crabs (*Uca vocans*) and plant roots (*Avicennia marina*) on carbon, iron, and sulfur biogeochemistry in mangrove sediment. Limnology and Oceanography 51, 1557–1571.

Kristensen, E., Andersen, F.Ø., Holmboe, N., Holmer, M., Thongtham, N., 2000. Carbon and nitrogen mineralization in sediments of the Bangrong mangrove area, Phuket, Thailand. Aquatic Microbial Ecology 22, 199–213.

Kristensen, E., Bouillon, S., Dittmar, T., Marchand, C., 2008. Organic carbon dynamics in mangrove ecosystems: a review. Aquatic Botany 89, 201–219.

Kristensen, E., Haese, R.R., Kostka, J.E. (Eds.), 2005. Interactions between Macro- and Microorganisms in Marine Sediments. Coastal and Estuarine Studies, 60. American Geophysical Union, Washington, DC, pp. 125–158.

Kristensen, E., Mikkelsen, O.L., 2003. Impact of the burrow-dwelling polychaete *Nereis diversicolor* on degradation of fresh and aged macroalgal detritus in coastal marine sediment. Marine Ecology Progress Series 265, 141–153.

Ku, T.C.W., Walter, L.M., Coleman, M.L., Blake, R.E., Martini, A.M., 1999. Coupling between sulfur recycling and syndepositional carbonate dissolution: evidence from oxygen and sulfur isotope compositions of pore water sulfate, south Florida platform, USA. Geochimica et Cosmochimica Acta 63, 2529–2546.

Kuehl, S.A., Nittrouer, C.A., DeMaster, D.J., 1986. Distribution of sedimentary structures in the Amazon subaqueous delta. Continental Shelf Research 6, 311–336.

Laanbroek, H.J., Veldkamp, H., 1982. Microbial interactions in sediment communities. Philosophical Transactions of the Royal Society B 297, 533–550.

Lacerda, L.D., Ribeiro, M.G., Gueiros, B.B., 1999. Manganese dynamics in a mangrove mud flat tidal creek in SE Brazil. Mangroves and Salt Marshes 3, 105–115.

Lasaga, A.C., Holland, H.D., 1976. Mathematical aspects of non-steady-state diagenesis. Geochimica et Cosmochimica Acta 40, 257–266.

Laursen, A.E., Seitzinger, S.P., 2002. The role of denitrification in nitrogen removal and carbon mineralization in Mid-Atlantic Bight sediments. Continental Shelf Research 22, 1397–1416.

Lee, K.-S., Dunton, K.H., 2000. Diurnal changes in pore water sulfide concentrations in the seagrass *Thalassia testudinum* beds: the effects of seagrasses on sulfide dynamics. Journal Experimental Marine Biology and Ecology 255. 201–214.

Lerman, A., 1979. Geochemical Processes, Water and Sediment Environments. Wiley, New York, NY.

Levin, L.A., Ekau, W., Gooday, A.J., Jorissen, F., Middelburg, J.J., Naqvi, W., Neira, C., Rabalais, N.N., Zhang, J., 2009. Effects of natural and human-induced hypoxia on coastal benthos. Biogeosciences 6, 2063–2098.

Li, Y.-H., 2000. A Compendium of Geochemistry: FromSolar Nebula to the Human Brain. Princeton University Press, Princeton.

Lohse, L., Epping, E.H., Helder, W., van Raaphorst, W., 1996. Oxygen pore water profiles in continental shelf sediments of the North Sea: turbulent vs. molecular diffusion. Marine Ecology Progress Series 145 63–75.

Lovell, C.R., 2005. Belowground interactions among salt marsh plants and microorganisms. In: Kristensen, E., Haese, R.R., Kostka, J.E. (Eds.), Interactions between Macro- and Microorganisms in Marine Sediments. American Geophysical Union, Washington, DC, pp. 61–84.

Lovley, D.R., 1991. Dissimilatory Fe(III) and Mn(IV) reduction. Microbiological Reviews 55, 259–287.

Lovley, D.R., 2000. Dissimilatory Fe(III) and Mn(IV)-reducing prokaryotes. In: Dworkin, M., Falkow, S., Rosenberg, E., Schleifer, K., Stackebrandt, E. (Eds.), The Prokaryotes. Springer, New York, NY.

Lovley, D.R., Phillips, E.J.P., 1987. Competitive mechanisms for inhibition of sulfate reduction and methane production in the zone of ferric iron reduction in sediments. Applied and Environmental Microbiology 53, 2636–2641.

Luther, G.W.III, Church, T.M., 1992. An overview of the environmental chemistry of sulphur in wetland systems. In: Howarth, R.W., Stewart, J.W.B., Ivanov, M.V. (Eds.), Sulphur Cycling on the Continents. Wiley, Chichester, pp. 125–142.

Luther, G.W., III, Ferdelmen, T.G., Kostka, J.E., Tsamakis, E.J., Church, T.M., 1991. Temporal and spatial variability of reduced sulfur species and porewater parameters in salt marsh sediments. Biogeochemistry 14, 57–88.

Luther, G.W.III, Kostka, J.E., Church, T.M., Sulzberger, B., Stumm, W., 1992. Seasonal iron cycling in the salt marsh sedimentary environment: the importance of ligand complexes with Fe(II) and Fe(III) in the dissolution of Fe(III) minerals and pyrite, respectively. Marine Chemistry 40, 81–103.

- Luther, G.W., III, Sundby, B., Lewis, B.L., Brendel, P.J., Silverberg, N., 1997. Interactions of manganese with the nitrogen cycle: alternative pathways to dinitrogen. Geochimica et Cosmochimica Acta 61, 4043–4052.
- Mackin, J.E., Aller, R.C., Ullman, W.J., 1988. The effects of iron reduction and nonsteadystate diagenesis on iodine, ammonium, and boron distributions in sediments from the Amazon continental shelf. Continental Shelf Research 8, 363–386.
- Mackin, J.E., Swider, K.T., 1989. Organic matter decomposition pathways and oxygen consumption in coastal marine sediments. Journal of Marine Research 47, 681–716.
- Madrid, V.M., Aller, J.Y., Aller, R.C., Chistoserdov, A.Y., 2001. High prokaryote diversity and analysis of community structure in mobile mud deposits off French Guiana: identification of two new bacterial candidate divisions. FEMS Microbiology Ecology 37, 197–209.

Marb, N., Holmer, M., Gacia, E., Barron, C., 2006. Seagrass beds and coastal biogeochemistry. In: Larkum, A., Orth, R., Duarte, C. (Eds.), Seagrasses: Biology, Ecology and Conservation. Springer, Dordrecht, pp. 135–157.

Marinelli, R.L., Boudreau, B.P., 1996. An experimental and modeling study of pH and related solutes in an irrigated anoxic coastal sediment. Journal of Marine Research 54, 939–966.

Marinelli, R.L., Jahnke, R.A., Craven, D.B., Nelson, J.R., Eckman, J.E., 1998. Sediment nutrient dynamics on the south Atlantic Bight continental shelf. Limnology and Oceanography 43, 1305–1320.

Martens, C.S., Albert, D.B., Alperin, M.J., 1998. Biogeochemical processes controlling methane in gassy coastal sediments – part 1. A model coupling organic matter flux to gas production, oxidation and transport. Continental Shelf Research 18, 1741–1770.

Martens, C.S., Berner, R.A., 1977. Interstitial water chemistry of anoxic Long Island Sound sediments. 1. Dissolved gases. Limnology and Oceanography 22, 10–25.

Martens, C.S., Chanton, J.P., 1989. Radon as a tracer of biogenic gas equilibration and transport from methane saturated sediments. Journal of Geophysical Research 94, 3451–3459.

Martens, C.S., Goldhaber, M.B., 1978. Early diagenesis in transitional sedimentary environments of the White Oak River estuary, North Carolina. Limnology and Oceanography 23, 428–441.

Martens, C.S., Haddad, R.I., Chanton, J.P., 1992. Organic matter accumulation, remineralization and burial in an anoxic marine sediment. In: Whelan, J.K., Farrington, J.W. (Eds.), Productivity, Accumulation, and Preservation of Organic Matter in Recent and Ancient Sediments. Columbia University Press, New York, NY, pp. 82–98.

Martens, C.S., Kipphut, G.W., Klump, J.V., 1980. Sediment–water chemical exchange in the coastal zone traced by *in situ* radon-222 flux measurements. Science 208, 285–288.

Martens, C.S., Klump, J.V., 1980. Biogeochemical cycling in an organic-rich marine basin – 1. Methane sediment–water exchange processes. Geochimica et Cosmochimica Acta 44, 471–490.

Martens, C.S., Klump, J.V., 1984. Biogeochemical cycling in an organic-rich marine basin – 4. An organic carbon budget for sediments dominated by sulfate reduction and methanogenesis. Geochimica et Cosmochimica Acta 48, 1987–2004.

Martin, J.B., Cable, J.E., Smith, C., Roy, M., Cherrier, J., 2007. Magnitudes of submarine groundwater discharge from marine and terrestrial sources: Indian River Lagoon, Florida. Water Resources Research 43, doi:10.1029/2006wr005266.

Martin, W.R., 2009. Chemical processes in estuarine sediments. In: Steele, J.H., Turekian, K.K., Thorpe, S.A. (Eds.), Encyclopedia of Ocean Sciences. Academic Press, Oxford, pp. 3549–3560.

Marvin-DiPasquale, M.C., Boynton, W.R., Capone, D.G., 2003. Benthic sulfate reduction along the Chesapeake Bay central channel. II. Temporal controls. Marine Ecology Progress Series 260, 55–70.

Marvin-DiPasquale, M.C., Capone, D.G., 1998. Benthic sulfate reduction along the Chesapeake Bay central channel. I. Spatial trends and controls. Marine Ecology Progress Series 168, 213–228.

Mayer, L.M., Chen, Z., Findlay, R.H., Fang, J., Sampson, S., Self, R.F.L., Jumars, P.A., Quetel, C., Donard, O.F.X., 1996. Bioavailability of sedimentary contaminants subject to deposit-feeder digestion. Environmental Science and Technology 30, 2641–2645.

McKee, B.A., Aller, R.C., Allison, M.A., Bianchi, T.S., Kineke, G.C., 2004. Transport and transformation of dissolved and particulate materials on continental margins influenced by major rivers: benthic boundary layer and seabed processes. Continental Shelf Research 24, 899–926.

McManus, J., Berelson, W.M., Coale, K., Johnson, K., Kilgore, T.E., 1997. Phosphorus regeneration in continental margin sediments. Geochimica et Cosmochimica Acta 61, 2891–2907.

McManus, J., Berelson, W.M., Severmann, S., Poulson, R.L., Hammond, D.E., Klinkhammer, G.P., Holm, C., 2006. Molybdenum and uranium geochemistry in continental margin sediments: paleoproxy potential. Geochimica et Cosmochimica Acta 70, 4643–4662.

- Meile, C., Berg, P., Van Capppellen, P., Tuncay, K., 2005. Solute-specific pore water irrigation: implications for chemical cycling in early diagenesis. Journal of Marine Research 63, 601–621.
- Mendelsson, I.A., McKee, K.L., Patrick, W.H., 1981. Oxygen deficiency in Spartina alterniflora roots: metabolic adaptation to anoxia. Science 214, 439–441.
- Meysman, F.J.R., Boudreau, B.P., Middelburg, J.J., 2003. Relations between local, nonlocal, discrete and continuous models of bioturbation. Journal of Marine Research 61, 391–410.
- Meysman, F.J.R., Boudreau, B.P., Middelburg, J.J., 2005a. Modelling reactive transport in sediments subject to bioturbation and compaction. Geochimica et Cosmochimica Acta 69, 3601–3617.

Meysman, F.J.R., Galaktionov, O.S., Cook, P.L.M., Janssen, F., Huettel, M., Middelburg, J.J., 2007. Quantifying biologically and physically induced flow and tracer dynamics in permeable sediments. Biogeosciences 4, 627–646.

Meysman, F.J.R., Galaktionov, O.S., Gribsholt, B., Middelburg, J.J., 2006. Bioirrigation in permeable sediments: advective pore-water transport induced by burrow ventilation. Limnology and Oceanography 51, 142–166.

Meysman, F.J.R., Galaktionov, O.S., Middelburg, J.J., 2005b. Irrigation patterns in permeable sediments induced by burrow ventilation: a case study of Arenicola marina. Marine Ecology Progress Series 303, 195–212.

- Michalopoulos, P., Aller, R.C., 2004. Early diagenesis of biogenic silica in the Amazon delta: alteration, authigenic clay formation, and storage. Geochimica et Cosmochimica Acta 68, 1061–1085.
- Middelburg, J.J., de Lange, G.J., van der Weijden, C.H., 1987. Manganese solubility control in marine pore waters. Geochimica et Cosmochimica Acta 51, 759.

Middelburg, J.J., Duarte, C.M., Gattuso, J.P., 2005. Respiration in coastal benthic communities. In: Del Giorgio, P., Williams, P. (Eds.), Respiration in Aquatic Ecosystem. Oxford University Press, New york, NY, pp. 206–224.

Middelburg, J.J., Klaver, G., Nieuwenhuize, J., Wilemake, A., de Hass, W., Vlug, T., van der Nat, J.F.W.A., 1996a. Organic matter mineralization in intertidal sediments along an estuarine gradient. Marine Ecology Progress Series 132, 157–168.

Middelburg, J.J., Levin, L.A., 2009. Coastal hypoxia and sediment biogeochemistry. Biogeosciences 6, 1273–1293.

Middelburg, J.J., Nieuwenhuize, J., Iversen, N., Høgh, N., de Wilde, H., Helder, W., Seifert, R., Christof, O., 2002. Methane distribution in European tidal estuaries. Biogeochemistry 59, 95–119.

Middelburg, J.J., Soetaert, K., Herman, P.M.J., Heip, C.H.R., 1996b. Denitrification in marine sediments: a model study. Global Biogeochemical Cycles 10, 661–673.

Miller, D., Geider, R., MacIntyre, H., 1996. Microphytobenthos: the ecological role of the "secret garden" of unvegetated, shallow-water marine habitats. II. Role in sediment stability and shallow-water food webs. Estuaries and Coasts 19, 202–212.

Miller, H.L., III, Meile, C., Burd, A.B., 2007. A novel 2D model of internal O₂ dynamics and H₂S intrusion in seagrasses. Ecological Modelling 205, 365–380.

Mitsch, W.J., Gosselink, J.G., 2000. Wetlands, Third ed. Wiley, New York, NY.

Moore, W.S., 1999. The subterranean estuary: a reaction zone of ground water and sea water. Marine Chemistry 65, 111–125.

Morel, F.M.M., Milligan, A.J., Saito, M.A., 2003. Marine bioinorganic chemistry: the role of trace metals in the oceanic cycles of major nutrients. In: Holland, H.D., Turekian, K.K. (Eds.), Treatise on Geochemistry. Pergamon, Oxford, vol. 6, pp. 113–143.

Morford, J.L., Emerson, S., 1999. The geochemistry of redox sensitive trace metals in sediments. Geochimica et Cosmochimica Acta 63, 1735–1750.

Morford, J.L., Martin, W.R., Francois, R., Carney, C.M., 2009. A model for uranium, rhenium, and molybdenum diagenesis in marine sediments based on results from coastal locations. Geochimica et Cosmochimica Acta 73, 2938–2960.

Moriarty, D.J.W., Iverson, R.L., Pollard, P.C., 1986. Exudation of organic carbon by the seagrass *Halodule wrightii* Aschers. And its effect on bacterial growth in the sediment. Journal of Experimental Marine Biology and Ecology 96, 115–126.

Morse, J.W., 1991. Oxidation kinetics of sedimentary pyrite in seawater. Geochimica et Cosmochimica Acta 55, 3665–3667.

Morse, J.W., 1994. Release of toxic metals via oxidation of authigenic pyrite in resuspended sediments. In: Alpers, C.N., Blowes, D.W. (Eds.), Environmental Geochemistry of Sulfide Oxidation. American Chemical Society, Washington, DC, pp. 289–297.

Morse, J.W., Berner, R.A., 1995. What determines sedimentary C/S ratios? Geochimica et Cosmochimica Acta 59, 1073–1077.

Morse, J.W., Dimarco, S.F., Sell, K.S., Hebert, A.B., 2003. Determination of the optimum sampling intervals in sediment pore waters using the autocovariance function. Aquatic Geochemistry 9, 41–57.

Morse, J.W., Luther, G.W., III, 1999. Chemical influences on trace metal-sulfide

interaction in anoxic sediments. Geochimica et Cosmochimica Acta 63, 3373–3378.
Morse, J.W., Millero, F.J., Cornwell, J.C., Rickard, D., 1987. The chemistry of the hydrogen sulfide and iron sulfide systems in natural waters. Earth-Science Reviews 24, 1–42.

Morse, J.W., Wang, Q., 1997. Pyrite formation under conditions approximation those in anoxic sediments: II. Influence of precursor iron minerals and organic matter. Marine Chemistry 57, 187–193. Mortimer, R.J.G., Harris, S.J., Krom, M.D., Freitag, T.E., Prosser, J.I., Barnes, J., Anschutz, P., Hayes, P.J., Davies, I.M., 2004. Anoxic nitrification in marine sediments. Marine Ecology Progress Series 276, 37–52.

Mortimer, R.J.G., Krom, M.D., Harris, S.J., Hayes, P.J., Davies, I.M., Davison, W., Zhang, H., 2002. Evidence for suboxic nitrification in recent marine sediments. Marine Ecology Progress Series 236, 31–35.

Mucci, A., Boudreau, B., Guignard, C., 2003. Diagenetic mobility of trace elements in sediments covered by a flash flood deposit: Mn, Fe and As. Applied Geochemistry 18, 1011–1026.

Myers, C.R., Nealson, K.H., 1988a. Bacterial manganese reduction and growth with manganese oxide as the sole electron acceptor. Science 240, 1319–1321.

Myers, C.R., Nealson, K.H., 1988b. Microbial reduction of manganese oxides: interactions with iron and sulfur. Geochimica et Cosmochimica Acta 52, 2727–2732.

- Nealson, K.H., 1997. Sediment bacteria: who's there, what are they doing, and what's new? Annual Review of Earth and Planetary Sciences 25, 403–434.
- Nealson, K.H., Saffarini, D., 1994. Iron and manganese in anaerobic respiration: environmental significance, physiology, and regulation. Annual Review of Microbiology 48: 311–343.
- Nixon, S., Ammerman, J., Atkinson, L., Berounsky, V., Billen, G., Boicourt, W., Boynton, W., Church, T., Ditoro, D., Elmgren, R., Garber, J., Giblin, A., Jahnke, R., Owens, N., Pilson, M., Seitzinger, S. 1996. The fate of nitrogen and phosphorus at the land–sea margin of the North Atlantic Ocean. Biogeochemistry 35, 141–180.

Orphan, V.J., House, C.H., Hinrichs, K.-U., McKeegan, K.D., DeLong, E.F., 2001. Methane-consuming archaea revealed by directly-coupled isotopic and phylogenetic analysis. Science 293, 484–487.

Palomo, L., Canuel, E., 2010. Sources of fatty acids in sediments of the York River Estuary: Relationships with physical and biological processes. Estuaries and Coasts 33, 585–599.

Papadimitriou, S., Kennedy, H., Kennedy, D., Duarte, C., Marba, N., 2005. Sources of organic matter in seagrass-colonized sediments: a stable isotope study of the silt and clay fraction from Posidonia oceanica meadows in the western Mediterranean. Organic Geochemistry 36, 949–961.

Peterson, P.H., Luettich, R.A.J., Micheli, F., Skilleter, G.A., 2004. Attenuation of water flow inside seagrass canopies of differing structures. Marine Ecology Progress Series 268, 81–92.

- Point, D., Monperrus, M., Tessier, E., Amouroux, D., Chauvaud, L., Thouzeau, G., Jean, F., Amice, E., Grall, J., Leynaert, A., Clavier, J., Donard, O.F.X., 2007. Biological control of trace metal and organometal benthic fluxes in a eutrophic lagoon (Thau Lagoon, Mediterranean Sea, France). Estuarine, Coastal and Shelf Science 72, 457–471.
- Post, J.E., 1999. Manganese oxide minerals: crystal structures and economic and environmental significance. Proceedings of the National Academy of Sciences 96, 3447–3454.
- Postma, D., Jakobsen, R., 1996. Redox zonation: equilibrium constraints on the Fe(III)/ S04^{2–} reduction interface. Geochimica et Cosmochimica Acta 60, 3169–3175.
- Povidisa, K., Delefosse, M., Holmer, M., 2009. The formation of iron plaques on roots and rhizomes of the seagrass Cymodocea serrulata (R. Brown) Ascherson with implications for sulphide intrusion. Aquatic Botany 90(4), 303–308.
- Precht, E., Franke, U., Polerecky, L., Huettel, M., 2004. Oxygen dynamics in permeable sediments with wave-driven pore water exchange. Limnology and Oceanography 49, 693–705.
- Raghoebarsing, A.A., Pol, A., van de Pas-Schoonen, K.T., Smolders, A.J.P., Ettwig, K.F., Rijpstra, W.I.C., Schouten, S., Damste, J.S.S., Op den Camp, H.J.M., Jetten, M.S.M., Strous, M., 2006. A microbial consortium couples anaerobic methane oxidation to denitrification. Nature 440, 918–921.
- Raiswell, R., Canfield, D.E., 1998. Sources of iron for pyrite formation in marine sediments. American Journal of Science 298, 219–245.

Rao, A.M.F., McCarthy, M.J., Gardner, W.S., Jahnke, R.A., 2007. Respiration and denitrification in permeable continental shelf deposits on the South Atlantic Bight: rates of carbon and nitrogen cycling from sediment column experiments. Continental Shelf Research 27, 1801–1819.

- Rao, A.M.F., McCarthy, M.J., Gardner, W.S., Jahnke, R.A., 2008. Respiration and denitrification in permeable continental shelf deposits on the South Atlantic Bight: N₂:Ar and isotope pairing measurements in sediment column experiments. Continental Shelf Research 28, 602–613.
- Rasheed, M., Wild, C., Franke, U., Huettel, M., 2004. Benthic photosynthesis and oxygen consumption in permeable carbonate sediments at Heron Island, Great Barrier Reef, Australia. Estuarine. Coastal and Shelf Science 59. 139–150.
- Reeburgh, W.S., 1969. Observations of gases in Chesapeake Bay sediments. Limnology and Oceanography 14, 368–375.
- Reeburgh, W.S., 1982. A major sink and flux control for methane in marine sediments: anaerobic consumption. In: Fanning, K.A., Manheim, F.T. (Eds.), The Dynamics of the Ocean Floor. D.C. Heath, Lexington, MA, pp. 203–218.

Reeburgh, W.S., 2007. Oceanic methane biogeochemistry. Chemical Reviews 107, 486-513.

- Reimers, C., Stecher, H.A., III, Taghon, G.L., Fuller, C.M., Huettel, M., Rusch, A., Ryckelynck, N., Wild, C., 2004. *In situ* measurements of advective solute transport in permeable shelf sands. Continental Shelf Research 24, 183–201.
- Reimers, C.E., Jahnke, R.A., Thomsen, L., 2001. *In situ* sampling in the benthic boundary layer. In: Boudreau, B.P., Jørgensen, B.B. (Eds.), The Benthic Boundary Layer. Oxford University Press, New York, NY, pp. 245–268.
- Rhoads, D.C., 1974. Organism–sediment relations on the muddy sea floor. Oceanography and Marine Biology: An Annual Review 12, 263–300.
- Rich, J., Dale, O., Song, B., Ward, B., 2008. Anaerobic ammonium oxidation (anammox) in Chesapeake Bay sediments. Microbial Ecology 55, 311–320.
- Rickard, D., 1997. Kinetics of pyrite formation by the H_2S oxidation of iron (II) monosulfide in aqueous solutions between 25 and 125 °C: the rate equation. Geochimica et Cosmochimica Acta 61, 115–134.
- Rickard, D., Luther, G.W., III, 2007. Chemistry of iron sulfides. Chemical Reviews 107, 514–562.
- Rickard, D., Schoonen, M.A.A., Luther G.W., III, 1995. Chemistry of iron sulfides in sedimentary environments. In: Vairavamurthy, M.A., Schoonen, M.A.A. (Eds.), Geochemical Transformation of Sedimentary Sulfur. American Chemical Society, Washington, DC, pp. 168–193.
- Riedel, G.F., Sanders, J.G., Osman, R.W., 1997. Biogeochemical controls on the flux of trace elements from estuarine sediments: water column oxygen concentrations and benthic infauna. Estuarine, Coastal and Shelf Science 44, 23–38.
- Robbins, J.A., 1986. A model for particle-selective transport of tracers in sediments with conveyor belt deposit feeders. Journal of Geophysical Research 91, 8542–8558.
- Robert, S., Blanc, G., Schäfer, J., Lavaux, G., Abril, G., 2004. Metal mobilization in the Gironde Estuary (France): the role of the soft mud layer in the maximum turbidity zone. Marine Chemistry 87, 1–13.
- Roden, E.E., Tuttle, J.H., 1993. Inorganic sulfur turnover in oligohaline estuarine sediments. Biogeochemistry 22, 81–105.
- Roden, E.E., Tuttle, J.H., Boynton, W.R., Kemp, W.M., 1995. Carbon cycling in mesohaline Chesapeake Bay sediments 1. POC deposition rates and mineralization pathways. Journal of Marine Research 53, 799–819.
- Roden, E.E., Wetzel, R.G., 1996. Organic carbon oxidation and supression of methane production by microbially Fe(III) oxide reduction in vegetated and unvegetated freshwater wetland sediments. Limnology and Oceanography 41, 1733–1748.
- Roden, E.E., Zachara, J.M., 1996. Microbial reduction of crystalline iron(III) oxides: influence of oxide surface area and potential for cell growth. Environmental Science and Technology 30, 1618–1628.
- Roy, M., Martin, J.B., Cherrier, J., Cable, J.E., Smith, C.G., 2010. Influence of sea level rise on iron diagenesis in an east Florida subterranean estuary. Geochimica et Cosmochimica Acta 74(19), 5560–5573.
- Rusch, A., Huettel, M., 2000. Advective particle transport into permeable sediments evidence from experiments in an intertidal sandflat. Limnology and Oceanography 45, 525–533.
- Ruttenberg, K.C., 2003. The global phosphorus cycle. In: Schlesinger, W.C. (Ed.), Treatise on Geochemistry. Elsevier, Amsterdam, pp. 585–643.
- Rysgaard, S., Thamdrup, B., Risgaard-Petersen, N., Fossing, H., Berg, P., Christensen, P.B., Dalsgaard, T., 1998. Seasonal carbon and nutrient mineralization in a high-Arctic coastal marine sediment, Young Sound, Northeast Greenland. Marine Ecology Progress Series 175, 261–276.
- Salwan, J.J., Murray, J.W., 1983. Trace metal remobilization in the interstitial waters of red clay and hemi-pelagic marine sediments. Earth and Planetary Science Letters 64, 213–230.
- Sampou, P., Oviatt, C.A., 1991. A carbon budget for a eutrophic marine ecosystem and the role of sulfur metabolism in sedimentary carbon, oxygen and energy dynamics. Journal of Marine Research 49, 825–849.
- Saulnier, I., Mucci, A., 2000. Trace metal remobilization following the resuspension of estuarine sediments: Saguenay Fjord, Canada. Applied Geochemistry 15, 191–210.
- Savrda, C.E., Bottjer, D.J., Gorsline, D.S., 1984. Development of a comprehensive oxygendeficient biofacies model: evidence from Santa Monica, San Pedro and Santa Barbara Basins, California Continental Borderland. AAPG Digital Bulletin 68, 1179–1192.
- Schaffner, L.C., Dellapenna, T.M., Hinchey, E.K., Friedrichs, C.T., Neubauer, M.T., Smith, M.E., Kuehl, S.A., 2001. Physical energy regimes, seabed dynamics and organism-sediment interactions along an estuarine gradient. In: Aller, J.Y., Woodin, S.A., Aller, R.C. (Eds.), Organism-Sediment Interactions. University of South Carolina Press, Columbia, SC, pp. 159–179.
- Schaffner, L.C., Diaz, R.J., Olsen, C.J., Larsen, I.L., 1987. Faunal charactersitics and sediment accumulation processes in the James River estuary, Virginia. Estuarine, Coastal and Shelf Science 25, 211–226.
- Schippers, A., Jørgensen, B.B., 2001. Oxidation of pyrite and iron sulfide by manganese dioxide in marine sediments. Geochimica et Cosmochimica Acta 65, 915–922.
- Schippers, A., Jørgensen, B.B., 2002. Biogeochemistry of pyrite and iron sulfide oxidation in marine sediments. Geochimica et Cosmochimica Acta 66, 85–92.

- Schippers, A., Neretin, L.N., Lavik, G., Leipe, T., Pollehne, F., 2005. Manganese(II) oxidation driven by lateral oxygen intrusions in the western Black Sea. Geochimica et Cosmochimica Acta 69, 2241–2252.
- Schloss, P.D., Handelsman, J., 2004. Status of the Microbial Census. Microbiology and Molecular Biology Reviews 68, 686–691.
- Scholz, F., Neumann, T., 2007. Trace element diagenesis in pyrite-rich sediments of the Achterwasser lagoon, SW Baltic Sea. Marine Chemistry 107, 516–532.
- Schoonen, M.A.A., Barnes, H.L., 1991. Reactions forming pyrite and marcasite from solution: II. Via FeS precursors below 100 °C. Geochimica et Cosmochimica Acta 55, 1505–1514.
- Schulz, H.N., Jørgensen, B.B. 2001. Big bacteria. Annual Review of Microbiology 55, 105–137.
- Seitzinger, S.P., Giblin, A.E., 1996. Estimating denitrification in North Atlantic continental margin sediments. Biogeochemistry 35, 235–259.
- Shaw, T.J., Gieskes, J.M., Jahnke, R.A., 1990. Early diagenesis in differing depositional environments: the response of transition metals in pore waters. Geochimica et Cosmochimica Acta 54, 1233–1246.
- Shimmield, G.B., Pedersen, T.F., 1990. The geochemistry of reactive trace metals and halogens in hemipelagic continental margin sediments. Reviews in Aquatic Sciences 3, 255–279.
- Shine, J.P., Ika, R., Ford, T.E., 1998. Relationship between oxygen consumption and sediment–water fluxes of heavy metals in coastal marine sediments. Environmental Toxicology and Chemistry 17, 2325–2337.
- Sholkovitz, E., 1973. Interstitial water chemistry of the Santa Barbara Basin. Geochimica et Cosmochimica Acta 37, 2043–2073.
- Shum, K.T., Sundby, B., 1996. Organic matter processing in continental shelf sediments – the subtidal pump revisited. Marine Chemistry 53, 81–87.
- Skrabal, S.A., Donat, J.R., Burdige, D.J., 1997. Fluxes of copper-complexing ligands from estuarine sediments. Limnology and Oceanography 42, 992–996.
- Skrabal, S.A., Donat, J.R., Burdige, D.J., 2000. Pore water distributions of dissolved copper and copper-complexing ligands in estuarine and coastal marine sediments. Geochimica et Cosmochimica Acta 64, 1843–1857.
- Skrabal, S.A., Lieseke, K.L., Kieber, R.J., 2006. Dissolved zinc and zinc-complexing ligands in an organic-rich estuary: benthic fluxes and comparison with copper speciation. Marine Chemistry 100, 108–123.
- Smith, C.R., DeMaster, D.J., Fornes, W.L., 2001. Mechanisms of age-dependent bioturbation on the bathyl California margin: the young and the restless. In: Aller, J.Y., Woodin, S.A., Aller, R.C. (Eds.), Organism–Sediment Interactions. University of South Carolina Press, Columbia, SC, pp. 263–277.
- Smith, C.R., Rabouille, C., 2004. Reply to comment by Boudreau on: what controls the mixed-layer depth in deep-sea sediments? The importance of POC flux. Limnology and Oceanography 49, 623–624.
- Snelgrove, P.V.R., Butman, C.A., 1994. Animal–sediment relationships revisited: cause verus effect. Oceanography and Marine Biology: An Annual Review 32, 111–177.
- Soetaert, K., Herman, P.M.J., Midelburg, J.J., 1996. Dynamic response of deep-sea sediments to seasonal variations: a model. Limnology and Oceanography 41, 1651–1668.
- Sommerfield, C.K., Nittrouer, C.A., Figueiredo, A.G., 1995. Stratigraphic evidence of changes in Amazon shelf sedimentation during the late Holocene. Marine Geology 125, 351–371.
- Sørensen, K.B., Canfield, D.E., 2004. Annual fluctuations in sulfur isotope fractionation in the water column of a euxinic marine basin. Geochimica et Cosmochimica Acta 68, 503–515.
- St-Cyr, L., Fortin, D., Campbell, P.G.C., 1993. Microscopic observations of the iron plaque of a submerged aquatic plant (*Vallisneria americana* Michx). Aquatic Botany 46, 155–167.
- Stephens, F.C., Louchard, E.M., Reid, R.P., Maffione, R.A., 2003. Effects of microalgal communities on reflectance spectra of carbonate sediments in subtidal optically shallow marine environments. Limnology and Oceanography 48, 535–546.
- Stone, A.T., Godtfredsen, K.L., Deng, B., 1994. Sources and reactivity of reductants encountered in aquatic environments. In: Bidoglio, G., Stumm, W. (Eds.), Chemistry of Aquatic Systems: Local and Global Perspectives. Kluwer, Brussels, pp. 337–374.
- Straub, K.L., Benz, M., Schink, B., Widdel, F., 1996. Anaerobic, nitrate-dependent microbial oxidation of ferrous iron. Applied and Environmental Microbiology 62, 1458–1460.
- Strous, M., Jetten, M.S.M., 2004. Anaerobic oxidation of ammonium and methane. Annual Review of Microbiology 58, 99–117.
- Stumm, W., Morgan, J.J., 1996. Aquatic Chemistry: Chemical Equilibria and Rates in Natural Waters. Wiley-Interscience, New York, NY.
- Suess, E., 1979. Mineral phases formed in anoxic sediments by microbial decomposition of organic matter. Geochimica et Cosmochimica Acta 43, 339–341.
- Sun, M., Aller, R.C., Lee, C., 1991. Early diagenesis of chlorophyll-a in Long Island Sound sediments: a measure of carbon flux and particle reworking. Journal of Marine Research 49, 379–401.

- Sun, M., Aller, R.C., Lee, C., Wakeham, S.G., 2002. Effects of oxygen and redox oscillation on degradation of cell-associated lipids in surfical marine sediments. Geochimica et Cosmochimica Acta 66, 2003–2012.
- Sunda, W.G., Kieber, D.J., 1994. Oxidation of humic substances by manganese oxides yields low-molecular-weight organic substrates. Nature 367, 62–64.
- Sundback, K., McGlather, K., 2005. Interactions between bethic macoalgal and microalgal mats. In: Kristensen, E., Haese, R.R., Kostka, J.E. (Eds.), Interactions between Macro- and Microorganisms in Marine Sediments. American Geophysical Union, Washington, DC, pp. 7–29.
- Sundby, B., Anderson, L.G., Hall, P.O.J., Iverfeldt, A., van der Loeff, M.M.R., Westerlund, S.F.G., 1986. The effect of oxygen on the release of cobalt, manganese, iron and phosphate at the sediment–water interface. Geochimica et Cosmochimica Acta 50, 1281–1288.
- Sundby, B., Silverberg, N., 1985. Manganese fluxes in the benthic boundary layer. Limnology and Oceanography, 30, 372–381.
- Sundby, B., Vale, C., Caçador, I., Catarino, F., Madureira, M.-J., Caetano, M., 1998. Metal-rich concretions on the roots of salt marsh plants: mechanism and rate of formation. Limnology and Oceanography 43, 245–252.
- Sundby, B., Vale, C., Caetano, M., Luther, G.W., 2003. Redox chemistry in the root zone of a salt marsh sediment in the Tagus Estuary, Portugal. Aquatic Geochemistry 9, 257–271.
- Sweeney, R.E., Kaplan, I.R., 1973. Pyrite framboid formation: laboratory synthesis and marine sediments. Economic Geology 68, 618–634.
- Swider, K.T., Mackin, J.E., 1989. Transformations of sulfur compounds in marsh-flat sediments. Geochimica et Cosmochimica Acta 53, 2311–2323.
- Taillefert, M., Neuhuber, S., Bristow, G., 2007. The effect of tidal forcing on biogeochemical processes in intertidal salt marsh sediments. Geochemical Transactions 8, 6.
- Tebo, B.M., Ghiorse, W.C., van Waasbergen, L.G., Siering, P.L., Caspi, R., 1997. Bacterially mediated mineral formation: insights into manganese(II) oxidation from molecular genetic and biochemical studies. In: Banfield, J.F., Nealson, K.H. (Eds.), Geomicrobiology: Interactions between Microbes and Minerals. Reviews in Mineralogy. Mineralogical Society of America, Washington, DC, pp. 225–266.
- Tejada-Martinez, A.E., Grosch, C.E., 2007. Langmuir turbulence in shallow water: part 2. Large-eddy simulation. Journal of Fluid Mechanics 576, 63–108.
- Tengberg, A., Almroth, E., Hall, P., 2003. Resuspension and its effects on organic carbon recycling and nutrient exchange in coastal sediments: *in situ* measurements using new experimental technology. Journal of Experimental Marine Biology and Ecology 285–286, 119–142.
- Thamdrup, B., 2000. Bacterial manganese and iron reduction in aquatic sediments. Advances in Microbial Ecology 16, 41–83.
- Thamdrup, B., Canfield, D.E., 2000. Benthic respiration in aquatic sediments. In: Sala, O.E., Jackson, R.B., Mooney, H.A., Howarth, R.A. (Eds.), Methods in Ecosystem Science. Springer, New York, NY, pp. 86–102.
- Thamdrup, B., Dalsgaard, T., 2002. Production of N₂ through anaerobic ammonium oxidation coupled to nitrate reduction in marine sediments. Applied and Environmental Microbiology 68, 1312–1318.
- Thamdrup, B., Fossing, H., Jørgensen, B.B., 1994a. Manganese, iron and sulfur cycling in a coastal marine sediment, Aarhus bay, Denmark. Geochimica et Cosmochimica Acta 58, 5115–5129.
- Thamdrup, B., Glud, R.N., Hansen, J.W., 1994b. Manganese oxidation and *in situ* manganese fluxes from a coastal sediment. Geochimica et Cosmochimica Acta 58, 2563–2570.
- Thamdrup, B., Glud, R.N., Hansen, J.W., 2007. Benthic carbon cycling in Young Sound, Northeast Greenland. In: Rysgaard, S., Glud, R.N. (Eds.), Carbon Cycling in Arctic Marine Ecosystem: Case Study, Youngs Sound. Meddelelser om Grønland, BioscienceCopenhagen. 58, pp. 138–157.
- Thamdrup, B., Rossello-Mora, R., Amann, R., 2000. Microbial manganese and sulfate reduction in Black Sea shelf sediments. Applied and Environmental Microbiology 66, 2888–2897.
- Therkildsen, M.S., Lomstein, B.A., 1993. Seasonal variation in net benthic Cmineralization in a shallow estuary. FEMS Microbiology Ecology 12, 131–142.
- Tobias, C.R., Harvey, J.W., Anderson, I.C., 2001. Quantifying groundwater discharge through fringing wetlands to estuaries: seasonal variability, methods comparison, and implications for wetland–estuary exchange. Limnology and Oceanography 46, 604–615
- Trimmer, M., Nicholls, J.C., Deflandre, B., 2003. Anaerobic ammonium oxidation measured in sediments along the Thames Estuary, United Kingdom. Applied and Environmental Microbiology 69, 6447–6454.
- Vairavamurthy, M.A., Orr, W.L., Manowitz, B., 1995. Geochemical transformations of sedimentary sulfur: an introduction. In: Vairavamurthy, M.A. and Schoonen, M.A.A. (Eds.), Geochemical Transformations of Sedimentary Sulfur. American Chemical Society, Washington, DC, pp. 1–14.
- Vale, C., Sundby, B., 1998. The interactions between living organisms and metals in intertidal and subtidal sediments. In: Langston, W.J., Bebianno, M.J. (Eds.),

Metal Metabolism in Aquatic Environments. Chapman and Hall, London, pp. 19–29.

van der Nat, F.-J.W.A., Middelburg, J.J., 1998. Effects of two common macrophytes on methane dynamics in freshwater sediments. Biogeochemistry 43, 79–104.

Vance-Harris, C., Ingall, E., 2005. Denitrification pathways and rates in the sandy sediments of the Georgia continental shelf, USA. Geochemical Transactions 6, doi:10.1186/1467-4866-6-12.

Volkenborn, N., Polerecky, L., Hedtkamp, S.I.C., Beusekom, J.E.E.V., Beer, D.D., 2007. Bioturbation and bioirrigation extend the open exchange regions in permeable sediments. Limnology and Oceanography 52, 1898–1909.

Wang, Q., Morse, J.W., 1996. Pyrite formation under conditions approximating those in anoxic sediments I. Pathway and morphology. Marine Chemistry 52, 99–121.

Ward, B.B., 2008. Nitrification in marine systems. In: Capone, D.G., Bronk, D.A., Mulholland, M.R., Carpenter, E.J. (Eds.), Nitrogen in the Marine Environment, Second ed. Academic Press/Elsevier, New York, NY.

Ward, B.B., Devol, A.H., Rich, J.J., Chang, B.X., Bulow, S.E., Naik, H., Pratihary, A., Jayakumar, A., 2009. Denitrification as the dominant nitrogen loss process in the Arabian Sea. Nature 461, 78–81.

Warnken, K.W., Gill, G.A., Griffin, L.L., Santschi, P.H. 2001. Sediment–water exchange of Mn, Fe, Ni, and Zn in Galveston Bay, Texas. Marine Chemistry 73, 215–231.

Westerlund, S.F.G., Anderson, L.G., Hall, P.O.J., Iverfeldt, A., van der Loeff, M.M.R., Sundby, B., 1986. Benthic fluxes of cadmium, copper, nickel, zinc, and lead in the coastal environment. Geochimica et Cosmochimica Acta 50, 1289–1296.

Westrich, J.T., Berner, R.A., 1984. The role of sedimentary organic matter in bacterial sulfate reduction: the G model tested. Limnology and Oceanography 29, 236–249.

Wetzel, R.G., Penhale, P.A., 1979. Transport of carbon and excretion of dissolved organic carbon by leaves and roots/rhizomes in seagrasses and their epiphytes. Aquatic Botany 6, 149–158.

Wever, T.F., Abegg, F., Fiedler, H.M., Fechner, G., Stender, I.H., 1998. Shallow gas in the muddy sediments of Eckernforde Bay, Germany. Continental Shelf Research 18, 1715–1739.

Whiting, G.J., Chanton, J.P., 2001. Greenhouse carbon balance of wetlands: methane emission versus carbon sequestration. Tellus B 53, 521–528.

- Wilkin, R.T., Barnes, H.L., 1996. Pyrite formation by reactions of iron monosulfides with dissolved inorganic and organic sulfur species. Geochimica et Cosmochimica Acta 60, 4167–4179.
- Wilson, A.M., Huettel, M., Klein, S., 2008. Grain size and depositional environment as predictors of permeability in coastal marine sands. Estuarine, Coastal and Shelf Science 80, 193–199.
- Winfrey, M.R., Zeikus, J.G., 1977. Effect of sulfate on carbon and electron flow during microbial methanogenesis in freshwater sediments. Applied Environmental Microbiology 33, 275–28.
- Wright, L.D., Prior, D.B., Hobbs, C.H., Byrne, R.J., Boon, J.D., Schaffner, L.C., Green, M.O., 1987. Spatial variability of bottom types in the lower Chesapeake Bay and adjoining estuaries and the inner shelf. Estuarine, Coastal and Shelf Science, 24, 765–784.
- Wright, L.D., Schaffner, L.C., Maa, J.P.Y., 1997. Biological mediation of bottom boundary layer processes and sediment suspension in the lower Chesapeake Bay. Marine Geology 141, 27–50.
- Yao, W., Millero, F.J., 1993. The rate of sulfide oxidation by δMnO₂ in seawater. Geochimica et Cosmochimica Acta 57, 3359–3365.
- Yao, W., Millero, F.J., 1996. Oxidation of hydrogen sulfide by hydrous Fe(III) oxides in seawater. Marine Chemistry 52, 1–16.

Zehr, J.P., Ward, B.B., 2002. Nitrogen cycling in the ocean: new perspectives on processes and paradigms. Applied and Environmental Microbiology 68, 1015–1024.

Zhu, Q., Aller, R.C., Fan, Y., 2006. Two-dimensional pH distributions and dynamics in bioturbated marine sediments. Geochimica et Cosmochimica Acta 70, 4933–4949.

Zhu, Z., Aller, R.C., Mak, J., 2002. Stable carbon isotope cycling in mobile coastal muds of Amanå, Brazil. Continental Shelf Research 22, 2065–2079.

- Ziebis, W., Forster, S., Huettel, M., Jorgensen, B.B., 1996a. Complex burrows of the mud shrimp *Callianassa truncata* and their geochemical impact in the sea bed. Nature 382, 619–622.
- Ziebis, W., Huettel, M., Forster, S., 1996b. Impact of biogenic sediment topography on oxygen fluxes in permeable seabeds. Marine Ecology Progress Series 140, 227–237.
- Ziegler, S., Benner, R., 1999. Dissolved organic carbon cycling in a subtropical seagrass-dominated lagoon. Marine Ecology Progress Series 180, 149–160.